Achievements, durativity and scales

Inaugural-Dissertation zur Erlangung des Doktorgrades der Philosophie (Dr. phil.) durch die Philosophische Fakultät der Heinrich-Heine-Universität Düsseldorf

vorgelegt von

Zsófia Gyarmathy

aus Budapest, Ungarn

Betreuerin: Univ.-Prof. Dr. Hana Filip

Düsseldorf, Juni 2015

Abstract

The goal of this dissertation is to provide a novel formal semantic analysis of achievement predicates (in Vendler's sense) in the scope of the progressive. The focus is on the single-event progressive interpretations of two subtypes of achievements, namely, culminations and happenings (in the sense of Bach 1986): *i*) a "preliminary process" interpretation characteristic of culminations, as in *John is arriving*, and *ii*) a "slowmotion camera" interpretation characteristic of happenings, as in *The bomb is exploding*. Given the differences in the interpretation of culminations and happenings in the progressive, I subscribe to this distinction among Vendler's achievements, following authors like Carlson (1981), Bach (1986), Filip (1993/99), Piñón (1997). I offer a characterisation of these and other aspectual classes in a scale-based framework for aspect. On my account, the scale-based characterisation of these predicates then interacts with a scale-based semantics of the progressive to derive the correct truth-conditions.

The analysis of a "slow-motion camera" interpretation of happenings presupposes that we do away with an absolute notion of the temporal trace function and relativise it to a granularity parameter, based on the independently proposed notion of *granularity functions* over degrees of scales (Sauerland and Stateva 2007, 2011). The distinct advantage of this new proposal is that it allows us to exploit the same set of tools in approaching several independent issues in the domain of aspectual classes: namely, the minimal parts problem of activities and the vagueness of the temporal boundaries of events.

My analysis of "preliminary process" progressives exploits the notion of a *scale of change* and a dual characterisation of culminations as being associated with two kinds of scales. The concept of the scale of change I employ is an amalgam of the scale of change concept of Beavers (2008, 2013) that captures the change associated with a predicate and Asher's (1992) concept of a perspective that captures the contextually relevant properties of an eventuality. The proposed framework is a generalization of scale-based accounts of aspectual classes in that the scale-based approach is general-

ized to nearly all verbal predicates. As a further departure from earlier scale-based approaches, and in particular, Beavers (2008, 2013), I argue that culmination predicates are associated with both two-valued scales of change characterising events in their extension, and the multi-valued scales of change characterising events of gradual change that I call their "cover events". Informally, the cover events of a culmination-achievement (e.g., *arrive at the station*) are events (e.g., walking to the station-events, driving to the station-events etc.) whose culminations belong to the extension of the culmination-achievement.

I propose that the semantics of the progressive operator involves partitivity over scales of change rather than events or event runtimes, and since culmination-achievements (in contrast to happenings) are associated with *both* two-valued and multivalued scales of change, their combination with the progressive is predicted. At the same time, since only culmination-achievements, but not accomplishments are associated with two-point scales, the difference between these two classes is upheld.

Thus, a scale-based framework for both aspectual classes (aka situation aspect) and grammatical aspect (aka viewpoint aspect) can be used to capture the difference between accomplishments, culminations and happenings and predict the interpretations of the progressive for different aspectual classes by appealing to scales of change over and above temporal and mereological relations in the domain of eventualities. Through employing scales of change and granularity functions as a means of accounting for the use of achievements in the progressive, the dissertation contributes to the growing work on scale-based approaches to verbal phenomena.

The present dissertation was published in hard copy by Logos Verlag in 2015, ISBN 978-3-8325-4076-0 (http://www.logos-verlag.de/cgi-bin/buch?isbn=4076).

ACKNOWLEDGEMENTS

It is a truism that writing a dissertation is not the accomplishment of a single individual, but is a synergy of interaction between the writer and numerous other people. This particular dissertation could not have come about without the valuable professional and personal help and contribution of several friends and colleagues. I would like to thank Hana Filip and Károly Varasdi for the supervision and support I received from them, as well as for the long, fruitful discussions and the countless comments and invaluable feedback they gave me on several drafts of the thesis. I am likewise indebted to Daniel Altshuler and Fabienne Martin who also read a number of earlier versions of my work, and provided me with insightful suggestions that benefited the dissertation to a great extent. Sincere thanks go to Peter Sutton for the important comments he had and great ideas he put forth on reading some chapters, and to Eva Kardos for reading a first draft of the dissertation and providing me with great insights and comments that contributed to the newer version. Thanks are due to Katalin Csatár and Péter Rebrus, as well, for helping me not only understand, but also appreciate the workings and the point of mathematical methodology. And I am grateful to members of the semantic discussion groups, as well as, all my teachers and fellow students in Düsseldorf and Budapest for both the specific feedback and the constant academic input they provided me. Needless to say, any errors and shortcomings that remained in the dissertation are all exclusively mine, though.

Finally, I would also like to express my thanks to all my friends and family for their personal support and their patience in the face of continuous (and after a while, I'm afraid, rather repetitive) declarations on my part of "being snowed under", "needing to do just one more week of work", and in general, "not needing to buy a raccoon". I truly appreciate the professional and personal support of all my friends, family and colleagues who helped in me in my research project. I dedicate this work to Sándor Gyarmathy, who did not have a chance to support me in this specific enterprise, but whose love and support was and remains an invaluable asset.

GLOSSING CONVENTIONS

ABL	ablative
ACC	accusative
ERG	ergative
IMPF	imperfective
PAST	past
PFV	perfective

- PROG progressive
- PRT preverbal particle
- SUBL sublative

Table of Contents

A	BSTRA	АСТ		i
A	CKNO	WLEDO	GEMENTS	iii
Gı	LOSSI	NG CO	NVENTIONS	v
TA	BLE (OF CON	JTENTS	vi
LI	ST OF	FIGUR	ES	ix
LI	ST OF	TABLE	S	x
LI	ST OF	DEFIN	ITIONS AND PRINCIPLES	xi
1	Inti	RODUC	TION	1
	 1.1 1.2 1.3 	Aspec 1.1.1 1.1.2 1.1.3 1.1.4 Issues Struct	tual classes Aspectual classifications Aspectual classifications Aspectual classifications Accomplishments and achievements Aspectual classifications Previous classifications of Vendler's achievements Aspectual classifications Ontology and terminology to be assumed Aspectual classer and proposal Aspectual classer ure of the dissertation Aspectual classer	1 4 6 9 11 16
2	Тне 2.1	CORE Vendl 2.1.1 2.1.2	DATA AND A REVIEW OF THE RELEVANT THEORIES er's achievements in the scope of the progressive	20 20 20 22
	2.2	Previo 2.2.1 2.2.2 2.2.3	Ous analyses of progressive achievements	28283235

		2.2.4	Type shifting approaches	38
		2.2.5	A non-partitive semantics for the progressive	43
	2.3	Scale-	based frameworks	44
		2.3.1	Scales in formal semantic analysis	44
		2.3.2	Former scale-based frameworks for aspectual analysis	47
3	The	E POINT	OF DEPARTURE FOR A NEW ACCOUNT OF PROGRESSIVE ACHIEVE-	
	MEN	JTS		54
	3.1	Durat	ivity	54
		3.1.1	The linguistic relevance of durativity	54
		3.1.2	The moment of change	57
	3.2	Existe	ential presupposition of culminations	59
		3.2.1	Culminations as presupposition triggers	59
		3.2.2	Extra soft presuppositions and the reverse direction	62
	3.3	The b	asic ingredients	66
		3.3.1	The semantic model	66
		3.3.2	Scales	68
4	Slo	W-MOT	TION INTERPRETATIONS	72
	4.1	The g	ranularity parameter	72
		4.1.1	Granularity functions	73
		4.1.2	Higher-order granularity parameters	78
	4.2	Granı	alarity-dependence and slow-motion readings	88
		4.2.1	Granularity-induced changes in durativity	88
		4.2.2	Granularity-relative temporal trace and durativity	89
		4.2.3	Constraining granularity and explaining slow-motion readings .	93
	4.3	Impli	cations of the granularity-based account	98
		4.3.1	Durativity and vagueness	98
		4.3.2	Endpoints and vagueness	100
		4.3.3	Minimal parts and granularity	103
5	Pre	LIMINA	ARY PROCESS INTERPRETATIONS	114
	5.1	Scales	s of change	114
		5.1.1	The scale of change as a series of snapshot characterisations	115
		5.1.2	The mapping to the scale of change	121
		5.1.3	Predicate-level scales	124
	5.2	Scale	structures by aspectual class	134

		5.2.1	Primary scale structures	134
		5.2.2	Differentiating types of Vendler's achievements	137
		5.2.3	Verbs and verb phrases	138
5	.3	The pr	rogressive	141
		5.3.1	Integrating a definition of the progressive into the framework	141
		5.3.2	The components of the progressive	143
		5.3.3	False positives and negatives and the role of the contrast set \ldots .	151
5	5.4	Culmi	nations and happenings in the progressive	155
		5.4.1	Deriving progressive culminations and happenings	155
		5.4.2	No partial completion	157
		5.4.3	Complements of aspectual verbs	161
5	5.5	Immir	nency and other characteristics	163
		5.5.1	Implication of imminency	164
		5.5.2	Probability of success	165
		5.5.3	Imminency instead of stages	166
		5.5.4	Modification by agentive adverbials	167
5	5.6	Beyon	d English	169
		5.6.1	The Russian imperfective and the Hindi perfective	170
		5.6.2	The restricted progressive in Hungarian	171
6 0	Con	CLUSIC	DN	174
Bibl	liog	RAPHY	4	177
App	END	DICES		188
A	ł	The m	athematical concepts used in the dissertation	188
E	3	The sc	alar trace	192
-				
Pla	GIAI	RISM D	ECLARATION	196

LIST OF FIGURES

1.1	Two sources of an event associated with a multi-valued scale of change.	14
2.1	A closed scale, e.g., [0, 1]	49
2.2	A bounded scale, e.g., $(0,1)$	49
4.1	The mappings assumed in the framework for modelling granularity	77
4.2	The return of the Sorites paradox	78
4.3	Three glasses of water	79
4.4	Granularity level	84
4.5	Overlapping granularity functions and atomic parts	84
4.6	Points distinguished at a granularity level	85
4.7	Duration switch through a granularity switch	90
4.8	γ as the granularity of the maximally precise temporal trace	102
5.1	An illustration of the scale of change	118
5.2	The cover event relation	127
5.3	Illustration of the scales of satisfaction and pre-satisfaction	133
5.4	The primary scale structure by aspectual class	137
5.5	Illustration: the progressive in a scale-based account	150

LIST OF TABLES

1.1	Aspectual classifications	3
1.2	Classifications of achievements	7
1.3	Terms used in the dissertation for subtypes of Vendler's achievements	10
2.1	Readings of progressive achievements	28
2.2	The aspectual classes differentiated in the framework of Beavers \ldots .	52
3.1	Scale complexity	71
5.1	The predicate schemata of Klein (1994)	118
5.2	Scale complexity, properties of change and aspectual classes	135

LIST OF DEFINITIONS AND PRINCIPLES

Definition 1	Incremental chain (Rothstein 2004)
Definition 2	Incremental relations (Rothstein 2004)
Definition 3	Movement Relation (MR) of Beavers (2013)
Definition 4	BECOME operator in event-semantics (von Stechow 1996) 58
Definition 5	Scale
Definition 6	Non-final subscale
Definition 7	Initial subscale
Definition 8	Minimally distinguishable interval size
Definition 9	Granularity level
Definition 10	Absolutely fine granularity
Definition 11	Temporal trace function—granularity-relative
Definition 12	Maximally precise temporal trace
Definition 13	Subdivisibility (aka divisive reference) (Krifka 1989, a.o.) 104
Principle 1	Subdivisibility down to granularity-relative extended parts 110
Definition 14	Scale of change (of an eventuality)
Principle 2	Non-degenerate scales of change imply change
Definition 15	Primary scales of change
Definition 16	Cover event
Definition 17	Secondary scales of change (of culminations)
Definition 18	Scale of satisfaction
Definition 19	Scale of pre-satisfaction
Definition 20	Telicity based on scale structure

Principle 3	Applicability condition of the progressive	146
Definition 21	The progressive in a scale-based account	149
Definition 22	The imperfective in a scale-based account	170
Definition 23	Scalar trace function	192

Chapter 1

INTRODUCTION

[B]eginnings start from somewhere (which is where whatever it is begins), and becomings turn Thing One into Thing Two by pushing it across a clearly defined boundary (the tooth was not carved, but now it is; the spider was not dead, but now it is). Unfortunately the universe doesn't work in such a simple-minded manner...

Terry Pratchett, Ian Stewart and Jack Cohen: The Science of Discworld

1.1 Aspectual classes

This dissertation is about achievements and an account of their interpretations in the progressive, as well as a novel representation of the difference between accomplishments and different kinds of achievements. Talking meaningfully about such questions presupposes fixing an ontology and terminology of aspectual classes. I will therefore first clarify where I stand with respect to the different schools of thought in the work on aspectual classification before presenting the core topic discussed and main proposals put forth in my thesis.

1.1.1 Aspectual classifications

This dissertation focuses on the categorization of achievements, and existing classifications differ as to i) whether achievements form a class different from accomplishments, and ii) whether achievements themselves have distinct subclasses. I first review aspectual classifications in general before turning to point i) in Section 1.1.2 and point ii) in Section 1.1.3.

The number of different classifications of verbal predicates into aspectual classes is immense. Nevertheless, since the work of Vendler (1957), mostly the same features recur as criteria for these classifications, with some small differences in their interpretation. Arguably the three most basic aspectual features of verbal predicates are the following (with one test of each feature given in parentheses for English):

- A basic distinction is between *statives* and *dynamic* predicates.
 - States are stative (they are admissible in the simple present on an episodic interpretation).
 - Activities, accomplishments and achievements are dynamic (they typically do not receive an episodic interpretation in the simple present).
- ♦ A second basic feature is *durativity*, distinguishing between predicates that are true at non-degenerate intervals and those which are true at instants.
 - Accomplishments and activities are durative (they typically do not combine with time point adverbials).
 - Achievements (and for many authors, states)¹ are instantaneous (they can combine with time point adverbials).
- The third distinction is between *telic* and *atelic* predicates, depending on whether or they are associated with a "natural endpoint".
 - ► *States and activities* are atelic (they can combine with a *for*-adverbial).
 - Accomplishments and achievements are telic² (they typically cannot combine with a for-adverbial unless it measures out their result state).

Table 1.1 summarizes some of the most influential (or for our purposes relevant) aspectual classifications, with the columns delineating sets of predicates that can be best equated across the classifications. Although many classifications draw on Vendler's (1957), it is important to be aware of their differences.³ Of course, it is impossible to provide an exhaustive list of such categorizations, or even do justice to those mentioned here and present the nuances thereof.

¹For instance, Vendler (1957) and Taylor (1977) regard states as true at instants, or more precisely, true at all instants of some interval, Comrie (1976) regards them as strictly durative, and Rothstein (2004) regards states as being true at both intervals and at instants of intervals at which they hold. von Stechow (2009b) holds that states are true only at instants, and there is usually a covert "throughout" adverb in sentences expressing that a state holds at an interval.

²Though, for instance, Comrie (1976) does not regard even archetypical achievements like *reach* as telic (cf. Comrie 1976, p. 47).

³For instance, the revised classification of Dowty (1979, Ch. 3) deviates much from Vendler's (1957) categorization, since "both accomplishments and achievements can be found in each of the four categories [of change of state predicates]" (p. 183). (Only two, instead of four, change of state categories are shown in Table 1.1, as I here disregard Dowty's (1979) agentive/non-agentive distinction that cross-cuts all his classes.)

Vendler (1957)	state	activity	accomplishment	achievement	L.
Kenny (1963)	state	activity	perfor	mance	
<i>Comrie</i> (1976)	state	activity	accomplishment	achievement	semelfactive
Taylor (1977)	State-verbs	Energeia-verbs	Kinesis	-verbs	-
Mourelatos (1978)	state	process	eve	ent	
			development	punctual occurre	ence
Dowty (1979, Ch. 3)	states interval ¦moment.	activity	complex change of state	single change of s	state
Carlson (1981)	dynamic ¦stative	activity	accomplishment	achievement mome	entaneous
Bach (1981)	state	process	eventered	ent instantaneous	S
	state	process	eve	ent	
Bach (1986)	dynamic ¦static		protracted	momentaneou	ns
				culmination hap	pening
Moens and Steed-	state		event		-
man (1988)		process	culminated process	culmination	point
Verkuyl (1989)	state	process	eve	ent	
Parsons (1990)	state	process	eve	ent	
			accomplishment	achievement	t
Pustejovsky (1991)	state	process	trans	ition	
Smith (1991)	state	activity	accomplishment	achievement	semelfactive
example	sit, own	walk, push a cart	walk to Boston	reach the top recognise	flash
	- () 	-		· · · · ·	-

Table 1.1: Aspectual classifications. Columns delineate the closest counterparts across classifications, though the exact boundaries among predicates differ (hence the dashed separator |). At this point, I want to stress that in this dissertation, I will focus on the differentiation between accomplishments and different subtypes of Vendler's achievements, but will disregard the details and complications relating to other aspectual classes. For example, I will mostly ignore the difference between interval (aka dynamic) states like *sit*, stage-level states like *be asleep* and object-level states like *be a mammal* (in the terminology of Dowty 1979, Ch. 3)—even though these distinctions among statives are as much relevant for semantics as the distinction between different kinds of telic predicates. Throughout the dissertation the term "state" will, unless indicated otherwise, be used refer to temporary states (dynamic and stage-level states), which are those that can combine with a time point adverbial like *at noon* (these are the *1-state predicates* of Klein 1994).

It is also important to note here that there are multiple lexical and grammatical factors contributing to the aspectual properties of a verb phrase (or sentence), most importantly, besides a verb's inherent meaning, the verb's arguments, such as the subject, the object and/or prepositional arguments. But over and above these, the aspectual class of a predicate can also be *shifted* in different contexts. As ter Meulen (1983, p. 178) put it, "we can stretch the standard use of just about any verb in a specific context". It was noted already in the seventies (cf. e.g., Comrie 1976, Steedman 1977, Dowty 1979), for instance, that all atelic verbs can receive a telic interpretation given a suitable context. "Thus if I know (and the addressee knows) that John is in the habit of swimming a specific distance every day (to prepare himself for a swimming race perhaps), then I can assert that today *John swam in an hour*" (Dowty 1979, p. 61), that is, *swim* here is used as an accomplishment rather than an activity.

1.1.2 Accomplishments and achievements

As can be seen from Table 1.1, accomplishments and achievements are collated by Taylor (1977) in his Kinesis-verbs, by Pustejovsky (1991) in his transitions, and also by Kenny (1963) in his performances.⁴ Verkuyl (1972, 1989, 1993) argued explicitly that achievements are a subset of accomplishments: namely ones with a process stage so short as to be negligible. One cornerstone of Verkuyl's argumentation is the dual-faced behaviour of achievements, i.e., that the same predicate may be instantaneous and durative, as well, given the right context. For instance, the predicate *draw a circle*

⁴However, as Mourelatos (1978) pointed out, Kenny's categorization is dubious inasmuch as the criterion Kenny proposes for performances, namely, "finish/not finish V-ing", only works for accomplishments, but not for achievements.

may be durative given a human agent, but it may also be instantaneous if the agent is a computer.

Pustejovsky (1991) also argued that given the right kind of context, achievements behave like an accomplishment in their temporal properties (for instance, when modified with a *for*-adverbial). However, Pustejovsky places the distinguishing feature between what are customarily accomplishments and achievements not in their temporal properties (i.e., the length of the preliminary process), but in agentivity:

we will argue that there is no further distinction necessary in terms of event structure for classifying these two aspectual types. Rather, achievements and accomplishments can be distinguished solely in terms of an agentive/non-agentive distinction. (p. 42)

Using agentivity as a distinguishing feature between achievements and accomplishments might not be fortunate, though: Dowty (1979, Ch. 3) already pointed out that agentivity is orthogonal to aspectual classes, arguing that there are in fact agentive and non-agentive predicates in all aspectual classes. Pustejovsky (1991) himself notes that there are sentences with achievements that seem to involve agents, such as Mary arrived at the party. But Pustejovsky argued that "agency is not intrinsically part of the [achievement] verb's meaning" (p. 43) based on sentences like The package arrived at the office and based on the infelicitousness of sentences in which intentional adverbials like deliberately modify achievement predicates like *Mary won the race deliberately. However, I still regard it as a problem for Pustejovsky's (1991) proposal that there certainly are also non-agentive accomplishments, such as The ball rolled off the table, which indicates that agentivity is irrelevant for the accomplishment/ achievement distinction. Pustejovsky's (1998) alternative distinction between achievements and accomplishments as boiling down to a difference in their focus on the initial or final subevent of the transition (in particular, a difference in what he calls event headedness) appears to me more compelling.

I argue that although there might be arguments (their felicitousness in the progressive and/or slow-motion readings) for regarding some or all of Vendler's achievements as accomplishments, the distinction between durative and non-durative telic predicates can and should be retained. The goal of my dissertation is to show how some typical durative uses of achievements can be explained in model-theoretic semantics.

1.1.3 Previous classifications of Vendler's achievements

There is no consensus regarding the ontology of non-durative predicates and the terminology used to refer to subclasses of these predicates. The term *achievement* was originally coined by Ryle (1949), whose classification, however, was not based on temporal/aspectual properties, and this term, in particular, did not correspond Vendler's (1957) class of achievements. Comrie (1976) and Smith (1991) use the term *achievement* for those of Vendler's achievements that are not semelfactives. Carlson (1981) and Dini and Bertinetto (1995) use the term *achievement* for those of Vendler's achievements which can naturally occur in the progressive without any shift (including an iterative shift) in their meaning — these correspond to what Bach (1986), who builds his ontology on Carlson's (1981), calls *culminations*. In contrast to Bach, Moens and Steedman (1988) use the term the *culminations* for those achievements of Vendler (1957) that are not semelfactives. The term *culmination* is, in addition, widely used to designate a culmination *point* rather than an aspectual class.

In Table 1.2, I compare some of the most important or for our purposes relevant subclassifications of the class of Vendler's achievements.⁵ The table is divided into three parts: the first part includes Vendler (1957) who assumed a single class of those predicates that can be characterised as non-durative and telic, using modern terminol-ogy.⁶ The second part includes authors who focused primarily on what can be called the semelfactive/non-semelfactive distinction, and the third part includes those who focussed primarily on what can be called the happening/culmination distinction.

The class of Vendler's achievements that is often assigned a category of its own is that of *semelfactives* (cf., e.g., Moens and Steedman 1988, Kearns 1991, Smith 1991).⁷ As for their characterisation, a widely (though not uniformly, cf. Filip 2011a for a discussion) accepted view today is that of Smith (1991) who regards *achievements* as telic, and *semelfactives* as atelic. This is an approach shared in essence by Moens and Steedman (1988), too, for whom the difference between different punctual eventuality predicates lies in whether or not they have a consequent state. The difference Comrie (1976) made

⁵Just as the general survey of aspectual classifications in Table 1.1 is not exhaustive, so in this case, many authors have been omitted. For instance, Mori et al. 1992 also argue for several distinctions within the class of Japanese non-durative predicates.

⁶Vendler (1957) did not use the term *telic*, yet. But for ease of exposition, I will continue to refer to received terminology even in the case of authors who did not use the relevant terms; e.g., I will use "non-semelfactive happenings of Bach (1986)", even though he never employed the term *semelfactive*.

⁷Comrie (1976) was the one to make the term widely known in aspectual discourse, borrowing it from Slavic linguistics, and he reserved the term for single-event uses of predicates like *cough*, while he used the term *iterative* for a series of coughs.

CLASSES	FEATURES	EXAMPLES					
(Vendler 1957)							
achievements	-durative, defin. time (= +telic)	reach, recognise, knock					
(Comrie 1976)							
achievements	(strictly non-extended)	reach					
semelfactives	(may be extended)	knock					
(Smith 1991)							
achievements	+telic	reach, recognise					
semelfactives	-telic	knock					
	(Moens and Steedman 1988)						
culminations	+conseq. (= +telic)	reach, recognise					
points	-conseq. (= -telic)	knock					
	(Carlson 1981)						
achievements	+point, +extended	reach					
momentaneous	+point, -extended	knock, recognise					
(Bach 1986)							
culminations	momentaneous culmination	reach					
happenings	momentaneous happening	knock, recognise					
(Filip 1993/99, Altshuler and Filip 2013)							
culminations	+temporal extent	reach					
happenings	-temporal extent						
semelfactives	full-cycle resettable	knock					
(other happenings)		recognise					
(Dini and Bertinetto 1995)							
achievements	+telic	reach					
punctuals	-telic						
event-punctuals		knock, explode					
state-punctuals		recognise					
	(Piñón 1997)						
right-boundary achievements	culmination	reach					
left-boundary achievements	inception	recognise					
(semelfactives?)		knock					

Table 1.2: Classifications and ascribed properties of achievements

between semelfactives and a predicate like *reach* is that the former describes events that are not strictly speaking punctual and may be construed as durative, while the events described by *reach* are strictly punctual. As Comrie's (1976) focus lies else-

where, it is not clear if this difference is intended to be a characterisation of semelfactives, but this distinction is not suitable for this classificatory purpose, as achievements like *recognise* or *explode* also have slow-motion durative uses like semelfactives (cf. e.g., Dini and Bertinetto 1995).

Carlson (1981), Bach (1986), Filip (1993/99) and Piñón (1997) focus on a different division among Vendler's achievements, namely, a division between those which are associated with a preliminary activity like *reach the top, die, win,* and those which are not, like *notice, recognise, explode, knock*: the former, but not the latter are natural in the progressive without any shift. Carlson (1981), Filip (1993/99) and Altshuler and Filip (2013) assume that the former (*culminations*), but not the latter, are *both* momentaneous *and* extended; Dini and Bertinetto (1995) assume that the former, but not the latter describe the right boundary of an extended event, namely, one during which a gradual change is happening. I myself will draw on the proposals of both Carlson (1981), Filip (1993/99) and Altshuler and Filip (2013) and assume that *culminations* like *arrive, die, win* are characterised by *both* gradual (extended) and non-gradual (instantaneous) change, and also on that of Piñón (1997), and assume that culminations are true of events that do not include gradual change, but are *preceded* by events of gradual change.

As will be discussed in Section 2.1, the progressive in English offers support for a tripartite subdivision of Vendler's achievements (cf. also Carlson 1981, Filip 1993/99 and Engelberg 2000 on the fine-grained distinctions among Vendler's achievements that can be made on the basis of the progressive). It supports the *semelfactive/achieve-ment* distinction of Smith (1991) inasmuch as the former but not the latter most easily receive an iterative interpretation without a plural or mass argument, as in *The light was flashing* (contrast this with *John was discovering crabgrass in his garden*). The progressive also supports the *culmination/happening* distinction of Bach (1986) inasmuch as the former, but not the latter, allow for a "preliminary process progressive", as in *John was arriving* (contrast this with the slow-motion interpretation of *The Challenger was exploding*).

1.1.4 Ontology and terminology to be assumed

1.1.4.1 Subtypes of Vendler's achievements

I assume that the following subclasses of Vendler's achievements should be distinguished minimally:⁸

- predicates like *reach* or *win*: these are the *culminations* of Bach (1986), the *achievements* of Dini and Bertinetto (1995) and the *right-boundary achievements* of Piñón (1997) these are associated with a preceding gradual change and a result state;
- ♦ predicates like *recognise* or *explode*: these are those *happenings* of Bach (1986) that are not semelfactives, the *s(tate)-punctuals* along with some of the *e(vent)-punctuals* of Dini and Bertinetto (1995), and the *left-boundary achievements* of Piñón (1997) these are associated with a preceding state and a result state;⁹

 (i) a. It's almost impossible for the human body to have survived the lack of oxygen and the cold temperature. (http://www.huffingtonpost.co.uk/2014/04/21/hawaii-flightteen_n_5184645.html)

However, unlike semelfactives, but similar to achievements, survive is not naturally iterable:

(ii) ?John was surviving heart attacks on the operation table for hours.

Despite the absence of an iterative reading, it can have an activity interpretation like semelfactives:

- (iii) a. This Guy Gave Up Food for Lent and Is Surviving On Beer Instead. (http://time.com/ 35534/man-gives-up-food-for-lent/)
 - She's surviving after the divorce. (http://dictionary.reference.com/browse/ surviving)

A similar predicate is *break a promise*, as in *Rebecca broke the promise (by never giving Jamaal any flowers)* from Engelberg (2000), who noted that breaking a promise "probably does not refer to an event at all". Such predicates might thus not even denote sets of eventualities, but might instead be second-order predicates, i.e., predicates of eventuality predicates. I will ignore such cases and focus on the three classes delineated by the semelfactive/achievement and culmination/happening distinctions.

⁹Piñón's (1997) distinction of right- and left-boundaries, however, is not sensitive enough to actually classify, say, *reach* as describing right- and *recognise* as describing left-boundary events, since events in the extension of both are both left- and right-boundaries of some eventuality (one being an event/state boundary, the other a state/state one). "Left-boundary" is especially a misnomer for predicates like *recognise*, which just signal a change from some state $\neg \phi$ to ϕ , and it is arbitrary for us to say that it's a

⁸Note that I do not preclude the existence of further classes. Indeed, there appear to be predicates which do not to fit into any of the standard aspectual classes. A case in point is *survive*, which generally behaves like a non-durative predicate, but—like semelfactives—does not imply any change from one state to another. Indeed, it implies the *absence* of an expected change (on its most basic reading, from not being dead to being dead), which is most perspicuous in cases when the argument of *survive* is some state, especially a state describing the absence of something:

♦ predicates like *knock*: these are those *happenings* of Bach (1986) that are semelfactives, the *points* of Moens and Steedman (1988), the *semelfactives* of Smith (1991), and most of the *e(vent)-punctuals* of Dini and Bertinetto (1995) — these are not boundaries (in the sense of Piñón 1997).¹⁰

1.1.4.2 Terminology

Table 1.3 provides an overview of the designation I use for subtypes of Vendler's achievements. Following Smith (1991), I will designate predicates of the first two types above as *achievements*, and predicates of the third type as *semelfactives*; following Bach (1986),¹¹ predicates of the second and third type will be called *happenings*, and predicates of the first type as *culminations*¹² or *culmination-achievements* (which I coined to be able to avoid the ambiguity that the term "culmination" introduces); combining Bach's and Smith's terminology, I call predicates of the second type (e.g., *explode*) as *non-semelfactive happenings*.

<i>examples terms for the most specific classes</i>		terms from Smith (1991)	terms from Bach (1986)
arrive, die, win	culminations, or culmination-achievements	achiovomonte	culminations
explode, recognise, notice	non-semelfactive happenings	achievements	hannonings
knock, flash	semelfactives	semelfactives	nappenings

Table 1.3: Terms used in the dissertation for subtypes of Vendler's achievements.

¹²Following Moens and Steedman (1988), the term "culmination" will also be used to refer to the pointlike change associated with an accomplishment or an achievement.

boundary of ϕ , but not $\neg \phi$; indeed, language does appear to have achievement pairs "profiling" one or the other, such as *forget–recall*, *appear–disappear*, *kill–revive*. Although the term "left-boundary" does successfully emphasize that the state characterizing these predicates is their *result state*, such a result state is also associated with "right-boundary" achievements like *arrive* or *die*.

¹⁰Alternatively, we can say that semelfactives describe punctually conceived transitions from a preparatory state to a post-state such that the two states are (type-)identical, adopting the "full-cycle" view of Talmy (1985, 1988). I prefer to retain the idea of semelfactives not being boundaries, as such a conception will lend itself easily to a uniform "binary scale of change" approach to all non-durative predicates in Section 5.2.

¹¹The reason I choose to follow the terminology of Bach (1986) rather than Piñón (1997), despite the fact that many modern-day scholars working on this distinction use Piñón's (1997) terms of left-and right-boundary achievements (cf. Heyde-Zybatow 2004, 2008, Malink 2008, Martin 2011), is that as noted in footnote 9, these latter designations might be misleading.

All non-durative predicates will be designated as *Vendler's achievements* or *achievements of Vendler*. I will use *temporal extendedness* and *durativity* interchangeably. Predicates and events that are not durative are designated as *non-durative, non-extended, punctual, instantaneous,* or *momentaneous* (that is, I will here use these terms interchangeably).¹³ I regard *temporal extendedness* as a special case of *extendedness* along some dimension, which may be temporal, spatial, or any abstract property dimension such as weight or darkness.

An eventuality predicate is *durative* if eventualities in its extension are durative. An eventuality, in turn, is *durative* (or *temporally extended*) if its temporal trace is extended, that is, it is a non-degenerate interval (an interval that is not a single point). Extendedness and durativity, as properties of individuals, are used for the way the objective temporal extent of events is *conceptualised* in a given context unless indicated otherwise, rather than as properties of their objective temporal extent, because this construal, rather than objective temporal extent is what determines the durativity feature of *predicates*.

1.2 Issues and proposal

The goal of this dissertation is an analysis of progressive achievements using a uniform semantics for the progressive that can derive the different readings that arise when applying this aspectual operator to predicates belonging to different aspectual classes. Although most accounts of the progressive (e.g., Bennett and Partee 1972, Dowty 1979, Landman 1992, Portner 1998, Bonomi 1999, Gendler Szabó 2004, Varasdi 2014) tend to focus on durative predicates, and especially accomplishments, my primary focus is on achievements.

In particular, I concentrate on two different single-event interpretations a progressive achievement can have, namely, "preliminary process" readings like *The train is arriving* and "slow-motion (camera)" readings like *The Challenger is exploding*. I maintain that in order to explain these readings, we must break with the tradition of regarding "achievements" as a homogeneous class building on Vendler (1957) and instead follow authors like Carlson (1981), Bach (1986), Smith (1991), Filip (1993/99), Dini

¹³These terms are often used synonymously. For instance, Bach (1981) calls events described by Vendler's achievements *instantaneous* events, while in Bach (1986), he uses the term *momentaneous* events. However, for instance, Beavers (2002) uses the term *momentary* and *punctual* as not synonymous. *Momentary events*, for him, hold at a single instant (these are described by momentary state predicates), while *punctual events* hold at exactly two instants (these are described by achievement predicates).

and Bertinetto (1995), Piñón (1997) who make various categorial distinctions among predicates of punctual events. Preliminary process progressives are characteristic of predicates called *culminations* by Bach (1986), while slow-motion scenarios are typical for what he calls *happenings*. These readings, in turn, differ from the regular, "internal viewpoint" progressive interpretation of accomplishments and activities.

The progressive therefore offers support for the classification of Vendler's achievements into further subclasses, in line with proposals of Carlson (1981), Bach (1986), Moens and Steedman (1988), Smith (1991), Filip (1993/99), Dini and Bertinetto (1995), Piñón (1997), Heyde-Zybatow (2008), Altshuler and Filip (2013), a.o. The subclasses are delineated by two dichotomies proposed several decades ago: *i*) semelfactives (e.g., *flash, knock, cough,* which easily receive an iterative interpretation) *versus* non-semelfactives (e.g., *recognise, explode, arrive, win, die,* for which an iterative reading is less frequent) (cf. e.g., Comrie 1976, Moens and Steedman 1988, Smith 1991), and *ii*) culminations (e.g., *arrive, die, win,* whose progressive is true during their preliminary process) *versus* happenings (e.g., *flash, knock, cough, recognise, explode,* which are not normally felicitous in the progressive on a single-event reading, and require a slowmotion scenario to this end) (cf. e.g., Carlson 1981, Bach 1986, Filip 1993/99, Piñón 1997).

Since the progressive is most often assumed to be a partitive operator, and achievements are generally taken to lack non-trivial part structures (following the original suggestion in Vendler 1957), an analysis of "preliminary process" progressives typically assumes either a non-uniform semantics of the progressive operator (e.g., Kearns 2003), or type-shifted/coerced durative achievements in the scope of the progressive (e.g., Rothstein 2004). Given different problems faced by these and other strategies that I will discuss, I propose a new analysis within a scale-based framework.

I will work out and employ a scale-based framework on which all predicates (with the exception of individual-level predicates) are characterised on the basis of a scale of change that captures the change associated with events in the denotation of a predicate. Crucially, the concept of the scale of change is a more general one than is common in existing scale-based approaches to aspect (Hay et al. 1999, Kratzer 2004, Filip and Rothstein 2006, Kennedy and Levin 2008, Rappaport Hovav 2008, Piñón 2008a, Beavers 2013, a.o.), and is simply a series of state extensions. My notion of a scale of change is an integration of the scale of change concept of Beavers (2002, 2008, 2013), and Asher's (1992) notion of a *perspective* which is a set of propositions describing an aspect of a situation that is relevant in a given context. A scale of change on my account is a temporally decomposed version of Asher's perspective in that a scale of change describes the contextually relevant aspects of *snapshots* of an event. This means that scales of change, just as Asher's (1992) perspectives, are primarily associated with events, and the scales of change associated with predicates are derived from these event-level scales: the scales of change of a predicate are the scales of change of all possible events of which the predicate holds.

A further radical departure from former scale-based approaches to aspect is that I will assume that some predicates can be associated with *two* kinds of scales of change. In particular, I will argue that culmination predicates like *arrive* are characterised by both binary and multi-valued scales of change. This is will be crucial for my account of preliminary process progressives. As noted by a number of researchers (Piñón 1997, Löbner 2002, Heyde-Zybatow 2008, Abusch 2010, Martin 2011, a.o.), culminations *presuppose* the occurrence of a prior gradual change: e.g., *arrive at the station* presupposes movement toward the station. This presupposed gradual change, I will argue, is of a special kind, namely, a gradual change that *elaborates* on the non-gradual change associated with the culmination (for *arrive at the station*, a change from not being at the station). To represent this presupposition, I introduce the concept of a *secondary scale of change* associated with a culmination predicate like *arrive*, which characterises this special kind of gradual change that is presupposed by the culmination — while its primary scale of change characterises the non-gradual change whose occurrence is part of the asserted content of the predicate.

Secondary scales thus do not describe properties inherent in events in the extension of culmination predicates, but *relational* properties, namely, the properties of events that I call *cover events*, whose occurrence is presupposed by the predicate. Informally, the cover events of a culmination-achievement (e.g., *arrive at the station*) are events (e.g., walking to the station-events, driving to the station-events etc.) whose culminations belong to the extension of the culmination-achievement. The secondary scales of a culmination predicate like *arrive* are not the scales of the events in its extension, but the scales of all its cover events.

I will then propose a scale-based definition of the progressive that integrates the progressive theory of Varasdi (2014) into my scale-based framework. On his account, the progressive of a predicate *P* holds of an event just in case the event's properties are indicative of only *P* of a contextually relevant set of predicates, or more specifically, just in case the partial event can only be a part of a *P*-event out of the available alternatives based on its inherent properties. The crucial innovation of my account is that the progressive involves a partitive component over not events or event runtimes, but over scales of change: roughly, on my scale-based account, the progressive of a pred-

icate *P* holds of some event if the scale of change of the event is a non-final proper part of the scale of change of some (possible) *P*-event, and only a *P*-event out of a contextually given set of alternative predicates.

Given the change from partitivity over events (or event runtimes) to partitivity over scales, the applicability condition of the progressive also changes from a durative event predicate (that is, a predicate denoting a set of durative events) to a predicate with a multi-valued scale of change — that is, a progressive can only apply to a predicate P if P is associated with multi-valued scales. These two applicability conditions differ just in case the scales of change of predicates and events in the extension of predicates are teased apart, which is what I argue for in the case of culminations. Then, rather than coercing one aspectual type into another one, what happens in the case of progressive achievements is that instead of the *two-valued primary scale* associated with an achievement, the aspectual operator has access to a multi-valued scale.

Given the way the formal system is set up, a multi-valued scale can only be associated with a durative event—just as in Beavers (2002, 2008). Thus, the multi-valued scale satisfying the applicability condition of the progressive needs to be associated with a durative event. My proposal is that in the case of culmination achievements like *arrive*, this durative event is a cover event, while in the case of happenings like *explode*, this durative event is an event in the extension of the happening itself, viewed at a fine granularity. Figure 1.1 illustrates these two sources of a multi-valued scale in the case of different achievements.



Figure 1.1: Two sources of an event associated with a multi-valued scale of change.

In the case of culminations, the applicability condition of the progressive is satisfied by the scale associated with a cover event rather than the scale of an event in the extension of the culmination predicate: this results in a preliminary process reading, the account of which builds on the idea of culminations like *arrive* being characterised by both two-valued primary scales (describing the non-gradual change asserted by the predicate) and by multi-valued secondary scales (describing the gradual change presupposed by the predicate). This dual characterisation encodes the observation made explicitly by Filip (1993/99) and Altshuler and Filip (2013) that some aspects of the linguistic behaviour of culmination predicates are similar to that of happenings and some other aspects of their behaviour are similar to that of accomplishments.

In the case of happenings like *explode*, the multi-valued scale is associated with an event in the extension of *explode* itself. This event may be conceived of as instantaneous in an everyday context, but is construed as durative in the "fine (temporal) granularity" discourse context of the progressive, in which minute temporal differences are conversationally relevant. This results in a slow-motion reading of the progressive. Several researchers have noted that an extended event in one context can count as instantaneous in another context, but so far, no formally precise analysis of this observation has been put forth, as Landman (1991, §3.3) remarked. I will offer such a formal analysis by employing and refining the framework of granularity functions of Sauerland and Stateva (2007, 2011) (who themselves build on Krifka 2007), developed originally for an account of a subset of the vague predicates. This analysis will be shown to have relevance and applicability outside of achievements, and opens up several lines of potential future research (for example, on the minimal parts problem and on vague event boundaries).

The aim of the dissertation is a comprehensive account of the set of observations that have been made about single-event progressive achievement readings in English. As such, its goal is theoretical rather than empirical, though some novel empirical observations will be made with respect to the behaviour of progressive culminations. But the line of investigation is not extended to a systematic survey of a wide range of different achievement predicates: instead, I restrict the discussion to those typical achievement predicates that have formed the centre of debate on progressive achievements.

An important idea inherent in the approach of the thesis is to take formal characterisations to their logical conclusion. The motivation behind this is that if the full strength of a formal tool introduced into semantics (e.g., scales) is not utilized in the analysis of language (e.g., only scales with an infinite number of degrees are used), then that formal tool might not be the ideal one for language modelling, because it overgenerates. I propose that we should exploit all that a given formal tool can give us, even extreme cases satisfying the formal definitions, and use the *extreme* cases to explain *extremely* different linguistic behaviour. For example, I will take the notion of a scale as a linearly ordered set quite seriously, which leads us to not only binary scales as in Beavers (2008, 2013), but to scales consisting of a single degree (and perhaps even scales consisting of no degrees, as the emptyset \emptyset is also a linearly ordered set), which satisfies the formal definition of a scale.

A consequence of this approach, however, is that we often arrive at results that at first sight may appear unintuitive given the typical use of the relevant concepts in semantic analysis. For example, state predicates will also be characterised by "scales of change" — but because these scales only contain a single degree, what is entailed is *no* change. Thus, the scale of change encodes the change profile of a predicate, including the difference between predicates which entail some change and those which do not. Perhaps it would be better to use a wholly different term for this scale of change concept, such as *change profile* or *temporal perspective*, but in doing so it would not be apparent that this concept is a generalization of the notion of a scale as used in previous scale-based aspectual works.

Despite the exploitation of extreme formal cases, virtually all components of the analysis I will propose have their predecessors in formal semantics, and I will aim to motivate the introduction of each component and link it to related ideas in the literature. Indeed, in the analysis of progressive achievements I propose, I aimed to incorporate the insights of researchers working in several different frameworks which I believe is necessary in order to account for the complex patterns of behaviour showed by achievements in the progressive.

1.3 Structure of the dissertation

The structure of the dissertation is as follows. In Chapter 2, I will review the main data that I aim to account for, and the previous accounts that I draw on. I first categorise the readings available to achievements in the progressive, and then review the proposed analyses of progressive achievements. I will pinpoint aspects of each previous account of progressive achievements that are problematic and those which are useful and should be incorporated into an account of these progressives. I conclude that many of the problems faced by earlier analyses are rooted in assuming that the progressive involves partitivity over events, and therefore is allowed to apply to durative predicates only. The final part of the chapter is a review of former scale-based approaches to aspectual phenomena, the goal of which is again the identification of those aspects of different proposals that will be incorporated in my own scale-based framework.

Chapter 3 introduces the basic assumptions and considerations that I will use in my framework and analysis. I first review the relevant issues and insights relating to durativity and the moment of change. I conclude that although there may be reasons to suppose that there are no objectively instantaneous dynamic events, an event individual can be conceptualised as durative or instantaneous, and this conceptualisation affects whether a predicate used to describe the event is durative or not. I then discuss the existential presupposition about the occurrence of an activity that has been associated with culminations like *arrive* and *win*, and conclude that this presupposition is much weaker than generally assumed, but is highly important in the characterisation of this set of predicates. The chapter will be concluded by the presentation of the basic components of the model that I will use in my analysis. The special components of the formal semantic framework I use are a granularity and a context parameter of interpretation, and scales, for which I provide a detailed formal characterisation that aims to be as general as possible in order to be compatible with all previous scale-based approaches.

The main innovative contributions of the dissertation are in Chapters 4 and 5, containing an analysis of slow-motion and preliminary process readings, respectively.

Chapter 4 is dedicated to a formal analysis of how an achievement predicate in the sense of Vendler can be durative in a slow-motion scenario (and therefore, this chapter will focus primarily on the happenings of Bach 1986). I present first in detail the components necessary for analysing granularity-sensitive phenomena, building on the granularity function parameter of interpretation of Sauerland and Stateva (2007, 2011). I then show how changes in durativity (i.e., changes in the conceptualization of eventualities as instantaneous or durative), and in particular, slow-motion readings, can be accounted for by relativising the temporal trace function to a temporal granularity parameter (a granularity function over time as a scale). The results of this chapter can be directly exploited in event semantic frameworks with an event mereological approach to the progressive (since the proposed analysis only exploits the granularity parameter of the framework and so reference to scales of change are not necessary). Predictions and consequences of this account will be discussed, such as the link between durativity and vagueness, as well as the welcome theoretical gain in being able to model the vagueness of the temporal boundaries of events, and to approach the minimal parts question of activities within a more general framework of granularity.

In Chapter 5, I present a novel scale-based account of aspectual classes and the progressive in order to be able to account for not only slow-motion progressive readings, but also preliminary process interpretations. I first introduce a novel notion of a scale of change, integrating ideas from Beavers (2008, 2013) and Asher (1992), and the novel notions of a cover event and secondary scales of change, building on the existential presupposition associated with culminations. Then I present a characterisation of predicates belonging to different aspectual classes based on their associated scale structure. I follow Beavers (2013) in having two properties of scales of change influence aspectual properties of their associated predicates: scale complexity determines durativity and specificity of the required change determines telicity. I concentrate mainly on durativity, and extend to states Beavers's (2008) idea of how scale complexity determines durativity by determining the gradualness of the associated change, and, more pertinently for the purposes of the dissertation, I characterise culminations with both two-valued and multi-valued scales.

I then formulate a scale-based definition of the progressive operator by adapting Varasdi's (2014) account to my scale-based framework. On my analysis, the input requirement of the progressive is a predicate associated with a multi-valued scale of change rather than a durative predicate (where a durative predicate is a predicate whose extension contains durative events). The chapter will be concluded by a brief look at aspect in some languages other than English to show that previous accounts of typological variations can be incorporated into the proposed scale-based framework. In addition, I will show that by changing another parameter specific to the scale-based definition of the progressive, namely, whether or not the progressive has access to secondary scales of change, we can account for a difference between the English and the more restricted Hungarian progressive.

In Appendix A, I give the notation and definition of the most important mathematical concepts I use. In Appendix B, I give a detailed characterisation of the scalar trace function that maps time intervals to parts of the scale of change of an eventuality the formal details of this are not directly relevant to my analysis, but serves to ensure that the scale-based system works in the expected way formally.

The dissertation is organized around the following topics:

 Culminations and preliminary process progressives (which is the single-event progressive achievement reading discussed most frequently in former accounts): this forms the topic of Sections 2.1, 2.2 and 3.2 and Chapter 5.

The analysis of these progressives is built on Gyarmathy (2011, 2012, 2014).

Durativity, granularity changes and slow-motion readings: These are the focus of Section 3.1 and Chapter 4.

The discussion in these chapters is built on Gyarmathy (2012, 2015).

♦ A scale-based approach to aspectual phenomena, and in particular, the characterisation of aspectual classes and the semantics of the progressive operator: this is the topic of Sections 2.3, 5.1, 5.2 and 5.3.

The ideas in these parts are built on Gyarmathy (2011, 2014).

The overall encompassing theme of the dissertation is the exploitation of scalebased approaches to time and aspect in accounting for the behaviour of progressive achievements. The analysis of slow-motion readings in Chapter 4 relies on conceiving of time as a scale, while the analysis of preliminary progressives in Chapter 5 on conceiving of the change associated with all predicates as a scale.

Chapter 2

THE CORE DATA AND A REVIEW OF THE RELEVANT THEORIES

This chapter presents the data about progressive achievements that will serve as the object of study in the rest of the thesis. First, different uses of the progressive when applied to Vendler's achievements will be charted and the slow-motion and preliminary process interpretations isolated as specific to achievements. Second, I will review former accounts of progressive achievements with the aim of uncovering their respective weak points and advantages that I will incorporate in my own analysis. Finally, the chapter is concluded with an overview of different scale-based approaches to verb semantics with the aim of again selecting aspects of the various analyses to be incorporated in my own analysis.

2.1 Vendler's achievements in the scope of the progressive

2.1.1 The progressive as an aspectual operator

The imperfective, the progressive and the perfective, as well as the perfect according to some authors, is a so called *viewpoint aspect* (aka *grammatical aspect*), a perspective on states of affairs. They are often, though far from exclusively, signalled with morphosyntactic means; for instance, the English progressive is signalled with the auxiliary *be* plus the *-ing* morpheme. It is not at all trivial to compare aspectual operators and systems across different languages, as detailed by Dahl (1985) and Thieroff

(2000), a.o.¹ For instance, English, and, in fact, Germanic languages in general, are often argued not to have a perfective/imperfective distinction (cf. Dahl 1985, p. 167). Although the distinction between the simple past (or, in general, the non-progressive form) and the progressive in English is often analysed as a perfective/imperfective difference (Comrie 1976, p. 23), the English non-progressive form can be both perfective *and* imperfective (Comrie 1976, p. 25). Although the term "progressive" is often used in works on English aspect interchangeably with "imperfective", the two terms should be kept distinct even in an analysis of English. An example for an imperfective but not progressive sentence is *John is in the kitchen*, for instance.

Viewpoint aspect interacts with aspectual classes (aka *situation aspect*, or *lexical aspect*): viewpoint aspectual operators operate on an eventuality predicate to output another eventuality predicate.² The output of the progressive is an atelic predicate — a state according to, e.g., Vlach (1981), Bach (1981), Moens and Steedman (1988), or an activity according to, e.g., Mourelatos (1978), Kearns (1991). The semantics of the progressive should then ideally reflect the constraints on the aspectual class of its input predicate and account for the aspectual class of its output.

The most common approach to the semantics of the progressive, going back to at least Jespersen's (1931)'s idea of the progressive as describing an event that serves as a "temporal frame" encompassing another event, is that the progressive presents an eventuality "from within" (cf. e.g., Comrie 1976). This is reflected in analyses of the progressive as a partitive operator over time intervals (e.g., Bennett and Partee 1972, Dowty 1979) or events (e.g., Bach 1986, Krifka 1992, Filip 1993/99). Given this partitive view of the progressive and a view of Vendler's achievements as non-durative (both of which views can, and have been called into question, as we will see), an incompatibility of the progressive with Vendler's achievements is expected, since they denote events without proper parts, which cannot be "viewed from within". Whether this expectation is borne out is the topic of the following section.

¹Despite problems of crosslinguistic identification of aspectual operators, it is customary to utilize the same terms in the description of the aspectual system of different languages, if only to account for semantic similarities. The basic aspectual distinction (originating from Slavic linguistics) is assumed to be the perfective/imperfective, and the progressive and the habitual are generally assumed to be a subtype of the imperfective (cf. e.g., Comrie 1976, Thieroff 2000), while the perfect is often regarded as a special type of the perfective (cf. e.g., Bertinetto et al. 2000).

²Of course, not all authors assume this view of viewpoint aspect, or even a distinction between viewpoint and situation aspect and/or viewpoint aspect and tense (cf. for instance the framework of Reichenbach 1947 who analysed both tense and viewpoint aspect as tenses), but a comparison of the various approaches is beyond our scope here.

2.1.2 Readings of the progressive of Vendler's achievements

Achievements, since the work of Vendler (1957), are often regarded as unacceptable in the progressive based on the infelicitousness of examples such as (2.1) from Dowty (1979).³

(2.1) ?(At this moment) John is noticing a stranger in the room.

At the same time, it is a well-known fact that many achievements do appear in the progressive, as the following examples from Dowty (1979) show:

- (2.2) a. John was dying.
 - b. John is gradually realizing that you are right.
 - c. John has been discovering crabgrass in his yard for six weeks.

Thus, the adage that achievements "lack progressive tenses" (Dowty 1979, p. 54) is well-known to be a simplified generalisation, as "there are at least occasional acceptable examples of achievements with progressives" (Dowty 1979, p. 137). A question that arises then is which achievement predicates can easily used in the progressive, and authors like Carlson (1981), Filip (1993/99), Dini and Bertinetto (1995) have argued that it is those achievements which are associated with an extended gradual change—*culminations* in the terminology of Bach (1986).

An important observation about progressive achievements is that "these exhibit the same failure of inference from progressive to simple tense as do accomplishments" (Dowty 1979, p. 137). In other words, achievements are generally assumed to display the *imperfective paradox* just like accomplishments. For example, (2.3b) does not follow from (2.3a).

- (2.3) a. John was dying.
 - b. John died.

³Note that different authors take different sets of predicates to be achievements, which adds to the confusion surrounding progressive achievements. Many of the authors discussed here, including Moens and Steedman (1988), Piñón (1997), Rothstein (2004), Martin (2011) do not include semelfactives in their class of achievements, but may be in disagreement about less well-established distinctions, such as the status of predicates like *explode* or *kill*. I will continue to adhere to the current convention of using the term *achievement* for all event predicates which are generally non-durative in their behaviour except for semelfactives (without, however, wishing to commit to the assumption that these form a single, unified category).

Note that this generalisation is only valid only for the natural progressive interpretation of culmination predicates like *die* and *arrive*, but not if we take into account all of the progressive readings available to Vendler's achievements, which is what I will turn to now.

All of the readings I list below have been recognised and noted in previous works, but they have not, to my knowledge, been systematically categorised and characterised. Most researchers have tended to focus on the difference between the interpretation of the progressive when applied to accomplishments as opposed to culmination-achievements like *reach* (e.g., Kearns 2003, Rothstein 2004, Martin 2011). There is little attention given to other readings of progressive achievements, and apart from Filip (1993/99), Dini and Bertinetto (1995) and Engelberg (2000), researchers have given little or no attention to delineating subtypes of Vendler's achievements that give rise to one or another reading.

Building on observations of Carlson (1981), Filip (1993/99), Dini and Bertinetto (1995), Engelberg (2000), Kearns (2003) and Rothstein (2004), I propose that the main types of interpretation that are available to Vendler's achievements in the progressive in English are as follows,⁴ with examples from, among others, Comrie (1976), Dowty (1979), Verkuyl (1989), Piñón (1997), Rothstein (2004).

1. Preliminary process reading

This is the *preliminary stage*-focusing reading of Smith (1991), the *preliminary circumstance* reading of Kearns (2003), the *preparatory phase of result state* interpretation of Dini and Bertinetto (1995), and the most prominent use of progressive achievements as analysed by Piñón (1997), Rothstein (2004), Martin (2011).

These sentences display the most typical use of the progressive in the case of the non-semelfactive achievements of Vendler. They often imply the imminency of the culmination⁵ and display the imperfective paradox. Examples are as follows:

- (2.4) a. Rebecca was reaching the summit when it began to rain.
 - b. Astrid was winning the race when we arrived.
 - c. We visited the wounded soldier, who was dying.

⁴Here I only focus on aspectual-temporal uses of the progressive, and disregard, for example the "interpret(at)ive progressive" (cf. e.g. Bertinetto et al. 2000), which Kearns (2003)'s example, *In sud-denly uncovering and seeing the key he was looking for, he was finding it,* shows is also available for achievements.

⁵The imminency meaning component of such progressives has been noted for English and several Romance languages, cf. Bertinetto (2000), as well as several Germanic languages, cf. Ebert (2000).

Not all achievements can be used in this sense, as witnessed by the unacceptability of the following sentence:

(2.5) #Anita was recognising Peter when I walked in.

I follow Carlson (1981), Filip (1993/99), Dini and Bertinetto (1995), Martin (2011) in maintaining that it is *culmination-achievements* that can appear on this progressive reading.

2. Slow-motion (camera) reading

This is the *momentaneous progressive reading* of Dini and Bertinetto (1995), as well as the *slow-motion movie* reading noted by Filip (1993/99, pp. 18, 114), and the *slowed down* scenario mentioned by Comrie (1976, p. 43).

On this reading, an event described by the achievement predicate, which by default is regarded as durationless, behaves like durative eventualities, that is, we "zoom in" to view it as an extended event. Then the predicate describing this event behaves like an accomplishment, and is therefore felicitous in the progressive, and displays the imperfective paradox. Examples include the following, with progressively more radical changes in granularity from an everyday context to an extreme scientific one:⁶

- (2.6) a. John is gradually realizing that you are right.
 - b. Look at the screen, the Challenger is exploding now.
 - c. And now the subject is coughing. [during a film in an anatomy lecture]

As Comrie (1976, p. 43) argues, this reading is only available to those predicates which describe events that in reality are not instantaneous, but take (some small amount of) time — which is true of *happenings* (i.e., semelfactives and predicates like *recognise*). It is typically (though not by necessity) unavailable for predicates like *reach* (*culminations*), which, however much we "zoom in", could always be construed in a way that they refer to only the very final moment of some gradual change.

⁶(2.6a) is not strictly speaking a *slow-motion* reading, as it does not require an atypical, fine-grained context like (2.6b) or (2.6c). I include (2.6a) here to indicate that I believe all of these progressive sentences should be grouped together, the reason for which is that these are the cases in which an achievement predicate is genuinely used as an *accomplishment* predicate. (2.6a) will be analysed in detail in Section 5.4.1.
3. Iterative reading

The iterative reading of the progressive is a well-known, multi-event (i.e., *plurac-tional*, cf. Bertinetto and Lenci 2012) interpretation. Examples for this reading are as follows:

- (2.7) a. The light was flashing.
 - b. John was coughing.

On the iterative reading, we typically do not find the imperfective paradox, that is, from *The light is flashing*, we can usually infer that *The light flashed*. As the examples show, this is the most natural progressive interpretation for *semelfac-tives*. This reading is not specific to this aspectual class,⁷ but it is easy to explain why the iterative reading is most natural with semelfactives. An iterative interpretation is dependent on the *relative size* of two intervals: i) the size of the interval of evaluation and ii) the interval size which is the sum of the typical time an event of the relevant type takes and the time needed to "reset" the state of the affairs so that an event of the same type with the same participants can reoccur again (cf. Filip 1993/99, p. 114–5 and Vanden Wyngaerd 2001). Since, for example, a flash can reoccur as soon as one iteration is over, flashings easily fit multiple times into any reasonably-sized interval in everyday contexts.

This is reinforced by the fact that if the meaning of an achievement is shifted so that its result state is implied to hold for only a very short time, then an iterative reading becomes available to them, as in Beavers's (2008) example:

(2.8) [In a context of Nancy suffering from acute amnesia:] ?Sid's Mohawk will stun Nancy over and over again for five minutes.

But felicitous examples of an iterative reading of a non-semelfactive predicate in the progressive may also arise simply if the relative size of the interval of evaluation is great enough, as in (2.9a) from Vlach (1981) for the culmination predicate *win*, or as in (2.9b) (adapted from Vanden Wyngaerd 2001) for an accomplishment predicate.

- (2.9) a. Max was winning for a year.
 - b. Fred has been swimming across the pool for the past 3 hours.

⁷Cf. Carlson (1981), Vlach (1981), Vanden Wyngaerd (2001). For example, (2.9b) exemplifies a progressive accomplishment that can have an iterative reading in a neutral context.

4. Futurate progressive reading

In English (and in a number of other languages such as Icelandic and Judeo-Spanish, according to Bertinetto 2000), the progressive may take on a futurate meaning (e.g., *I am writing the letter to the dean tomorrow*) as opposed to a from-within-the-eventuality viewpoint. This is a reading that is, again, not specific to achievements, but is available to them, as (2.10) shows.

(2.10) Mary's plane is arriving tomorrow morning.

Although this reading is similar to the "preliminary process" reading in that it is a *prospective* interpretation, it is different from that reading. First, the futurate reading has *no* imminency implication, that is, (2.10) is not synonymous with (2.11).

(2.11) Mary's plane is about to arrive.

Second, as noted above, while the preliminary process reading is specific to achievements (for no progressive accomplishment or activity that is not modified with a time adverbial do we get an imminency implication), the futurate reading arises for other aspectual classes, as well (cf. *I am writing the letter to the dean tomorrow* for an accomplishment, or *I am running tomorrow morning* for an activity). Third, the futurate progressive, but not any other reading of the progressive, is known to imply that an eventuality of the relevant kind is planned or "predetermined" (as noted by, e.g., Dowty 1977, 1979). This predeterminancy meaning component explains why some predicates, semelfactives among them, are odd on the futurate reading in a neutral context, as (2.12) shows.⁸

(2.12) ?The light is flashing tomorrow morning.

Crucially, while the preliminary process reading is available to all culminations, the futurate reading is not naturally available for culminations like *win*, which, like semelfactives, are not generally compatible with their occurrence being predetermined:

(2.13) ?John is winning the race tomorrow morning.

⁸Contrast (2.12) with *The light is flashing in a minute,* which appears to sound much better, arguably because a pattern of flashing in which individual flashes are separated by a few minutes, as opposed to days, is much easier to assume.

5. Commentary reading

Kearns (1991) calls this use *anecdotal*, while Rothstein (2004) calls it a *slow-motion* or *film-commentary* reading.

This reading is available, for instance, when the speaker is conveying the happenings in a movie to a hearer in the next room. An example for this reading is the following, which can be uttered closely before *or after* the time of Mary's actual spotting of her arch enemy:

(2.14) Mary is spotting her arch enemy at the party at the moment.

Importantly, Rothstein (2004) argued that this reading does not display the imperfective paradox. The reason for this is, intuitively, that this reading is used when "exactly timing the utterance is not important" (Kearns 1991, p. 168).

This reading, again, appears not to be restricted to achievements: *Mary is pouring wine into the glass,* for instance, can be uttered even if, on the film that we describe, Mary has just finished pouring.

I have not included in the list above a further felicitous case of progressive achievements, namely, the case when the predicate has an argument that is a plural or a mass term. This is another multi-event, pluractional use (beside the iterative) on which achievement predicates are acceptable in the scope of the progressive, as noted by Dowty (1979) and Filip (1993/99), a.o. It is often not differentiated from the iterative reading, which it should be, because, as opposed to the iterative reading, this is not specific to semelfactives in the sense that a plural argument reading is *naturally* available for other types of Vendler's achievements, as well, without any special conditions. Examples include the following:

- (2.15) a. Trains were arriving at the station the whole afternoon.
 - b. John has been discovering crabgrass in his yard for six weeks.

The reason I did not include this reading in the list above is that the input predicate of the progressive is here not in the same form as in the case of the other readings. In any case, just as the iterative reading, this progressive use is not specific to Vendler's achievements, cf. *John was building houses during the last months*.

What we see is that there are different ways in which Vendler's achievements can interact with the progressive, and their most natural progressive readings reveal cat-

egorial differences among them. Table 2.1 summarizes the most neutral use of the progressive as applied to different subtypes of Vendler's achievements.

CLASS	EXAMPLE	PROGRESSIVE READING
culminations	reach, die	preliminary process
non-semelfactive happenings	recognise, explode	slow-motion
semelfactives	<i>jump, flash</i>	iterative (also slow-motion)

Table 2.1: The most natural reading of the progressive for the three subtypes of Vendler's achievements

Most of the readings available to Vendler's achievements are not particular to this class and should thus be analysed within a broader theory of how these interpretations of the progressive arise. This leaves the first two readings (the preliminary process and slow-motion interpretations) as specific to this class, and these cases exhibit characteristics similar to progressive accomplishments, at least in displaying the imperfective paradox. At the same time, both these readings differ from the most natural reading of progressive accomplishments, which is what Kearns 2003 aptly calls the "component process" reading: a progressive accomplishment like John is writing a letter is true of the process part of a complete (possible) event of John writing the letter completely. A preliminary process progressive like John is arriving is true of a process that is not generally regarded to be a component part of an event of which the input predicate (John $arrive_{\emptyset}$) holds (though see Section 2.2.2 for a view on which it does). A slow-motion progressive like The Challenger is exploding does hold of a component process, but it requires a "fine granularity" discourse context which a progressive accomplishment does not. In the rest of the dissertation, it will be these two readings of the progressive of Vendler's achievements that I will focus on.

2.2 Previous analyses of progressive achievements

2.2.1 Charting existing approaches

An analysis of progressive achievements has both some differences and some similarities among aspectual classes to account for. First, it should ideally employ a semantics for the progressive that is invariant across aspectual classes (activities, accomplishments, culminations, and happenings alike), the core of which is the "internal view-point" meaning component.

Second, the analysis should be able to capture and explain the difference between *culminations* and *accomplishments* in the progressive. Differences in behaviour involve (cf. e.g., Rothstein 2004, Martin 2011, and Chapter 5):

- infelicitous modification by adverbs of completion such as halfway through,
- non-acceptability as complements of aspectual verbs, such as start,
- different interpretation (imminency or fine granularity),
- non-implication to the existence of a "partial event" of the type denoted by the predicate input to the progressive.

Third, the analysis should also capture and explain the differences in the interpretation *across* groups of achievements in the progressive, as seen in Section 2.1.2, predicting the relevant single-event readings:

- preliminary process reading (Rebecca was reaching the summit);
- ♦ slow-motion reading (The Challenger is exploding now).

Most analyses of progressive achievements have focussed on the preliminary process reading characteristic of culmination-achievements, because this is the single-event progressive interpretation that is available for achievements in normal, everyday scenarios. Slow-motion readings like *Look at the screen, the Challenger is exploding* characteristic of happenings, in contrast, require fine-grained scenarios that are less common in everyday contexts.

And fourth, the analysis should account for the common meaning core within each of the above groups of progressive achievement.

There are several different approaches in previous works to satisfying some or all of the above desiderata. Existing accounts of progressive achievements can be categorised according to a handful of criteria which characterise the basic ontology and semantics they assume.

Progressive as a partitive operator. The first potential dividing line concerns the semantics of the progressive. In particular, theories could differ as to whether the semantics of the progressive involves partitivity (over the domain of time intervals or eventualities) or not.

1. To this question, most approaches have an affirmative answer, that is, the progressive is assumed to involve partitivity in Bennett and Partee (1972), Dowty (1979), Moens and Steedman (1988), Landman (1992), Filip (1993/99), Portner (1998), Bonomi (1999), Kearns (2003), Deo (2009), Varasdi (2014), alike.

2. Exception to this strictly partitive account are Vlach (1981), Asher (1992) and Kamp and Reyle (1993), who put forth similar suggestions, namely, that the progressive is true during the process that *leads* to truth of its input predicate.

The analysis I will propose aims at a unification of the insights of both approaches: following the latter authors, I will assume that the progressive is not a partitive operator over time intervals or eventualities, but I will embrace the partitive approach in assuming that the progressive involves partitivity over *scales of change*. Such a move makes a difference just in case we tease apart scales of change and eventualities in the extension of a predicate, which is exactly what I will propose in the case of culminations.

Two progressives. A point of divergence among different approaches to progressive achievements is whether the semantics of the progressive is different when applied to accomplishments and (culmination-)achievements.

- 1. Some authors would like to retain a uniform semantics for the progressive and assume that the semantics of the operator is the same whatever predicate it applies to. This is the approach of Moens and Steedman (1988), Filip (1993/99) and Rothstein (2004). I myself will also endorse this view.
- 2. Some authors, on the other hand, posit different semantics of the progressive operator when applied to achievements and accomplishments. This is the view held by Kearns (2003) and Martin (2011). These authors assume that while the progressive refers to the component process of accomplishments (being a partitive operator by default), it refers to the preliminary process of culmination-achievements. Piñón (1997) should perhaps also be listed among these authors, though his case is unique in that the progressive operator itself does not have different semantics in different cases, but because he treats progressive achievements as complex lexicalized expressions, Piñón also does not derive the semantics thereof via a combination of the meaning of achievement predicates and the generic progressive operator.

Coercion. A third point of difference among theories of progressive achievements is whether or not a (culmination-)achievement is assumed to be coerced to describe durative events in the scope of the progressive.

- 1. Perhaps the most traditionally accepted view is that progressive achievements do involve coercion into a durative type. This is the view expounded by Moens and Steedman (1988), Rothstein (2004) and Martin (2011).
- 2. Some authors, however, do not assume an instantaneous to durative coercion in the case of progressive culminations. This is the view held by Filip (1993/99), Altshuler and Filip (2013) and Kearns (2003). Piñón (1997) is also an author to reject coercion in prohibiting the application of the progressive to achievements. I myself will likewise embrace the view that no instantaneous to durative coercion should be posited for culminations in the scope of the progressive.

Achievement types. A fourth point where approaches to progressive achievements (just as aspectual classifications, in general) differ is the distinctions they make among achievements, and, in particular, whether a distinction between culminations and happenings is assumed.

- 1. Approaches building on Vendler (1957) do not make such a fine-grained ontological distinction. Authors subscribing to this view are Dowty (1979), Moens and Steedman (1988), Kearns (2003), Rothstein (2004).
- Most authors working on the aspectual characteristics of achievements, however, argue for a need for a distinction between culminations and happenings. Proponents of this view are Carlson (1981), Filip (1993/99), Piñón (1997), Martin (2011). I myself also endorse this view.

Durative input requirement. Based on these considerations, if one goes along with the most received semantics of the progressive as a partitive operator and of achievements as non-durative, then there are perhaps two basic approaches to resolving the clash between the non-durativity of achievements and the durativity requirement imposed on its input predicate by the progressive:

- 1. The progressive that applies to achievements has a special semantics, in which case it either does not involve partitivity, or it accesses the presupposed (rather than asserted) content.⁹
- 2. The achievement in the scope of the progressive is coerced to be durative.

Martin (2011) is an author who appears to endorse both solutions, that is, beside positing durative achievements, she also concludes that achievements are not compatible with the "standard progressive". The reason she subscribes to both assumptions is that she allows for durative achievements also outside the scope of a progressive operator, based on an inspection of (French) data with aspectual verbs and adverbs of completion, and so she needs to explain why progressive achievements have a preliminary process, rather than component process interpretation, meaning that they do not entail the partial realization of a (durative) achievement (e.g., no partial arrival is entailed by *be arriving*).

2.2.2 Culminations entailing a change with temporal extent

In contrast, Filip (1993/99) subscribes to neither of the above two assumptions. The reason she can do so is that she regards culminations as having *temporal extent*, just like accomplishments: "[p]rotracted events and culminations are extended in time" (Filip 1993/99, p. 111).

Filip does not belong with authors like Kenny (1963) or Verkuyl (1989, 1993), however, who deny any linguistically relevant difference between these classes. She differentiates culminations from accomplishments by saying that "[c]ulmination events differ from protracted events in that they have no expression in their semantic description associated with the process preceding the culmination (see Pustejovsky 1988:30ff.)" (*op. cit.*, p. 111). But if there is no expression of the processes preceding culminations in their semantic contents, then how is it encoded? If culmination events are extended in time, then the extended part must be encoded in some way, and, what is more, in a way that is different from the encoding of the extended process part in the case of accomplishments.

⁹Cf. the difference between the asserted content-manipulating imperfective operator IMPF needed for durative predicates and the presupposed content-manipulating IMPF-2 needed for achievements in Piñón (2008b), who made explicit the exact semantics of the imperfective operator that is needed by Malink (2008) in his account of *negated* achievements, and the same therefore applies to the account of progressive achievements in Heyde-Zybatow (2008) who employs the same framework as Malink (2008).

Pustejovsky himself presents various solutions to the achievement/accomplishment distinction. In Pustejovsky (1991), for instance, he locates this distinction in the specification of the "activity" part or lack thereof in their semantic description, but for him, this activity part is an agentive activity (rather than a generic process), and so the difference for him boils down to agentivity (accomplishments) *vs.* nonagentivity (achievements). He, of course, then may have problems differentiating non-agentive accomplishments (such as *the ball roll*_{\emptyset} *to the bottom of the slope*) from achievements (cf. Section 1.1.2).

In what follows, I inspect three other possible options for differentiating culminations from accomplishments if we wish to retain the idea that culminations have temporal extent.

Solution 1: Presupposed activity part. One option open to Filip (1993/99) is to say that lack of the process part in the semantic description of achievements is related to the view accepted by many authors (e.g., Piñón 1997, Löbner 2002, Heyde-Zybatow 2008, Martin 2011) about the presupposed content of achievements. That is, we could say that although culminations are associated with a temporally extended *change*, the occurrence of the extended process associated with this change is *presupposed*, rather than part of their asserted content, and so they simply describe instantaneous change. Taking this view, however, would mean that we need to assume that the progressive of a culmination describes this presupposed process, and so it is either a culmination-specific progressive (accessing the presupposed process of its input, as the IMPF-2 imperfective operator in Piñón 2008b), or the culmination in its scope is coerced to include the process part in its semantic description.

One may wonder if a uniform, *disjunctive* semantics for the progressive cannot be given, after all. We could say that a progressive is a partitive operator over an event that is part of *either* the asserted, *or* the presupposed content. This definition of the progressive would lead to wrong results, however. For instance, *John managed to open the door* presupposes that John was *trying* to open the door, and so during the process of John trying to open the door — however small or non-existent chance John has of actually opening the door — (2.16) is predicted, incorrectly, to be true.

(2.16) John is managing to open the door.

Thus, a disjunctive, uniform semantics for the progressive is not tenable, and we are left with a culmination-specific definition.

Solution 2: Vague vs. sharp boundaries. Another way to potentially make a difference between culmination-achievements and accomplishments is to say that the former, but not the latter have a *sharp* (right) boundary. However, as noted by Altshuler and Filip (2013), some accomplishments, such as *write an email*, can also have a sharp boundary; e.g., as Altshuler and Filip (2013) put it, writing of an email can be "understood as culminating in the moment when the 'send' button is pressed".

Thus, having a sharp boundary is not specific to culminations, and is not suitable to distinguish culminations and accomplishments, because even accomplishments with a sharp boundary can be complements of aspectual verbs as in (2.17a), while the same is not true of culminations (Dowty 1979, Mittwoch 1991, Piñón 1997, Kearns 2003, Rothstein 2004, see Section 5.4.3), cf. (2.17b):¹⁰

- (2.17) a. John started/finished writing an email.
 - b. *Mary started/finished arriving at the station.

It is a general issue with regarding culminations as extended that we need to resort to alternative methods and extra tools for explaining their *non-durative*-like behaviour (such as compatibility with temporal point adverbials like *at noon* and reduced compatibility with adverbs of completion like *halfway* or as arguments of aspectual verbs like *finish* and of *spend X time*). If culminations are extended, then it is surprising that a temporal frame adverbial such as *in a few minutes* only has an "after" reading when it modifies culminations, as in *John will arrive in a few minutes*, while it is ambiguous between an "after" and a "duration" reading when it modifies accomplishments, as in *John will build a house in two years* (cf. Vendler 1957, Dowty 1979, Kearns 2003, a.o.), where on an "after" reading, an eventuality happens *after* the given amount of time as measured from an event-independent reference time, and on a "duration" reading an eventuality itself takes the given amount of time.

Indeed, it appears that culminations only behave like a durative predicate with respect to the progressive, while in all other respects, they function as punctual pred-

¹⁰Also, given that *finish* arguably refers to the right boundary of an accomplishment, i.e., it results in a culmination predicate, it is unclear why *John finished building a house* is felicitous, if the culmination of *build a house*, as opposed to that of culmination-achievements (which is what *finish building a house* is), is vague. Of course, as will be discussed in Section 4.3.2, the precise location of the endpoint of events is, indeed underdefined, but this is not a point of difference for different aspectual classes. For instance, *Microsoft won the race for computer operating software in the 1990s* (from *Distribution Agreements Under the EC Competition Rules* by Valentine Korah & Denis O'Sullivan) is a culmination-achievement despite the fact that the exact point of the winning is vague, and even more so than the point when a house-building can be said to culminate.

icates (though see Chapter 5 for some exceptions). Importantly, their combinability and interpretations with *temporal adverbials* appears to display a strictly non-durative behaviour. It would be a welcome aspect of our semantics if it were able to explain why the durative behaviour of culminations is specific to a position in the scope of the progressive.

Solution 3: Salience. A third option for differentiating between accomplishments and temporally extended culminations is with reference to some kind of conceptual salience: we can say that the final punctual change is conceptually more salient in the case of culminations (but not in the case of accomplishments) than the process leading up to this punctual change. This solution captures the idea of Filip (1993/99) and Alt-shuler and Filip (2013) most closely. This proposal is highly compelling, but it is difficult to see how it can be incorporated in model-theoretic semantics. The analysis I will propose is an attempt at this incorporation and the formalization of Filip's (1993/99) idea of culmination-achievements sharing characteristics with both accomplishments and happenings.

It should be noted that a somewhat similar route taken by Pustejovsky (1998) who (instead of his agentivity-based distinction in Pustejovsky 1991) uses the notion of "event headedness". On his account, events in an event structure are ordered by relative prominence, and the head is the "most prominent subevent in the event structure of a predicate" (*op. cit.*, p. 72). Thus, Pustejovsky differentiates accomplishments like *build a house* and culminations like *arrive* by postulating that in the case of the former, the head is the initial subevent, while in the case of the latter, the head is the final subevent. But note that on Pustejovsky's (1998) account, the second part of the event structure of a transition (a telic predicate) is the result state, rather than the culmination point. This specific theory therefore introduces a distinction between accomplishments and achievements with respect to the importance of their result state, which might be potentially unwanted.

2.2.3 A lexicalization approach

A different solution is presented by Piñón (1997), the proposal of whom is worth discussing here, because although Piñón discusses progressive achievements only very briefly, he is the only one, as far as I am aware, who is explicitly committed to the route that is perhaps the most natural one for accounts disallowing progressive achievements. In particular, Piñón (1997) argues that progressive achievements are marginal and always involve an "unpredictable meaning shift" and so they are always *lexicalized* cases. His most general argument for a lexicalization approach is that the progressive in these cases does not mean that the eventuality described by the predicate itself is in progress:

In [*Rebecca was reaching the summit when it began to rain*], *was reaching* may be paraphrased as 'was approaching': it is not the actual attainment of the summit that was in progress. In [*Astrid was winning the race when we arrived.*], *was winning the race* has 'was ahead in the race' as a paraphrase—the winning itself was clearly not in progress. The question in [*Are you finding everything okay*] may be rephrased as 'Is your search for what you are looking for going well?'. Again, it is not the eventuality described by the achievement (here: a finding) that is in progress. Finally, *was dying* in [*We visited the wounded soldier, who was dying*] may be paraphrased as 'was suffering and on the verge of death': the death itself was not in progress. (p. 279)

Piñón's problem with having progressive achievements derived through a compositional mechanism of interpretation combining the meaning of the progressive and of the relevant achievement thus appears to be that it is the *preliminary process*, rather than the *component process* that the progressive describes in their case, much as authors like Smith (1991), Kearns (2003) and Martin (2011) maintain. So what Piñón (1997), like these authors, holds is that the progressive cannot have the same semantics when it applies to achievements as it does when it applies to accomplishments, as the following passage shows:

If achievements denote instantaneous eventualities, then we do not expect semantically regular progressives to be possible, precisely because eventualities without duration are never in progress. (p. 279)

But then it seems to me that it would be more parsimonious to follow authors like Kearns (2003) and assign a special meaning to the progressive operator itself when it applies to achievement predicates, namely, one which results in a preliminary process reading. This could even explain the most specific argument of Piñón (1997) about *dying*, which runs as follows:

In each case, there seems to be an element of irreducible *lexicalization* involved. Why, for example, can we not refer to a captured spy in her final minutes before committing suicide as someone who is dying? (p. 279)

I will argue that this question can be answered even without recourse to an achievement-specific semantics for the progressive, but it is straightforward given a special, *preliminary process* semantics of this operator and the assumption that the type of the process leading up to an event falling under a specific achievement is *lexicalized*¹¹ (which is actually exactly what Rothstein 2004 contends, cf. below): since in the minutes before a spy commits suicide, the preliminary process associated with dying (increasing suffering and/or deterioration of health) is not happening.

What a lexicalization approach of progressive achievements would (weakly) predict is that it should only be a handful of predicates that are accessible in the progressive (though, strictly speaking, the essence of lexicalization approaches in general is that "anything goes" within some bounds, so in theory, all achievements could have a lexicalized progressive). As such, a lexicalization approach cannot easily explain the observation cited in Section 2.1.2 that happenings like *recognise* cannot appear on the preliminary process reading. A pure lexicalization approach in itself makes no predictions about which achievements will have a progressive form lexicalized and what the associated meaning thereof will be, and so nothing guarantees that it is *arriving* rather than *noticing* that is lexicalized.

In addition, as noted by Hana Filip (p.c.), if progressive achievements are lexicalized, we would expect that they are encoded as a simple lexical item (a word) in at least *some* language, but I do not know of such a language. If a vast cross-linguistic study should verify that there is indeed *no* language in which progressive achievement meanings are encoded in a simple lexical item (a word), then this is arguably not an accidental gap, and would severely weaken a lexicalization account of progressive achievements.

However, one welcome aspect of the fact that a lexicalization approach does not predict a uniform semantics for all progressive achievements is that it is compatible with some, but not all progressive achievements implying the imminency of the culmination. Although prototypical progressive achievements like *arriving at the station* or *reaching the top* do have this implication, this is not true of all progressive culminations. Culminations like *win* and *die* can appear in the progressive without implying that the culmination is imminent (cf. Section 5.5.1), as in the sentence *From the day we are born, we are dying*. On a lexicalization approach, this lack of uniformity is easily accommodated.

Achievements as boundaries. Although I do not endorse Piñón's (1997) idea that progressive achievements are all lexicalized, the general approach of Piñón (1997) to achievements as *boundaries* appears to me to capture a fundamental component of

¹¹So that it is not the progressive achievement construction that is lexicalized, but merely the type of the preliminary process for each achievement predicate.

their meaning and aspectual characteristics. The idea of achievements as boundaries was entertained already by Mourelatos (1978, 1981) who suggested that achievements "capture either the inception or the climax of an act" (Mourelatos 1978, p. 416).¹² And conversely,

there cannot be an accomplishment without a closely related end-point achievement — one cannot say "I wrote/shall write the letter" if he cannot say "I finished/shall finish the letter." (Mourelatos 1978, p. 417)

These are observations I will exploit in my notion of a *cover event* associated with culmination-achievements like *arrive* which is a key ingredient of the analysis of progressive achievements I will propose.

2.2.4 Type shifting approaches

2.2.4.1 Accomplishments as the coerced type

Given that not all achievements can appear in the progressive and given that when they do, they typically behave like progressive accomplishments, progressive achievements are most standardly analysed as involving a form of coercion into an accomplishment-like type that can be input to the progressive operator.

Analyses that rely on a type shifting operation, while explaining the commonalities between progressive achievements and accomplishments (for instance, the imperfective paradox), often have difficulty explaining many achievement-specific characteristics of progressives. For instance, most accounts do not address the question why a progressive culmination is generally true only during the very final interval of its "activity part" (say, the journey to the goal in the case of arriving somewhere), that is, why only a very small, final part of the relevant activity is "used" in the coercion of the achievement into a durative predicate.

Rothstein (2004, p. 143–5) does contemplate the question, and introduces a special pragmatic constraint about the culmination of a progressive achievement being a good possibility for explaining the high probability of the culmination (cf. Section 5.5.2), and, by indirect means, the imminency meaning component (cf. Section 5.5.1) of such progressives. By having such a constraint, Rothstein will have difficulty accounting for

¹²If we consider the implications of this view, then what we see is that semelfactives like *flash* or *knock* are excluded from the class of achievements—in accordance with the view of Smith (1991)— since they are not usually the inception or the climax of another event.

cases of progressive achievements without either of these features, as *dying* or *winning*: why is the pragmatic constraint inactive in these cases, but not, say, in the case of *reaching*? More importantly, as I will argue in Section 5.5.2, there are scenarios where no high probability is attached to the culmination, and yet a progressive achievement does have an implication of imminency. However, I do find Rothstein's solution compelling in linking together these two features. What I will therefore suggest is an *opposite* route, that is, accounting for the intuitively high probability of culmination as a cancellable inference based on imminency.

The crux of the issue is that in itself, an account of progressive achievements relying on type shifting of the achievement predicate in the scope of a progressive to a regular accomplishment, like the account of Moens and Steedman (1988), will, in absence of extra explanatory tools, predict a too homogeneous picture with respect to the interpretation and behaviour of progressive achievements: if an achievement in the progressive is coerced into an accomplishment, then we expect it to behave like one. This led some proponents of the event coercion/type shifting accounts to introduce a separate, dedicated category of accomplishments for analysing progressive achievements: "derived accomplishment" for Rothstein (2004), and "durative achievement" for Martin (2011).

Both authors locate the difference between regular and such special accomplishments in the activity part. The difference between a *derived accomplishment* and a regular one is that the activity part of the former, but not the latter is *lexically empty* (Rothstein 2004, p. 49); while the difference between a *durative achievement* and an accomplishment is that the former but not the latter *presupposes* its activity part (Martin 2011). These are two opposing views about the lexical specification of the crucial activity part, and I will side with Martin (2011) — based on evidence inspected in Section 3.2 — in maintaining that the activity part is lexically specified for culmination-achievements and accomplishments alike, but is presupposed in the case of the former. (Indeed, lack of lexical specification in the case of achievements also leads to "false positives", as I will argue below, that is, progressive achievements are predicted to be felicitous which should not.)

2.2.4.2 Derived accomplishments as the coerced type

On the account of Rothstein (2004), achievements in the scope of the progressive are coerced into derived accomplishments, which are like regular accomplishments in having an activity part describing the activity the agent is engaged in, a BECOME-event describing the change in the theme, and a culmination, but their activity part is lexically empty.

Because Rothstein (2004) has no aspectual class distinction between culminations and happenings, the way Rothstein (2004) aims to explain why *arrive*, but not *notice* can be coerced into a derived accomplishment is to incorporate *incrementality* into the definition of an accomplishment, thereby ensuring that achievements such as *recognise* cannot be coerced into a durative eventuality, since it is not generally preceded by an incremental change. However, Rothstein (2001) herself argued that there are non-incremental accomplishments such as *repair a computer* or *sing the baby asleep*, which means that incrementality in the traditional sense cannot be an integral component of the definition of accomplishments.

Rothstein (2001, 2004) thus argues that incrementality in the sense required by accomplishments in general, is some *inherent linear ordering over the subevents of a particular change*. This is an idea that I will embrace and directly build into the notion of a scale of change in my framework. The reason a different characterisation of this inherent linear ordering is needed is that the exact formalisation that Rothstein provides leads to false positives.

In particular, Rothstein (2004) defines accomplishments based on the incremental chain specific to their BECOME-event, and the incremental relation holding between this BECOME-event and the process part of the accomplishment, where she defines incremental chains and relations as follows (Rothstein 2004, p. 107–8) :

Definition 1 (Incremental chain (Rothstein 2004))

Let e be a BECOME event. An incremental chain C(e) is a set of parts of e such that:

- 1. the smallest event in C(e) is the initial bound of e
- 2. for every $e_1, e_2 \in C(e)$: $e_1 \sqsubseteq e_2$ or $e_2 \sqsubseteq e_1$
- 3. $e \in C(e)$

Definition 2 (Incremental relations (Rothstein 2004))

Let e_1 be an activity, e_2 be a BECOME event, and $C(e_2)$ be an incremental chain defined on e_2 . INCR $(e_1, e_2, C(e_2))$ (e_1 is incrementally related to e_2 with respect to the chain $C(e_2)$) iff there is a contextually available one-one function μ from $C(e_2)$ onto PART (e_1) (the set of parts of e_1) such that for every $e \in C(e_2) : \tau(e) = \tau(\mu(e))$.

The problem, I propose, is that by defining incrementality as a particular relation between eventuality parts (namely, parts of the activity and the change that is simultaneously going on), rather than through an explicit reference to a scale, Rothstein's account may fail to exclude events like *recognising John* from incremental, derived accomplishments.

The reason for this is that her definition of a BECOME event — as an event such that "at the time immediately preceding the beginning of e, $\neg \phi$ is the case and at the time immediately following the end of e, ϕ is the case" (Rothstein 2004, p. 107) — does not exclude the possibility that the BECOME event is an extended, non-incremental event with $\neg \phi$ being true throughout, and ϕ at its final instant. It would still have "individuable" parts (a first part at which $\neg \phi$ is true, and a second one at which ϕ is true), each part "has a distinguishable upper bound", and "these parts have a natural and inherent order".

Then, since Rothstein (2004, p. 49) argues that the activity part of accomplishments derived from achievements is not lexically constrained, it is possible to have a one-to-one mapping with some activity and the relevant BECOME event: say, Mary is walking towards John, getting closer and closer, and at one point, she gets close enough and recognises John. Then we could have the activity of walking toward John up until the moment of recognition be the image of the subevent of the BECOME event at which Mary is not yet aware that the person is John, and the final moment of the activity the image of the momentaneous event of her recognising John. Still, it would be odd to say during Mary's walk that ?*Mary is recognising John*, despite the possibility of coercion into an accomplishment structure of the required form.

I will adopt the idea from Rothstein (2004) that reference to an inherent ordering is a crucial component of accomplishments and therefore is a necessary condition for the acceptability of a progressive achievement on the preliminary process reading. Where I will differ from Rothstein's analysis is that I will assume that culminations do in fact lexicalise this inherent ordering and that reifying and referencing this ordering as a scale can tackle the issues that arise for event coercion-based accounts.

2.2.4.3 Durative achievements as the coerced type

Martin's (2011) account differs from that of Rothstein (2004) in that she assumes that culmination-achievements presuppose the occurrence of an activity (rather than having a lexically empty activity part), and in that she makes finer-grained ontological distinctions, building on Piñón (1997), differentiating non-semelfactive happenings and

culminations. I will adopt both of these ideas, but in contrast to Martin (2011), I will argue that lexical association with a specific kind of activity part is sufficient to derive the correct semantics for progressive culmination-achievements, and there is no need to posit durative achievements as the coerced input type for the progressive.

The problem with positing durative achievements is that apparently extra assumptions are needed to explain the infelicitiousness of modification by agentive adverbials and *halfway through*. While Rothstein (2004) does note these characteristics of progressive achievements, she does not offer an explanation of these restrictions within her theory. Martin (2011), in contrast, pays due consideration to these questions, but — by virtue of positing durative achievements — all traditional avenues of explanation relying on the instantaneous nature of achievements is closed off.

So Martin (2011) assumes a principle according to which "an adverb can modify an event whose occurrence is presupposed by a verb if and only if it modifies in the same way the event(s) whose occurrence is asserted by the same verb" to account for the restriction on agentive adverbial modification. However, such a principle, like Rothstein's principle of probable culminations, is either too specific (referring only to *agentive* adverbials) to be explanatorily satisfactory, or — in its present form — too general to be descriptively adequate. For instance, there are actually several adverbs which modify the "presupposed event" and the "asserted event" independently for achievements, as Rothstein (2004, p. 44) argues on the basis the following examples:

- (2.18) a. John was dying for a long time, but he actually died quickly.
 - b. It was very turbulent while the plane was landing, but we actually landed smoothly.
 - c. #Mary was writing a book slowly, but she actually wrote it quickly.

In fact, Rothstein argues that this is a point of difference between achievements and accomplishments: modification of the progressive form of the first (cf. (2.18a)–(2.18b)), but not the second (cf. (2.18c)) can be independent from the modification of the event predicate itself.

So it seems that even the most well-developed and well thought-out event type coercion accounts of progressive achievements face difficulties in accounting for some aspectual class differences, namely, the exclusion of happenings from preliminary process progressives, and/or the infelicitousness of some adverbial modifications of progressive culminations. I take this to indicate that the problem is rooted in the initial

assumption of these accounts, namely, that the achievement predicate in the scope of the progressive is durative, which is an assumption I will do away with.

2.2.5 A non-partitive semantics for the progressive

The preceding discussion indicates that partitive theories of the progressive (the most generally accepted approach to the progressive) run into complications in the case of progressive culminations like *arriving*.

Let us therefore take the non-partitive analysis in Asher (1992), on whose theory of the progressive, a progressive PROG(φ) is true during a process part normally *lead*ing to or cotemporaneous with a φ -event. It should be noted here that Bonomi (1999, pp. 185-6) showed that by using the "prior to or cotemporaneous with" temporal relation instead of a mereological (part-of) relation, Asher (1992) makes wrong predictions about accomplishments in some cases.¹³ More importantly for our purposes, Asher's (1992) non-mereological formulation also makes wrong predictions for achievements as it stands. Owing to the non-strict temporal ordering relation Asher uses in the definition of the progressive, $PROG(\varphi)$ is predicted on his account to be assertable and true at the point where a non-durative event type φ is true. Thus, Mary is arriving at the station should be true at the exact point when Mary arrived at the station. This does not appear to be intuitively correct, but the more serious problem is that just as Rothstein's (2004) analysis was shown to predict false positives for happenings, so does Asher's (1992). On his definition, any progressive achievement, including progressive happenings like #Mary is noticing the picture should be always assertable at the point its input predicate is made true which is not the view generally accepted. Happenings are only felicitous in the progressive on a slow-motion reading, but not in everyday settings. Thus, absence of an event-mereological component in a definition of the progressive is no guarantee to the prediction of the correct truth-conditions of progressive achievements.

- (i) a. The water is descending to that meadow.
 - b. The water is wetting the meadow.

¹³One of the examples Bonomi (1999) brings is that of a brook which has just been diverted and is approaching a meadow; in this case, at some time long before the water from the brook reaches the meadow, when (ia) can be truthfully asserted, (ib) will also be incorrectly predicted to be true: in the current situation, the water descending to the meadow is a normal course of events, but then the water will also automatically wet the meadow in these normal courses of events. Because the current state of affairs is not required to be a part of the wetting of the meadow, only temporally prior to it, (ib) is predicted to be true, while it is intuitively false.

Perspectives. Although I will not employ Asher's (1992) temporal relation or his notion of normalcy in the analysis of progressive I propose, there is an aspect of his account that I will adopt, namely, the concept of *perspectives*. A perspective is a proposition describing some characteristic, some aspect of an eventuality. What the admissible perspectives are is defined by the discourse context. Asher defines the progressive relative to a perspective admissible in a given context. Among others, this makes it possible for him to account for the fact that two accomplishments describing a mutually contradictory outcome can be simultaneously in progress in a situation, such as in Heim's example, *Irene is cooking fish stew, but the cat is eating the fish* (these pose problems for the *inertia worlds* approach of Dowty 1979 to the progressive).

Just as Asher (1992), I will use existential quantification over what will correspond to his perspectives in my account of the progressive. On the scale-based account I propose, discourse contexts select a *series of snapshot descriptions* that capture the contextually relevant aspect of the change associated with an eventuality, and a progressive is true if this series of snapshot descriptions can be embedded into a series of snapshot descriptions characterising the change associated with the input predicate of the progressive.

The essence of my account is therefore to integrate the advantages of several earlier theories, namely, theories both of the progressive (and progressive achievements in particular), and of scale-based characterisations of aspectual classes.

2.3 Scale-based frameworks

2.3.1 Scales in formal semantic analysis

Scales have been successfully put to use in the analysis of gradable adjectives such as *tall* (see, e.g., von Stechow 1984, Rotstein and Winter 2004, Kennedy and McNally 2005, Kennedy 2007), analysed as mapping objects to degrees on a scale. The most common formal characterisation of a *scale* is a triple $\langle \delta, S, \langle S \rangle$ consisting of a dimension δ (such as length, warmth, price), a set *S* of *values* (or *degrees* or *points*) and a linear ordering $\langle S \rangle$ over the set of values. For instance, $\langle WARMTH, \mathbb{R}, \langle \mathbb{R} \rangle$ is a scale of warmth with no minimal or maximal value, while $\langle LENGTH, \mathbb{R}_0^+, \langle \mathbb{R} \rangle$ is a scale of length with a minimal, but no maximal value.

The semantics of scales, just as the semantics of time (cf. e.g., Landman 1991), can be either interval-based (as in von Stechow 1984, Hay et al. 1999) or (more commonly)

point-based. Although, for instance, Kennedy (2001) and Schwarzschild and Wilkinson (2002) bring for arguments for an interval-based semantics, I will, for simplicity, adopt here the more common point-based analysis.

There is some vagueness surrounding the more minute details about the concept and features of a scale (cf. also Solt 2015). The following is a list of questions with respect to scales which are often left implicit or not addressed:

- 1. What objects can serve as the carrier set (only abstract degrees, or concrete objects, as well)?
- 2. Related to the previous question, are degrees numbers (or at least abstract points of measurement)?
- 3. What are admissible ordering relations?
- 4. Is the ordered set dense or discrete (or are both admissible)?

As regards the first issue, the general (but usually only implicit) assumption was originally that a scale is composed of a set of abstract objects, degrees, which concrete objects can be mapped to at particular times. This has mostly been the tacit assumption in scale-based works on gradable adjectives (e.g., Kennedy 2001, Kennedy and McNally 2005, Kennedy 2007). On the other hand, as scales gradually found their way into the semantics of verbs, this implicit assumption has been silently rejected by many authors. Kratzer (2004), for instance, argues that the degrees making up a scale can be concrete objects (e.g., the ordered parts of an apple in the case of predicates like *eat an apple*).

It is also often implicitly or explicitly taken for granted (e.g., by Kennedy and Mc-Nally 2005, Sauerland and Stateva 2007, 2011) that the degrees of a scale are *(real) numbers* or at least abstract representations of measurement isomorphic to a subset of the real numbers — despite the simple characterisation of a scale as a linearly ordered set — and so we have a *metric* automatically defined over scales. On these approaches, the dimension of the scale would be pivotal (since all scales are made up of numbers) — even though, as Sassoon (2010, p. 173) pointed out, the notion of a dimension might not be clear enough to encode the distinction between some scales, such as width and length. In contrast, the ordering relation $<_S$ of a scale *S* would always need to be the smaller-than or greater-than relation over real numbers, and so its role would be minimal, only defining the *direction* of the order. On the other hand, if we want to retain maximum generality and remain neutral on what ordered sets can serve as scales, we can have any linear order serve as $<_S$ (for instance, the earlier-than relation over time points), and degrees can be objects, state descriptions, numbers or any other things that are elements of an ordered set.

Favouring the more general approach to scales is particularly advantageous if we apply a scale-based framework in the verbal domain, in which case it is possible to directly adopt classical homomorphism-based approaches to incremental predicates such as Krifka (1992), allowing linearly ordered sets of parts of a concrete entity to be scales (as proposed by Kratzer 2004, for instance). But even in the semantics of adjectives, authors like Cresswell (1976) and Bale (2008, 2011) have employed scale-based approaches in which degrees are not numbers, but *equivalence classes* under a primitive ordering relation (such as the *taller-than* relation over individuals). In scale-based works, there have been arguments both for (e.g., Bale 2011) and against (e.g., Solt and Gotzner 2012) conceptualizing degrees as equivalence classes rather than numbers.

Those who do not view degrees as numbers argue that not all scales allow for measuring distances between degrees (Cresswell 1976, Bale 2011, van Rooij 2011a). A core argument of these authors is that not all adjectives combine with measure phrases like *2 meters*, and the reason for this, they argue, is that while some adjectives, like *tall*, are associated with an *interval scale*, the scale of others like *happy* is merely an *ordinal scale*, in the terminology of measurement theory, i.e., the former, but not the latter has a metric defined over it. Cresswell (1976) was perhaps the first to note that we can make comparisons like *more beautiful* without units of measurement, and that for some adjectives which do allow for measures, a new relation can be defined over the degrees of their scale based on the "natural metric" of the base relation (say, spatial distance in the case of *tall*).

A further potential advantage of defining a metric over and above scales themselves is that we have the additional freedom of defining the metric in different ways for potentially different scenarios. For instance, when the issue is the loudness of sound events, the ordered scale could be the sound events progressing from the least loud to the most loud, and the metric would be logarithmic when discussing a human agent, but linear when discussing absolute loudness.

As regards the third issue (what kind of ordering relation is assumed?), scale-based approaches nearly unanimously assume linear ordering. However, the path argument of motion verbs in Beavers (2012) is not linearly ordered, even though he also assumes scales to be usually linearly ordered sets with this exception.

As regards the fourth question (namely, whether scales are dense or discreet), this issue is rather less dealt with. Average scales (and certainly those associated with grad-able adjectives) are generally dense, but there are certainly discrete cases, as well (such as a scale corresponding to the number of some kind of entity, such as 1 *cat*, 2 *cats*, 3 *cats*, ...). And one of the main insights of Beavers (cf. Beavers 2002, 2008, 2013) is that there are scales (corresponding to non-gradual changes) which comprise exactly two degrees, which are not only not dense but are discrete, being finite.

2.3.2 Former scale-based frameworks for aspectual analysis

The vast majority of authors employing a scale-based semantics for aspect focus on accounting for *telicity* (e.g., Hay et al. 1999, Filip and Rothstein 2006, Kearns 2007, Filip 2008, Kennedy and Levin 2008, Piñón 2008a, Rappaport Hovav 2008), and to my knowledge, the only author to focus on a scale-based approach to *durativity* is Beavers (2002, 2008, 2013). The reason is probably that scale-based semantics found its way into the verbal domain primarily through deadjectival verbs like *empty, lengthen* (which present a natural bridge from adjectives to verbs) in order to account for their ambiguous behaviour with respect to telicity (cf. e.g., Hay et al. 1999). Scale-based semantics was then extended to all predicates with an (implicit or explicit) argument that measures out events in its denotation (cf. e.g., Filip and Rothstein 2006).

In this respect, too, the theory of Beavers (2002, 2008, 2013) is more general, extending to a far wider range of predicates (though not all predicates, e.g., he argues that *see, smell, play, ponder* are not associated with scales), including those which belong to Vendler's class of achievements. Given that my goal is to account for the accomplishment/culmination/happening distinction and offer a scale-based semantics for the progressive, I will build most closely on Beavers (2002, 2008, 2013), but I will use an even more general framework. But first, let us inspect briefly whether other scale-based frameworks could not be extended or used for these purposes.

2.3.2.1 Localist/conceptual semantic approaches

Before turning to scale-based approaches to verbs in formal semantics, it is worth discussing a closely connected school of thought. Localist approaches originating with Gruber (1965) hold that motion and spatial location plays a key role in the construal of a wide range of domains. This is captured perhaps most concisely by the Thematic Relations Hypothesis (Jackendoff 1983, p. 188), according to which the principal event-, state-, path- and place-functions in any semantic domain are those used for motion and location. On such a localist/conceptual semantics approach, many different predicates (predicates describing motion, covering, change of possession, change of properties, performance, creation and consumption, alike) can be captured using the same structure of *state descriptions* (using the abstract predicate BE) and *structure-preserving mappings* between the event, time, and a path argument. And as Jackendoff (1996, p. 335) put it,

the measuring out relations between different parameters of the situation can be formally differentiated in terms of which axes are [connected via structure-preserving binding].

The point of connection of scale-based frameworks in formal semantic theories to localist/conceptual semantic approaches is that on many scale-based approaches, the different dimensions of measuring out for different predicates are unified in the notion of a scale, which is homomorphically mapped to the event, and the dimension of the scale differentiates between different measuring out relations.

I will strive to the make the connection between scale-based and conceptual semantic approaches even closer, extending it to states and to the relation between states and dynamic events.

2.3.2.2 Closed and bounded scales

Kearns (2007) and Kennedy and Levin (2008) (building on Hay et al. 1999) assume a semantics for deadjectival verbs according to which predicates associated with a *closed scale* have default telic interpretations. Let us assume that we extend scales to a wider range of predicates along the line Beavers (2002, 2008, 2013) does, namely, to those implying change in some theme. We now need to find a way to capture the achievement/ accomplishment difference, which are all telic predicates, and hence are associated with closed scales on this account. Let us focus on the culmination/accomplishment distinction. Since Kearns (2007) and Kennedy and Levin (2008) use scales containing an infinite number of degrees, what comes to mind as a possibility is the adoption of the idea proposed in Section 2.2.2 that culminations, but not accomplishments specify a sharp endpoint.¹⁴ But at this point, we reach a contradiction, as accomplishments as telic predicates are associated with a closed scale, but a closed scale has one specific maximal element (cf. Appendix A), which means that accomplishments have a sharp final boundary.

¹⁴Recall that, as discussed in Section 2.2.2, this characterisation may lead to problems, though.

Also, we cannot say that the difference between accomplishments and culminations is instead that the maximal element is not part of the scale for the former, because that would result in an open scale which is associated with an atelic interpretation. As an illustration, consider a closed scale, which is necessarily bounded, as well. An example for this scale is the set of real numbers from 0 to 1, including both endpoints, that is, the interval [0, 1] (see Figure 2.1).

Figure 2.1: A closed scale, e.g., [0, 1]

Then, consider a scale, which is *bounded*, but not closed. The set of real numbers from 0 to 1, excluding the endpoints is such a scale (see Figure 2.2). In this case, al-

Figure 2.2: A bounded scale, e.g., (0, 1)

though the scale is not closed, it does have a lower and an upper bound, e.g., 0 and 1, respectively, in our example of the interval (0, 1). However, what is relevant for telicity is closedness, and *not* boundedness, as the following example shows:

(2.19) John was walking toward the station.

0-

(2.19) is atelic, even though the path (where we assume that a path is a totally ordered set of points) extending to the station is *bounded* (or at least, has an upper bound): it cannot extend further than the station, i.e., the location of the station is an upper bound on it, as in Figure 2.2.

The only way it seems to me that we can model the achievement/accomplishment difference in this model is if we go with the analysis of Beavers (2002, 2008, 2013) who proposed that achievements are associated with a scale consisting of exactly two points, while the scale of accomplishments is multi-valued (i.e., contains more than two points). Since finite discrete scales are by necessity closed, the inherent telicity of achievements (cf. Filip 2008) follows naturally in this case.

2.3.2.3 Scales as defining an ordering over events

On the view expounded in Filip and Rothstein (2006) and Filip (2008), a scale (a linear order) induces a partial order on the set of events, where the events ordered below another event are stages of the "greater" event, in the sense of Landman (1992). Telic predicates on this account denote sets of events that are *maximal* with respect to the ordering over a contextually relevant set of events induced by the scale associated with the predicate. As Filip (2008) stresses, maximalization over events presupposes an external linear order (a scale) over and above the mereological partial ordering over events (and the maximalization operator over events therefore has a stricter input requirement than the maximalization operator over individuals).

However, this analysis of telicity will undergenerate in the case of achievements, as pointed out by Kardos (2012), and also Filip (2008) herself. Since achievements are non-durative, events in their denotation cannot be ordered as stages of one another. With no ordering, maximalization cannot apply, and, thus, achievements do not come out as telic. The only way I can think of to extend the event maximalization account to achievements is, again, by employing Beavers's (2002, 2008, 2013) idea about achievements being associated with two-point scales and also his proposal that events in the denotation of achievements consist of exactly *two* parts (cf. Section 3.1.2). Then these two event parts can be ordered based on the two-point scale projected by an achievement predicate, and achievements are then telic because they only have maximal, two-part events in their denotation.

2.3.2.4 Scales as encoding change in an individual

The starting point of Beavers (2002, 2008, 2011, 2012, 2013) is that dynamic predicates are associated with some change, where "change is defined as some theme transitioning to and maintaining a new value along some property scale, which is an incremental argument" (Beavers 2013, p. 684). Beavers uses an event semantic framework in which a dynamic predicate has an extra argument for a scale that he calls a *scale of change*. Beavers then adapts the mapping relations of Krifka (1992, 1998) to a homomorphism between the event and the scale argument of a predicate by substituting the scale of change for the incremental theme or the path argument as follows:

Definition 3 (Movement Relation (MR) of Beavers (2013))

Each part of an event *e* corresponds to a part of its associated scale *s* and vice versa; temporal adjacency in *e* corresponds to spatial/scalar adjacency in *s*, and the initial and final points in *e* are mapped uniquely to the initial and final points in *s* respectively.

This way, it is the scale argument that measures out an event, and different scalar properties determine different aspectual properties of predicates (cf. Beavers 2013, p. 684):

i) The *specificity* of the endpoint along the scale that needs to be minimally reached determines the telicity of a predicate.

If a predicate specifies an endpoint along the scale, then the predicate is telic; if a predicate does not specify an endpoint along the scale, then it is atelic.

ii) The *mereological complexity* of the scale determines the gradualness of the change associated with the predicate thereby determining durativity.

Minimally complex scales (i.e., scales with exactly two degrees) correspond to non-gradual change, and are associated with non-durative predicates; complex scales (i.e., scales with at least three degrees) correspond to gradual change, and are associated with durative predicates.

I will endorse these correspondences between the scale of change and the aspectual properties of predicates.

As for the first property of specificity, Beavers (2011, 2013) specifies four *degrees of affectedness*, which encodes how specific the predicate is about the theme's progress on the scale. He differentiates the following degrees of affectedness with respect to event predicate ϕ , where result'(x, s, g, e) specifies that the change that theme x undergoes on scale s during event e ends in degree g:¹⁵

- a. *x* undergoes *quantized change* iff for all *e* such that ϕ applies to *e*: $\exists s[\mathsf{result}'(x, s, \mathbf{g}_{\phi}, e)]$ (*x* reaches a specific state \mathbf{g}_{ϕ} on *s*, e.g., *peel*, *break*, *shatter x*)
- b. *x* undergoes *nonquantized change* iff for all *e* such that ϕ applies to *e*: $\exists s \exists g [result'(x, s, g, e)]$ (some result state *g* obtains, e.g., *cut*, *widen*, *lengthen x*)

¹⁵Below, I reproduce the formulas of Beavers (2011, p. 358) with a slight modification. In the formulas of Beavers (2011, p. 358), ϕ is formally not an event predicate despite Beavers referring to it as such in the text, because ϕ is the antecedent of an implication. The characterisations I give below are faithful to Beavers's intentions.

	Simplex scale (and/or event)	<i>Complex scale (and/or event)</i>
quantized	Achievements break a vase, kill Bill	Accomplishments load the wagon, eat the apple
nonquantized	N/A	Degree achievements cool the soup
potential	Semelfactives hit (once), slap (once)	Activities beat, pummel
unspecified	Semelfactives cough (once)	Activities watch TV

Table 2.2: The aspectual classes differentiated in the framework of Beavers (2002, 2008, 2011, 2013).

- c. *x* has *potential for change* iff for all *e* such that ϕ applies to *e*: $\exists s \exists \theta [\theta(x, s, e)]$ (there is not necessarily any actual change, meaning no incremental MR relation need obtain, e.g., *hit, wipe, scrub, rub x*)
- d. *x* is *unspecified for change* iff for all *e* such that ϕ applies to *e*: $\exists \theta'[\theta'(x, e)]$ (no scale, e.g., *see, smell, play (as children), ponder x*)

However, the aspectual distinctions that can be drawn on the basis of these scalar properties is insufficient to distinguish culminations from happenings. Table 2.2 reproduced from Beavers (2013, p. 692) summarizes the aspectual distinctions that are made in this framework. As Beavers (2013, p. 704) himself notes, his account does not reveal why some achievements like *arrive*, but not some others like *notice*, are felicitous in the progressive. Given that an explanation of this is one of the prime goals of the present dissertation, I need to enhance Beavers's framework.

Also, I wish to offer a scale-based semantics for the progressive according to which the progressive involves partitivity over scales of change (for the reason that event partitive analyses have been shown to face problems, cf. Section 2.2). But one cornerstone of Beavers's framework is that not all predicates are associated with a scale, which, just like authors such as Filip and Rothstein (2006), Filip (2008) and Rappaport Hovav (2008), he uses to explain various differences in the linguistic behaviour of scalar and non-scalar predicates. Predicates with no scale of change argument on the analysis of Beavers (2002, 2008, 2011, 2013) include *see, smell, play, ponder, cough once, watch TV, laugh at.* Since many of these predicates do appear in the progressive (cf. e.g., Fred *is watching TV*), it is evident that the progressive cannot refer to Beavers's scale of change.

In order to be able to account for the difference between culminations and happenings (and accomplishments) and in order to generalize the scale-based framework to all predicates, I adopt a slightly different framework as Beavers, in which I retain his scale of change as the *incremental argument*, but in addition I introduce a different, more general *scale of change* based on the notion of a *perspective* from Asher (1992) (cf. Section 2.2.5).

Chapter 3

THE POINT OF DEPARTURE FOR A NEW ACCOUNT OF PROGRESSIVE ACHIEVEMENTS

In this chapter, I present the basic considerations and formal semantic ingredients that form the starting point of my scale-based analysis of the two single-event readings of progressive achievements: preliminary process and slow-motion readings. First, I study the durativity property of eventualities and predicates and discuss the notion of an instantaneous change. Then, I examine the gradual change that is associated with culmination predicates like *arrive* to establish the relation between these predicates and corresponding events of gradual change. Finally, I introduce the basic domains and parameters of my framework that will be used in later chapters in the analysis of progressive achievements.

3.1 Durativity

In what follows, I discuss topics and present some considerations that are relevant to the semantics of achievements and which will form the starting point for my analysis of slow-motion and preliminary process interpretations. In this section, I examine the notion of durativity relevant for linguistics.

3.1.1 The linguistic relevance of durativity

The aspectual classification of eventuality predicates generally relies on features of *dynamicity*, *telicity*, and *durativity* (cf. Section 1.1). However, the status of durativity

as a linguistically relevant classificatory feature has often been called into question. Authors like Kenny (1963), Taylor (1977), Verkuyl (1989), Pustejovsky (1991) deny that achievements need to be assigned a class of their own, implying or explicitly proposing that the difference in the temporal structure of events described by achievements and accomplishments is non-existent or not important. In contrast, authors such as Vendler (1957), Comrie (1976), Dowty (1979), Moens and Steedman (1988), Smith (1991), Mittwoch (1991), Filip (1993/99), Piñón (1997), Engelberg (2000), Rothstein (2004) maintain that such a difference is linguistically relevant. The most prominent argument of those who lump achievements together with accomplishments is the two-faced behaviour of certain predicates with respect to durativity, while the argumentation of those assigning a separate class to achievements involves several differences in the syntactic and semantic behaviour of achievements and accomplishments.

Note that while telicity is a feature of predicates, rather than eventuality individuals (the very same running-event can be described both with the atelic predicate *run* and with the telic predicate *run to the river*), durativity is an inherited property of predicates, namely, a property inherited from eventualities that a predicate can be used to describe. As Comrie (1976, p. 42) remarked,

Strictly speaking, it is the situation, rather than the verb, that is punctual, though for convenience we shall retain the traditional practice of using the term 'punctual verb' for a verb referring to a punctual situation.

Or as Beavers (2008, fn. 5) put it,

Unlike telicity, which is a property of predicates, durativity is a property of events, although predicates may encode sortal constraints on durativity. Thus I refer to predicates that select for durative or punctual events as "durative predicates" and "punctual predicates" respectively.

In other words, we must differentiate the durativity of linguistic expressions (verbs and verb phrases) and the durativity of events in their extension. The debate on durativity is about the linguistic level, i.e., whether this distinction is sensible to make at the level of verbs and verb phrases. But those who argue against such a distinction (e.g., Verkuyl 1972, 1989, 1993) also maintain that there is no such thing as a non-durative *event* — and this forms one cornerstone of their argument. In contrast, I take the position that while there might not be such a thing as a non-durative dynamic event, particular events *count* as durative or non-durative in specific contexts, and the durativity feature of predicates depends on whether or not the events they describe are construed as durative or non-durative in a particular context. This means that an account of the fact that non-durative predicates can be used as durative predicates in

slow-motion scenarios as in *The Challenger is exploding* needs to be explained by an account of when particular *events* are regarded as durative or non-durative, which will be topic of Chapter 4.

And the fact that the same verb may function as an achievement in one context, and as an accomplishment in another, does not mean that we should do away with the distinction between accomplishments and achievements as aspectual classes, *pace* Verkuyl (1989, 1993). This would be like arguing against telicity as a semantically relevant feature on the basis of examples such as Dowty's (1979, p. 61) *John swam in an hour* when it is understood that John is in the habit of swimming a specific distance every day. Indeed, if telic and atelic predicates are differentiated based on differences in linguistic behaviour (such as modification by *in-* and *for-*adverbials), we should proceed no differently for durative and non-durative predicates. Achievements and accomplishments show different behaviour with respect to several phenomena, including the following (cf. Vendler 1957, Dowty 1979, Filip 1993/99, Piñón 1997, Engelberg 2000, Rothstein 2004, a.o.):

- Frame-adverbials (e.g., *in an hour*) are ambiguous between a "duration" and an "after" interpretation when they modify an accomplishment predicate, as in (3.1a), but can only have an "after" interpretation when they modify an achievement predicate, as in (3.1b).
 - (3.1) a. John will build the house in 2 years. (*after/duration*)
 - b. John will arrive/notice the picture in two minutes. (after/*duration)
- ♦ Accomplishments may appear in the progressive (cf. (3.2)), but many achievements cannot do so (cf. (3.3a)), and even if a progressive achievement is felicitous, it will have a special interpretation, as in (3.3b)–(3.3c) (cf. Section 2.1).
 - (3.2) John is building a house.
 - (3.3) a. #John is noticing Mary.
 - b. John is arriving at the station. (*preliminary process*)
 - c. Look at the screen, the Challenger is exploding. (*slow-motion*)
- Accomplishments, but not achievements, can occur as complements of aspectual verbs like to *finish*, *stop*, *start*.
 - (3.4) a. John stopped building the house. (accomplishment)

- b. #John stopped arriving/noticing the picture. (achievement)
- Accomplishments, but not achievements can be modified with adverbs of completion like *completely*, *partly*, *halfway through* (though see Section 5.4.2).
 - (3.5) a. John has partly built the house. (accomplishment)b. #John has partly arrived/noticed the picture. (achievement)

In the same way a semantic account is needed to explain when a telic or atelic use of a predicate arises, an account has to be provided how certain verbs and verb phrases can behave as a durative or a non-durative predicate in different contexts. But abolishing the punctual/durative distinction between achievements and accomplishments would come at the price of losing a uniform account of the above differences. Given that these differences do exist, it appears that the durativity distinction is, indeed, relevant linguistically, and some changes are construed as *instantaneous*.

3.1.2 The moment of change

There is a problem relating to the representation of achievements as predicates describing instantaneous change, namely, the problem of the representation in a two-valued model-theoretic framework of the *moment of change* (cf. discussions of this topic in Dowty 1979, Landman 1991 and Altshuler and Schwarzschild 2013 for details). Briefly the problem is the following. In order to model instantaneous change from $\neg \phi$ to ϕ , we need to refer to *two* states of affairs, one in which ϕ is false, and one in which it is true. But then this means that at all the time points between these two points, either $\neg \phi$ or ϕ holds. That is, if time is taken to be *dense*, there is either no final moment at which $\neg \phi$ holds, or no initial moment at which ϕ holds. We cannot select two points such that $\neg \phi$ holds at one, ϕ at the other, such that there are no other points between them. But then how can we model instantaneous change in a dense model of time?

Those who consider achievements not to be durative are therefore divided: some regard them as instantaneous, i.e., true at exactly one instant or moment of time, while others hold that achievements are true at exactly two instants. For instance, Rothstein (2004) regards achievements as true at two instants:

Achievements are genuinely near-instantaneous changes from $\neg \Phi$ to Φ , consisting of a starting point, the final instant at which $\neg \Phi$ holds and a stopping point, the adjacent instant at which Φ holds. (p. 185)

The same stance is taken by Beavers (2008, 2013), for whom achievements describe events consisting of *exactly two* parts (cf. e.g., Beavers 2013, p. 691). However, this account commits us to time being *discrete*, i.e., for every moment, there is a moment immediately preceding it and another one immediately following it.

Another way to overcome the problem of instantaneous change is to assign a special status to the moment of change, cf. the following discussion from Rothstein (2004, p. 197):

I have assumed a "conventional" theory of change in which a minimal change from $\neg \phi$ to ϕ consists of two instants, one at which $\neg \phi$ holds and an adjacent one at which ϕ holds, but Kamp (1979a,b) raises the possibility that change is a primitive notion and that an event of change holds at the instant which is the event of moving from $\neg \phi$ to ϕ , at which neither $\neg \phi$ nor ϕ holds.

Such a move entails a non-classical, multi-valued logic, which some may consider to be a high price to pay (see also the discussion of Landman 1991, §5.3 on this point). If we reject the idea that there is a moment in time at which neither ϕ , nor $\neg \phi$ holds, wishing to remain within a classical two-valued logic, and do not want to commit us to a discrete view of time, then how can we model instantaneous change from $\neg \phi$ to ϕ ?

A third solution to this puzzle is to relegate the change information to outside the instant which is taken to be the moment of change, i.e., at which an achievement predicate is taken to hold true. This is the solution of Dowty (1979), von Stechow (1996) and Piñón (1997). Dowty (1979) uses the BECOME operator to capture change, and in a dense model of time, he (initially, in a time point-based model) defines BECOME ϕ as "true at *t* iff ϕ is true at *t*, ϕ is false at *t*' for some time *t*' earlier than *t*, and for all times *t*" later than *t*' but earlier than *t* ϕ is also false at *t*"" (Dowty 1979, p. 76). Thus, BECOME ϕ and ϕ can both be true of the same instant, but their truth conditions are different. This poses no conceptual or philosophical problem when predicates are predicates of time points and intervals, but it may raise questions in an event-semantic framework if a dynamic and a stative predicate is true of the same *eventuality*.

von Stechow (1996) adopts Dowty's (1979) decompositional analysis in an eventsemantic framework and defines BECOME as follows:

Definition 4 (BECOME operator in event-semantics (von Stechow 1996))

 $\llbracket BECOME(P)(e) \rrbracket = 1$ iff e is the smallest event such that P is not true of the pre-state of e but P is true of the target state of e. The pre-state of e is the state that holds immediately before the event e occurs. The target state is the state reached at the end of the event.

This definition eludes the problem of predicating a stative and a dynamic predicate of the same event through reference to pre-states and target states. A similar solution is employed by Piñón (1997), who proposes that although achievements are true of literally instantaneous events, these are invariably the *boundaries* of some extended event.

What I aim to capture in the framework I will present (using a granularity-relative notion of durativity and instantaneousness) is the intuition that even if there might be no literally instantaneous changes, we can *conceive* of changes as being instantaneous by virtue of the fact that some differences are too small to count as relevant in a context. As for modelling instantaneous change itself (at a particular granularity), I will adopt the third line of solution presented above, because it is based on the restrictions a predicate lays on the intervals immediately preceding events in its denotation, and this is an approach which offers us a way to account for the difference between culminations and happenings, as well — which is the topic I now turn to.

3.2 The existential presupposition associated with culmination-achievements

3.2.1 Culminations as presupposition triggers

A widely accepted idea (Piñón 1997, Löbner 2002, Heyde-Zybatow 2008, Malink 2008, Abusch 2010, Martin 2011) is that some achievements — those that are the right boundary of a gradual change — *presuppose* the occurrence of an eventuality, namely, an activity. The basis for this presupposition is that achievements describe events that are boundaries of another eventuality (Mourelatos 1978, Piñón 1997, Heyde-Zybatow 2008).

The standard presupposition test of embedding under negation underpins this claim (cf. e.g., Malink 2008, Martin 2011): (3.6a), just like its positive counterpart, implies (3.6b) (though of course this implication can be cancelled, cf. below).

(3.6) a. John didn't win the race.

b. \Rightarrow John participated in the race.

We can find similar (cancellable) implications for all other culminations. For example, for *arrive*:

- (3.7) a. John didn't arrive at the station.
 - b. \Rightarrow John was going toward the station.

Accomplishment verbs do not appear to carry such an existential presupposition in English, inasmuch as no comparable entailment is found when an accomplishment verb is embedded under negation:

- (3.8) a. John didn't read the book.
 - b. \Rightarrow John was reading the book.

However, based on recent work by Zinova and Filip (2014), it is also worth investigating in more detail the presuppositional trigger status of culminations, too. According to Zinova and Filip (2014), in Russian, even though the occurrence of the activity part is implied when a perfective *accomplishment* verb is embedded under negation (indicating a presupposition), the activity part is asserted, not presupposed in the case of perfective accomplishment verbs (as other tests of presuppositions reveal), and the entailment patterns observed can be explained by a *scalar implicature* rooted in the implicative relations between the imperfective and the perfective. For one, unlike in the case presuppositions in general, no projection from the antecedent of conditionals is observed.

This is, unsurprisingly, borne out by English accomplishments, as well, as seen by the non-entailment from (3.9a) to (3.9b), adapting Zinova and Filip's (2014) example:

- (3.9) a. If John read the textbook, he will pass the exam.
 - b. \Rightarrow John was reading the textbook.

This contrasts with the behaviour of expressions with genuine presuppositions such as *quit smoking,* since beside negation and modals, conditional antecedents are an archetypal *hole* for presuppositions, i.e., presuppositions are preserved in their scope (Karttunen 1973). Thus, (3.10a) entails (3.10b):

- (3.10) a. If John quit smoking, he won't die of lung cancer.
 - b. \Rightarrow John had been smoking.

Now, in the case of achievements, we observe the pattern consistent with presuppositional expressions (cf. Abusch 2010), (3.11a) entailing (3.11b) (example from Zinova and Filip 2014), and (3.12a) entailing (3.12b).
- (3.11) a. If John won the marathon, he will celebrate tonight.
 - b. \Rightarrow John participated in the marathon.
- (3.12) a. If John arrived at the station, he met Mary.
 b. ⇒ John was going toward the station.

We observe the same pattern of behaviour for the "Hey, wait a minute test!" first suggested by Shanon (1976). He proposed that a speaker presupposes Q on uttering a sentence S if it is suitable for the hearer to respond with 'One moment, I did not know that Q!'. This is exemplified in the dialog in (3.13).

- (3.13) a. Sam quit smoking.
 - b. Hey, wait a minute: I didn't know that Sam smoked!

Based on the English versions of examples considered by Zinova and Filip (2014), we again (and quite similarly to their Russian counterparts) find non-presuppositional behaviour in the case of accomplishments, as expected, that is, (3.14b) is perhaps slightly odd in response to (3.14a):

(3.14) a. Sam read the textbook.

b. ?Hey, wait a minute: I didn't know that Sam was reading the textbook!

On the other hand, achievements, again, do appear to presuppose the occurrence of the corresponding activity part, as (3.15b) seems a legitimate response to (3.15a), and (3.16b) to (3.16a):

- (3.15) a. Sam won the race.b. Hey, wait a minute: I didn't know that Sam was participating in the race!
- (3.16) a. Sam arrived at the station.
 - b. Hey, wait a minute: I didn't know that Sam was going toward the station!

It should be remarked that some native speakers considered the dialogue in (3.14) as not that odd, which should, however, not be taken as evidence for the potential presuppositional status of the occurrence of the activity part of accomplishments (cf. the non-presuppositional behaviour in (3.9)). Instead, it indicates that the "Hey, wait a minute!" test might not be so reliable as, for instance, the test relying on the antecedent

of a conditional. This ties in with the view of Tonhauser et al. (2013, p. 81), who raised objections against "Hey, wait a minute" as a reliable test for presuppositions.

On the other hand, we can establish that the existential entailment of achievements is a presupposition rather than a *conversational implicature* by its cancellation possibilities (Beaver and Geurts 2013). Presuppositions (as in (3.17)) can be cancelled under negation (at least some presuppositions, cf. Abusch 2010, who in fact categorizes *win* as a *soft presupposition trigger* for *participating in the race*), but not in unembedded contexts, while conversational implicatures (as in (3.18)) can be cancelled when unembedded. As (3.19) and (3.20) shows, achievements appear to pattern like presuppositions (though see more on this below).¹

- (3.17) a. John didn't stop smoking. He never smoke.b. #John stopped smoking. In fact, he never smoke.
- (3.18) John ate most of the cookies. In fact, he ate them all.
- (3.19) a. John didn't win the race. He never participated.b. #John won the race. In fact, he never participated.
- (3.20) a. John didn't arrive at the station. He was never going toward the station.b. #John arrived at the station. In fact, he was never going toward the station.

Thus, although preservation of an existential entailment under negation in itself is no guarantee for presuppositional status, further tests of presupposition appear to corroborate that culmination-achievements (like *win, arrive, reach,* or *die*) do presuppose the occurrence of a specific kind of activity leading up to them. That is, presupposition tests actually reveal that the activity part leading up to these achievements is *not* lexically empty, *pace* Rothstein (2004).

3.2.2 Extra soft presuppositions and the reverse direction

There is, however, a worry with the existential presupposition of culminations that is not generally discussed, namely, that there may be cases where such culmination-

¹Native speakers have voiced their concern that (3.19b) might, in fact, be assertable in a scenario in which John won the race without participation by virtue of the fact that everyone else was disqualified. In such a scenario, however, I will predict that the progressive is infelicitous, i.e., that *John was winning the race* is not assertable prior to his winning the race by everyone else being disqualified. But this and similar considerations will be grounds below to argue that culminations are what I will call extra soft presupposition triggers.

achievements are true of an event *without* the occurrence of an event of gradual change, which I call their cover event. A case in point is Piñón's (1997) spy who dies by committing suicide (which is instantaneous, and there is no associated cover event), or the case of winning a race by virtue of everyone else being disqualified mentioned in footnote 1 (in which case, again, there is no extended cover event of participating successfully in the race). Also, Dorothy could arrive at the station in a blink of the eye by simply clicking her ruby slippers without moving towards the station. In contrast, a true presuppositional trigger like *quit smoking* can never occur without its presupposition, that is, smoking.

Note that it is a well-known fact that some presuppositions, such as the factive implication of *discover*, are weak and context-dependent and may be obviated. Abusch (2010) calls these *soft presuppositions*, and she includes achievements with a preparatory phase among soft presupposition triggers, along with, e.g., contrastive statives like *bachelor* and verbs of accompanied motion like *accompany*. However, the absence of entailment for these cases is invariably discussed in embedded contexts: in the scope of negation, question or the antecedent of a conditional. What we see in the case of culmination-achievements is that their presupposition can be obviated even in positive, non-embedded contexts, that is, the following is truthfully assertable in the disqualifications scenario:

(3.21) John won the race, even though he never even participated.

Even for (at least some of) the other soft presupposition triggers, such obviation in a positive indicative sentence is impossible (unless they are used in a metaphorical sense):

- (3.22) a. #Cardan is a bachelor, even though Cardan is not a male person.
 - b. #Cardan accompanied Grouse to the bar, even though Grouse didn't go to the bar.

Should we conclude based on the possibility of cancellation in a non-embedded context that the existential presupposition of culmination-achievements is a conversational implicature rather than a presupposition? I would not believe that is the correct conclusion, as, contrary to conversational implicatures, the implication to the occurrence of the activity part does not seem to be *calculable* from general principles of cooperative conversation, but seems to be conventionalized. In addition, the experimental work by Bill et al. (2014) also indicates that the presupposition of the culmination predicate *win* is different from scalar implicatures: they found that the inference from *The bear didn't win the race* to the bear participating in the race is much more easily cancelled than the inference from *Some of the lions have balloons* to not all of the lions having balloons. Bill et al. (2014) emphasize that this provides evidence that presuppositions and scalar implicatures should be analysed as two separate inferences based on distinct mechanisms, *pace* Chemla (2009) and Romoli (2015) who proposed a unified account of these. Of course, further experimental work is necessary to show that the existential presupposition of *win* and other culminations is different from all other kinds of conversational implicatures, as well, but I regard the lack of calculability as strong indication that the existential inference from culminations is indeed a presupposition.

However, if the inferences from culmination-achievements are presuppositions, then they are not only soft presuppositions as Abusch (2010) proposed, but based on their cancellation properties, they are of a yet different kind, namely, what I would call *extra soft* presuppositions. Thus, we can extend the hierarchy of presuppositions proposed by Abusch (2010) as follows:

- hard presuppositions cannot be cancelled;
- soft presuppositions can be cancelled in embedded contexts;
- extra soft presuppositions can be cancelled.

For example, while culminations like *arrive* are extra soft presupposition triggers for the occurrence of the activity (typically) leading up to them, they are at least soft presupposition triggers for their pre-state: *John arrived at the station* (just as *John didn't arrive at the station* and *If John arrived at the station, then...*) implies that John was not at the station beforehand, and this implication cannot be cancelled.

What I would like to point out, though, is that whenever there is in fact no cover event (the death of the spy, winning by others being disqualified, arriving somewhere magically), then the relevant achievement in the progressive cannot be asserted truthfully, that is, it is in these cases not true prior to culmination that, e.g., the spy was dying, that John was winning the race or that Dorothy was arriving at the station. Thus, the availability of a progressive for the achievement is inherently entwined with the occurrence of a cover event of a specific type (or more precisely, of some type from a specific set of types).

I propose that we can account for this connection and the extra soft presuppositional trigger status of culminations by thinking in a reverse temporal direction, drawing on the idea voiced by Mourelatos (1978, p. 417) that "there cannot be an accomplishment without a closely related end-point achievement—one cannot say »I wrote/shall write the letter «if he cannot say »I finished/shall finish the letter«". We can say that *whenever* a gradual change belonging to a set of specific types occurs (e.g., for dying, a detrimental deterioration of health or decreasing amount of oxygen supply to the body resulting in loss of life; for winning, successful participation in a race, coming in first place; for arriving somewhere, travelling, driving or walking somewhere), then its culmination can be described with the relevant culmination predicate (*die, win, arrive*), *without exception*. Then a soft presupposition arises if all of these gradual changes entail it: e.g., all gradual changes leading up to an arrival entail movement toward the goal, which is what is presupposed by *arrive*.

This direction of the entailment establishes a close connection between a culmination predicate and gradual changes of a specific type (and nothing precludes there to be multiple types satisfying this entailment, cf. the different cases listed for dying and arriving above), while leaving space for culminations happening without any cover event. But a preliminary process progressive of a culmination is only available if there is a cover event of the relevant type in progress, because only then is there a gradual change leading to the culmination.

Such a reverse thinking about the relevant entailments is similar to the *reverse analysis* of the progressive exploited first by Varasdi (2010, 2014), who argues that the old observation (cf. Dowty 1979) that the entailment from the progressive to the perfective is not valid, while the reverse entailment from the perfective to the progressive is, should form the centre of an account of the progressive. Similarly, I argue that the entailment from the occurrence of a gradual change to a culmination, rather than the non-valid entailment from a culmination to the occurrence of some activity, can form the basis for an account of the extra soft presuppositional trigger behaviour of culminations and an account of their progressive interpretation.

This is the way in which I embed the observation of authors like Carlson (1981) Filip (1993/99), Dini and Bertinetto (1995), Heyde-Zybatow (2008) and Martin (2011) that it is culmination-achievements that are acceptable in the preliminary process progressive. But my account of the progressive exploits a *reverse* thinking — in comparison to the approaches of authors like Heyde-Zybatow (2008) and Martin (2011) — about the relation between a culmination and a gradual change preceding it. Another respect in which my account differs from previous analyses is that the relation between an achievement and its cover events will be mediated through *scales of change*. The way I will capture the relation between events of gradual change and culminations like *arrive* is by characterising culminations with two kinds of scale of change,

namely, extended scales associated with gradual change, and binary scales associated with non-gradual change. The benefit of this dual characterisation is that we can retain the idea that culminations denote sets of instantaneous events, and so we have no problem explaining that they behave like non-durative predicates in most linguistic contexts (e.g., when modified by temporal adverbials); at the same time, by having the truth conditions of the progressive depend on scales of change instead of events directly, we can explain why culminations, but not happenings, can combine with the progressive at an everyday granularity.

3.3 The basic ingredients

In this section, I present the basic assumptions and components of the semantic framework that is presupposed by my analysis of slow-motion and preliminary process progressives, motivated by the considerations about durativity and the existential presupposition of culminations discussed in the foregoing.

3.3.1 The semantic model

Philosophical commitments. This is a dissertation in linguistics, and the same linguistic phenomena could be approached on the basis of very different philosophical principles. However, in this dissertation I will build upon the following general assumptions. I assume a Neo-Davidsonian event-semantic semantic framework with realist commitment as Parsons (1990), that is, I assume that a model contains a set of eventualities (including states) that are *concrete particulars* just like physical objects. Events, similarly to physical objects, exist independently of the language used to describe them and the mind of language users. (Note that a realist need not assume that, for instance, the accomplishment/activity distinction is encoded at the level of event particulars — this is an assumption I, in fact, do not endorse.)

On a realist approach, with some simplification, the following basic commitments are accepted, which have implications for the framework and analysis I will present:

- there is a mind-independent reality in which entities and events have properties of their own;
- denotation is a correspondence between language and the mind-independent reality;
- sentential meaning is truth-conditional.

A verbal predicate of a natural language denotes a subset of the set of eventualities and its semantics is characterised through reference to this set. This is the general approach of model-theoretic semantics, whether or not one endorses realism (cf. Zimmermann 2011): the meaning of a predicate of the object language is characterised in terms of the model. Semantic explanations make use of what we assume about reality: what kind of properties a particular entity bears. Some of these properties are *cognitively relevant*, and therefore have a reflection in natural language. I assume, for example, that the length that an event takes is such a property. Of course, how a certain property is reflected in the semantics of a natural language construction depends equally on the natural language in question. For example, the length of an event may get expressed depending on the particular linguistic context and/or lexical semantic features of a construction.

Domains and parameters of interpretation. In the framework I use, expressions are interpreted with respect to a model M, a possible world w from the set of possible worlds W, a variable assignment function g, a context c from the set of all possible discourse contexts C and a granularity parameter $\{\gamma_i\}$ that is a set of granularity functions over different scales from the set \mathbb{G} of all granularity functions.²

The context parameter *c* of interpretation is the same as in Asher (1992), and will be used to capture the properties of the change associated with an eventuality that is relevant in a given discourse context through the context-relative function $\sigma_{c,w}$ — to be introduced in Section 5.1 — that outputs the scale of change associated with an eventuality in a given context. This function will be used for an analysis of preliminary process progressive sentences like *John is arriving*. The context parameter will also be implicitly used in the definition of the progressive operator itself, which, building on Varasdi (2014), I assume makes reference to a contextually determined set of alternative predicates.

The granularity parameter $\{\gamma_i\}$ is based on the granularity function parameter of information of Sauerland and Stateva (2007, 2011) and will be discussed in Section 4.1. This parameter is used in an analysis of granularity-based changes in the durativity feature, and by extension, slow-motion readings like *The Challenger is exploding*.

The domains of the model are the set U of *individuals*, set E of *eventualities* with a dedicated subset $E_{\text{stat}} \subseteq E$ of *states*, and the set T of *time points* (or *moments*, or *instants*). Note that I follow the Neo-Davidsonian approach of Parsons (1990) who as-

²Usually, I will only indicate the single granularity function that is relevant for interpretation, for simplicity.

sumes that any kind of verbal predicate can introduce an eventuality argument, even states — hence the subset E_{stat} within our domain of eventualities. However, while many scholars today assume that all states introduce a Davidsonian eventuality argument (for a discussion, see Maienborn 2005), I follow Kratzer (1995) and assume that it is in fact only stage-level states, but not individual-level predicates, that do so.

The set of individuals and eventualities are both ordered with respect to the usual *part-of* (or mereological) relations \Box_U and \Box_E over individuals and eventualities, respectively, forming a join-semilattice as in Link (1983) and Bach (1986). The set of times is linearly ordered by the usual *earlier-than* relation $<_{\mathcal{T}}$, forming the linearly ordered set $\mathcal{T} = \langle T, <_{\mathcal{T}} \rangle$. The set of *time intervals* (or *periods*) $\operatorname{Int}(\mathcal{T})$ is formed in the usual way as convex subsets of T. (For the definitions of generic mathematical terms, here and in what follows, I refer the reader to Appendix A.) I will also use the term *token* for individuals and *type* for predicates.

For the purposes of the dissertation, I do not need a dedicated domain of scales. Instead, scales are constructed from the domains of the model.

3.3.2 Scales

Recall from Section 2.3.1 that formal semantic theories using scales differ in a number of respects, and perhaps most importantly, whether or not they assume that the degrees of scales are measurement points, i.e., whether or not the distance between two degrees can be measured. I aim to offer an analysis which can be incorporated into all scale-based approaches, and to this end, I use a definition of scales that is a common denominator across all approaches. In particular, I adopt the most simple conception of scales as follows:³

Definition 5 (Scale)

A scale S is an ordered set $(S, <_S)$, where S is a nonempty set of points (called degrees or values), and $<_S$ is a linear order over S, such that S has the greatest lower bound and least upper bound properties.

The requirement that S has the greatest lower bound and least upper bound properties ensures that each nonempty subset of S that is bounded from below has an

³Since for the sake of generality I do not subscribe to the idea that the set *S* of degrees is a subset of the real numbers, the dimension component δ in a common definition of scales (cf. Section 2.3.1) does not play a role in my framework, and is therefore dropped. The subscript *S* of $<_S$ can also be suppressed if no confusion arises from using simply <.

infimum, and each nonempty subset which is bounded from above has a supremum (see Appendix A). This is implicitly accepted by all scholars using scales in formal semantics, and will be necessary for defining granularity functions over scales in Section 4.1.1.

No other properties of a scale are constrained. A scale can be either open or closed and bounded or unbounded (in either directions), finite or infinite and dense, discrete, or neither dense nor discrete. The degrees of a scale need not be numbers or any abstract degrees of measurement. A metric is defined over and above scales through an independent mapping, namely, the *numerifying function* num that I will introduce in Section 4.1.1. Those who wish to adopt an approach on which the degrees of the scales inherently define a metric can assume that this numerifying function is defined for all scales, and the metric it defines over a scale is identical to the inherent metric of the scale.⁴

Following common practice in referring to ordered sets, I will use the term *scale* to refer to both the relational structure $S = \langle S, \langle S \rangle$, as well as its carrier set S. The function carrier outputs the carrier set S of a scale $\langle S, \langle S \rangle$:

(3.23) If
$$S = \langle S, \langle S \rangle$$
 then carrier $(S) \stackrel{\text{def}}{=} S$.

In the remainder of the dissertation, I restrict my attention to two kinds of scales: those whose degrees are *elements* of a domain of the model (i.e., elements of U, E or T) and those whose degrees are *subsets* of a domain of the model (i.e., elements of $\wp(U)$, $\wp(E)$ or $\wp(T)$). So let let $\mathcal{X} = \{U, E, T, \wp(U), \wp(E), \wp(T)\}$. Then the set of all scales, notated as \mathbb{S} , is defined formally in (3.24).

(3.24)
$$\mathbb{S} \stackrel{\text{def}}{=} \bigcup_{D \in \mathcal{X}} \left\{ \langle S, \langle S \rangle \mid S \subseteq D \text{ and } \langle S \text{ is a linear order over } S \right\}$$

An important subset of the set S of all scales is the set S_{stat} of those scales whose degrees are *sets of states*:

(3.25)
$$\mathbb{S}_{\mathsf{stat}} \stackrel{\text{def}}{=} \left\{ \langle S, \langle S \rangle \in \mathbb{S} \mid \forall s \in S \colon s \in \wp(E_{\mathsf{stat}}) \right\}$$

⁴This is just the most direct way to integrate my proposal in such a framework, the more natural one would be just to do away with the numerifying function and redefine all definitions in which it features in a way that the definitions refer directly to the degrees of the scales.

These are the potential candidates for *scales of change* that I will introduce in Section 5.1. The scale of change and, as I will argue in Section 4.2, the temporal trace τ are on the present framework relativised to granularities, the purpose of which is to model the slow-motion interpretation of progressive achievements.

 \mathbb{D} will be used to designate the set of all *degrees* of all scales (defined in (3.26)).

$$\mathbb{D} \stackrel{\text{def}}{=} \bigcup_{\langle S, \langle S \rangle \in \mathbb{S}} S$$

Any convex subset (i.e., interval) *i* of the carrier set of a scale $\langle S, \langle S \rangle \in S$ is also a scale (with the ordering over it being the restriction of $\langle S \rangle$ to interval *i*), and it is called a *subscale* or subinterval of S. Of special interest will be bounded subscales, so I use \mathbb{B} to designate the set of all *bounded intervals* (see Appendix A) of all scales (see (3.27)).

(3.27)
$$\mathbb{B} \stackrel{\text{def}}{=} \bigcup_{\mathcal{S} \in \mathbb{S}} \{i \mid i \in \mathsf{Int}(\mathcal{S}) \land i \text{ is bounded}\}$$

The part-of relation over scales corresponds to the subset relation. Just as in the case of events, the restriction of the part-of relation to non-final parts is an important relation, which will feature in my analysis of the progressive, so I define it below.

Definition 6 (Non-final subscale)

A scale S' is a non-final subscale of scale S, notated as $S' \subset_{nf} S$ iff $\operatorname{carrier}(S') \subset \operatorname{carrier}(S)$ and $\langle_{S'}=\langle_{S}|$ carrier(S') (that is, the ordering of S' is the restriction to its carrier set of the ordering of S) and $\exists d \in \operatorname{carrier}(S) : \forall d' \in \operatorname{carrier}(S') : d' \langle_{S} d$ (where $\langle_{S} is$ the strict ordering of S).

I also define the initial subscale relation as follows, using the *non-strict* subset relation in this case:

Definition 7 (Initial subscale)

A scale S' is an initial subscale of scale S, notated as $S' \subseteq_{ini} S$ iff $carrier(S') \subseteq carrier(S)$ and $\langle_{S'}=\langle_{S}|$ carrier(S') and $\forall d' \in carrier(S'): \forall d \in carrier(S): d \leq_{S} d' \rightarrow \exists d'' \in carrier(S'): d'' \leq_{S} d$ (where \leq_{S} is the non-strict ordering of S). **Scale complexity.** The number of degrees in a scale will play an important role in the present framework. Building on Beavers (2002, 2008, 2013), I make the following distinctions in scale (or scalar) complexity:

No. of deg.	Scale complexity
2<	<i>multi-valued</i> (or <i>complex</i> or <i>extended</i>)
2	<i>binary</i> (or <i>two-valued</i> or <i>simplex</i>)
1	singleton (or one-point or degenerate)

Table 3.1: Scale complexity

Time as a scale. Taking Definition 5 of a scale as a set of linearly ordered points (satisfying the greatest lower bound and least upper bound properties), time straightforwardly qualifies as a scale. Conceiving of time as a scale is not at all a novel idea, as its scalar aspect is highlighted by gradable adjectives and adverbials like *late, early* (cf. von Stechow 2009a), and *before, after*. And the α *time* constituents in time adverbials like *for two hours* are *measure phrases*, denoting the intervals of time of the relevant size (cf. Krifka 1989, Schwarzschild 2002).

For our purposes, the point of conceiving of time as a scale is that this will enable us to apply to time the framework of granularity functions of Sauerland and Stateva (2007, 2011), which operate on degrees of a scale. Sauerland and Stateva (2007, 2011) already made use of the scalar aspect of time in this respect, in discussing temporal expressions such as *at 6 A.M.*, which exhibit the scalar vagueness that they analysed with granularity functions: *Jane arrived at six A.M.* can be true if Jane, in fact, arrived at 5:58. My goal in exploiting granularity functions over time is to model how the same event can be instantaneous in one context — in which a coarse granularity over time is assumed, and the endpoints of the event are not mutually distinguished — and durative in another context — in which a fine granularity over time is assumed, and the endpoints of the event are mutually distinct.

Chapter 4

SLOW-MOTION INTERPRETATIONS

In this chapter, I focus on slow-motion interpretations and offer an analysis of shifts between punctual and durative interpretations that are rooted in a change in granularity. To this end, I first adopt and enhance formal tools for analysing granularity-relative phenomena, namely, granularity functions which are parameters of interpretation mapping a degree on a scale to an interval containing it. Then I propose that durativity is not a property of eventualities and predicates *per se*, but is dependent on a granularity parameter: in particular, the temporal trace function is relativised to a temporal granularity parameter. The final part of the chapter is then devoted to the discussion of some implications of the proposal for phenomena not related to achievements, including the relation between vagueness and durativity, the vague temporal boundaries of events and the minimal parts problem of activities.

4.1 The granularity parameter

Slow-motion readings of progressive achievements like *Look at the screen, the Challenger is exploding* arise contexts in which fine temporal distinctions are relevant, that is, in "fine granularity" contexts. To model slow-motion readings, therefore, we need to use tools for modelling "granularity". In this section, I will describe the formal tools that can be used to capture the granularity-relativity of the interpretation of linguistic expressions in general. I will use the granularity functions of Sauerland and Stateva (2007, 2011), but I will introduce some enhancements and formal details that were left implicit or were not addressed by the authors.

4.1.1 Granularity functions

Building on Krifka (2007), Sauerland and Stateva (2007, 2011) (henceforth, S&S) propose to handle a form of vagueness (what they call *scalar vagueness*, cf. Section 4.3.1 below) through *granularity functions*, which they use as contextual parameters of interpretation. On their account, a granularity function maps a point of a scale to an interval containing it in a way that it partitions a scale into cells of equal size. Then the cells of the partition consist of points that are indistinguishable at that granularity.

Building on S&S, but modifying their formal characterisation somewhat,¹ a function γ_{S} : carrier(S) \rightarrow Int(S) is a *granularity function* over scale $S \in S$ if it satisfies the following restrictions, where num outputs a number for each point on a scale and which will be given a precise definition below, sup is the supremum, inf is the infimum and $-_{\mathbb{R}}$ is the usual subtraction operator over real numbers:

(4.1a)
$$\forall s \in \mathsf{carrier}(\mathcal{S}) \colon s \in \gamma_{\mathcal{S}}(s)$$

(4.1b)
$$\forall s \in \operatorname{carrier}(\mathcal{S}) \colon \gamma_{\mathcal{S}}(s) \text{ is convex}$$

(4.1c)
$$\forall s, s' \in \operatorname{carrier}(\mathcal{S}): \operatorname{num}(\sup (\gamma_{\mathcal{S}}(s))) -_{\mathbb{R}} \operatorname{num}(\inf (\gamma_{\mathcal{S}}(s))) = \operatorname{num}(\sup (\gamma_{\mathcal{S}}(s'))) -_{\mathbb{R}} \operatorname{num}(\inf (\gamma_{\mathcal{S}}(s')))$$

(4.1d)
$$\forall s, s' \in \operatorname{carrier}(\mathcal{S}) \colon \gamma_{\mathcal{S}}(s) = \gamma_{\mathcal{S}}(s') \text{ or } \gamma_{\mathcal{S}}(s) \cap \gamma_{\mathcal{S}}(s') = \emptyset$$

(4.1a) says that the set to which a granularity function γ_S maps a point has to include the point as element. (4.1b) states that the range of γ_S is made up of convex sets, i.e., intervals. (4.1c) requires the images of all points by γ_S to be of the same size (i.e., a granularity function defines a unit on the scale). (4.1d) is a fourth criterion on granularity functions which is at least implicitly assumed in S&S, and says that a granularity

¹First, (4.1d) is only implicit in S&S. Second, (4.1c) differs from S&S's formula, who use max and min, I assume for a maximum and minimum. The reason I depart from S&S and use suprema and infima, is that I wish to remain neutral as to which end(s) of the intervals can be open. S&S intend a granularity function to *partition* a scale, but given a dense scale, it is impossible for two neighbouring cells both to include the boundary. For instance, if one cell of the partition is [4.5, 5.5], then the following cell must be a left-open interval (5.5, 6.5] or (5.5, 6.5), that is, it cannot have a minimum.

In terms of the measure of the interval length, using suprema and infima instead of maxima and minima makes no difference, because the only case the supremum is not necessarily a maximum (and an infimum not necessarily a minimum) is when the interval is dense, but in that case, the measure of the open interval not containing the supremum and/or infimum (e.g., (5.5, 6.5)) and the measure of the closed interval containing either of those (e.g., [5.5, 6.5]) is the same.

In addition, I use the additional function num to mediate between scales and numeric operations like subtraction.

function *partitions* a scale, that is, the images of the points of a scale are either identical or non-overlapping. In other words, granularity functions define a partition of a scale with cells of the same size — in the terminology of Deo (2009), a *regular partition*.²

Note that the definition of granularity functions, in particular, (4.1c) presupposes the existence of a supremum and an infimum for the bounded subintervals of a scale. This is guaranteed on our framework by having Definition 5 require that scales have the least upper bound and greatest lower bound properties.

I will use G_S to designate the set of all granularity functions defined over a scale $S \in S$:

(4.2)
$$G_{\mathcal{S}} \stackrel{\text{def}}{=} \{ \gamma_{\mathcal{S}} : \operatorname{carrier}(\mathcal{S}) \to \operatorname{Int}(\mathcal{S}) \mid \gamma_{\mathcal{S}} \text{ satisfies the criteria in (4.1)} \}$$

The set \mathbb{G} of all granularity functions in a model is the set of all granularity functions over all scales in a model:

(4.3)
$$\mathbb{G} \stackrel{\text{def}}{=} \bigcup_{\mathcal{S} \in \mathbb{S}} G_{\mathcal{S}}$$

I will drop the scale subscript of γ if it is obvious from the context which scale the granularity function γ is defined on.

Defining a metric. A technical note is due at this point. S&S appear to follow the school of scale-based semantics according to which the carrier set S of a scale is a subset of the real numbers, or at least are abstract points of measurement with an inherent metric. Although many authors make this assumption, it is not uniformly accepted (cf. Section 2.3.1), and wishing to retain greatest generality, I myself defined a scale in Definition 5 as any linearly ordered set of points, which means that a scale does not necessarily have a metric on it inherently. Then, however, we need a measure on the scale to mediate between the scale and numeric operations like the subtraction in (4.1c), similar to the *length scaling* function of Gawron (2009) that assigns a real number to the points of scales (or in Gawron's terminology, *axes*).³

²Deo (2009) uses regular partitions over time to provide a suitably restricted domain for quantification for aspectual operators like the progressive. However, she does not use a function like granularity functions to assign cells of that partition to specific (time) points (or intervals), and for this reason the formal machinery she uses is insufficient for the more general purposes to which granularity functions can be put to use.

³van Rooij (2011b) also assumes that linear orderings form the input for an independent quantitative measurement, that is, measurement is defined over and above the basic linear order defined by, say, differences in tallness or time.

In order to define granularity functions as regular partitions, that is, express that granularity functions partition a scale into units of *equal size*, we first need to map the points of a scale to (real) numbers. This mapping is done by a numerifying function I notate as num, which is defined as follows.⁴ Let us define the *numerifying function* num: $\mathbb{D} \to \mathbb{R}$ from the set of all degrees \mathbb{D} to the set of real numbers \mathbb{R} such that it satisfies the following homomorphism requirement (where \mathbb{S} is the set of scales, $<_S$ is the ordering defined over the scale S and $<_{\mathbb{R}}$ is the usual strict smaller-than relation over the real numbers):

$$(4.4) \qquad \qquad \forall \langle S, \langle S \rangle \in \mathbb{S} \colon \forall s, s' \in S \colon s <_S s' \to \mathsf{num}(s) <_{\mathbb{R}} \mathsf{num}(s')$$

Apart from the homomorphism requirement, num is also required to define a *metric* (or *distance function*, see Appendix A) d_S for any scale S such that

$$(4.5) \qquad \forall \langle S, \langle S \rangle \in \mathbb{S} \colon \forall s, s' \in S \colon d_S(s, s') = \mathsf{abs}(\mathsf{num}(s) -_{\mathbb{R}} \mathsf{num}(s'))$$

where abs is the absolute value function over the set of real numbers and $'-_{\mathbb{R}}'$ is the usual subtraction on real numbers.

The numerifying function, unlike a measurement unit like *litre*, is not conventional and is not used in language explicitly. Also, just as 2°Celsius and 2 litres have nothing in common apart from the number itself, that is, the fact that they contain the same numeral does not reflect any deeper connection between them, assignment of the same number to two points on two different scales does not carry any information, either. The numerifying function is just an abstract, arbitrary measurement over scales, whose sole purpose is to define a metric over different scales.⁵

⁴num is not to be confused with a *measure function*, which maps objects to degrees, or a *measurement unit*, which maps objects to numbers.

⁵ Depending on our choice, we can have num be defined for *all* scales, as it now is, in which case the framework is equivalent with the approach in scale-based works according to which distance measures are inherently defined on all scales (cf. Section 2.3.1). If, on the other hand, we accept the arguments of Cresswell (1976), Bale (2011), van Rooij (2011a) that not all scales are associated with a metric, the function num can be simply assumed to be a *partial function* over the set of scales, not defined for, e.g., the scale of beauty or happiness.

Given the present formal account of granularity, the latter choice predicts that expressions referring to scales of beauty or happiness are not sensitive to granularity, because the application of *granularity functions* presupposes a metric (cf. requirement (4.1c)). As such, assuming that *strictly speaking* is a "more precise scalar approximator" in the sense of Sauerland and Stateva (2007, 2011), i.e., it indicates a change to a finer granularity context, we expect that it should be much odder to assert a sentence with *strictly speaking* containing a comparative of *beautiful* than a comparative of *tall*. And this appears to be borne out:

That said, the numerifying function may take the place of Bale's (2008) universal scale of comparison which he uses to account for inter-scalar comparisons, such as *Esme is more beautiful than Einstein is intelligent*. On our account, there is no need for a separate universal scale of comparison, because the set of numbers to which scales are mapped by the numerifying function can take on its function. And as the present discussion shows, there is independent need for such a numerifying function beside indirect comparisons, namely, to be able to measure distances and thereby ultimately model granularity changes.⁶

Representing an interval with a point. As it stands, the output of granularity functions are intervals, and not degrees of the scale. Solt (2014), who also uses the framework of S&S, draws attention to this fact, but reference to these intervals that she calls "coarse-grained degrees" is sufficient for her purposes. These "coarse-grained degrees" are intervals, and, as such, unlike points, but also "indivisible", and, as such, unlike regular intervals. Thus, using simply these intervals that are the output of granularity functions is a "departure from the more typical view of degrees as points, and also from existing theories that treat degrees as intervals" (Solt 2014, p. 528).

But for our purposes, we need to refer to regular, pointlike degrees at a coarse granularity over the scale (for instance, in order to be able to have the numerifying function num apply to them, which cannot, formally, take intervals as its argument). I therefore follow van Rooij (2011b), who models granularity changes through set inclusion over the set of individuals, and uses a *single individual* to represent an *equivalence class* of individuals of a finer-grained model at a coarser granularity. I in turn represent an equivalence class of degrees under a given granularity function with a single distinguished degree. To formally model this, all degrees that are mapped to the same interval by a granularity function are mapped to this single, distinguished degree (for which I choose the supremum) by a function called the *degreeifying function*.

⁽i) a. Strictly speaking, Anna is **taller** than Bill, but come on, their height difference is so negligible, they're of equal height for all intents and purposes!

b. ?Strictly speaking, Mittens is **more beautiful** than Coco, but negligibly so, and so they are equally beautiful for all intents and purposes.

As in my thesis I only consider scales over which granularity functions are defined, I ignore this and have num be defined for all scales for simplicity.

⁶And further motivation for such a numerifying function is given by Sassoon (2010) and Solt and Gotzner (2012) who bring arguments supporting the assumption that most scales used in natural language semantics have a metric defined over them.

So we define the (partial) *degreeifying function* deg: $\mathbb{G} \times \mathbb{D} \to \mathbb{D}$ from the set of all granularity functions \mathbb{G} and the set of all points \mathbb{D} such that it outputs a point, the representative of the interval to which the granularity function argument maps the point argument. The following criteria apply to deg, where sup is the supremum of an interval (see Appendix A):

(4.6a)
$$\forall s \in \mathbb{D} \colon \forall \gamma \in \mathbb{G} \colon \deg(\gamma, s) \text{ is defined iff } \gamma \text{ is defined in } s$$

(4.6b) $\forall s, s' \in \mathbb{D} \colon \forall \gamma \in \mathbb{G} \colon \gamma(s) = \gamma(s') \leftrightarrow \deg(\gamma, s) = \deg(\gamma, s')$

(4.6c)
$$\forall s \in \mathbb{D} \colon \forall \gamma \in \mathbb{G} \colon \deg(\gamma, s) = \sup(\gamma(s))$$

(4.6a) ensures that deg is defined only if its granularity function argument is defined over the scale that its point argument is an element of. (4.6b) is the criterion that ensures mapping all and only points in the same interval to the same point (the selected point represents the interval). (4.6c) ensures that the degreeifying function outputs the supremum of the interval as a representative of that interval (which is an arbitrary decision, we could equally well have chosen the infimum).

The degreeifying function deg is a technically necessary tool for measuring the distance of two points at a particular granularity, and will be used in the definition of granularity-relative temporal traces in Section 4.2.

Figure 4.1 offers a schematic overview of the mappings introduced here that are necessary and collectively sufficient to model granularity in the present scale-based framework which does not assume an inherent metric over scales.



Figure 4.1: The mappings assumed in the framework for modelling granularity

4.1.2 Higher-order granularity parameters

4.1.2.1 The problem of sharp boundaries

A point to be noted in connection with the granularity functions of S&S is that their analysis implies that although some points are not distinguishable at a given granularity (those that are mapped to the same interval by that granularity function), some nearby points belonging to different intervals *are*. An illustration of this problem is given in Figure 4.2.



Figure 4.2: The return of the Sorites paradox

Let us take the following example from S&S. On the scale of distance, they define three granularity functions, a fine, an average and a coarse granularity, which map the degree denoted by the expression *5 meters* to the following intervals:

 $\gamma_{\text{fine}}(5\text{m}) = [4.95\text{m}, 5.05\text{m}]$ $\gamma_{\text{mid}}(5\text{m}) = [4.75\text{m}, 5.25\text{m}]$ $\gamma_{\text{coarse}}(5\text{m}) = [4.5\text{m}, 5.5\text{m}]$

As can be seen, the unit of the fine granularity function is the smallest, and the image of a particular point is properly included in the image thereof by the coarser granularity functions.

However, with the granularity parameter set to "mid" in the 5 meter example above, a 4.74m-long rod would not qualify as 5 meters long, while a 4.75m-long one would. That is, although granularity functions successfully erase the unwanted distinction between some points (say, 4.75m and 4.76m), they do not do so for all unwanted distinctions (say, 4.74m and 4.75m). This is a serious shortcoming of granularity functions as a tool for analysing vagueness: they do not refer to the distance between degrees, but output arbitrary boundaries.

This is a replication of the Sorites paradox (at what point can we say that a set of sand particles is a heap?) at a higher level:⁷ the problem is that there is still a precise boundary between two sets of (indistinguishable) elements. However, another

⁷Cf. also the discussion about the higher-order Sorites paradox in van Rooij 2011b.

well-known theory of vagueness, that of supervaluations proposed by van Fraassen (applied in an analysis of vagueness first by Fine 1975), also suffers from an analogous problem,⁸ and short of introducing fuzzy logic or some form of continuous distribution into semantic analysis (such as in the recent probabilistic approaches of Lassiter 2011b or Sutton 2015), there appears to be little hope of overcoming this obstacle.⁹

One way to mitigate the effect of the higher-order Sorites paradox and avoid some of the problems posed by the return of sharp (as opposed to the expected vague) boundaries is, I propose, to combine granularity functions with a supervaluation-like method. Since both granularity functions and supervaluations serve to *relegate* unwanted distinctions to a higher level, superimposing a supervaluation-like approach over granularity functions relegates the unwanted distinctions made by granularity functions to a yet higher level, as I will illustrate. It should be noted at the outset that this move is not a resolution of the Sorites paradox, which, as I noted above, appears to defy a solution in existing frameworks, and the best we can achieve is to do away with as many unwanted distinctions as possible.

4.1.2.2 Minimal distinguishability

The rationale behind combining the theory of granularity functions with a supervaluation-like method is that what "granularity" (in the intuitive sense of the term) is about is the *size* of the differences that are relevant at that particular level of granularity, rather than a specific assignment of individuals and degrees to equivalence classes. Take the three glasses of water in Figure 4.3. At the everyday granularity (say, in a pub



Figure 4.3: Three glasses of water

scenario), the liquid content of the last two glasses on the right would be regarded as equal (even if the agent is told about the 2 cm^3 difference between them). But there

⁸E.g., the boundary between the amounts of sand particles which are *strictly* a heap, and the amounts which are a heap according to *some but not all* valuations is sharp.

⁹Indeed, as Peter Sutton (p.c.) noted, even probabilistic approaches fail to offer a way out of the Sorites paradox, because there is always a sharp cutoff-point when the probability of one predicate — say being red — applying to an individual is greater than the probability of another predicate — say, being orange — applying to it.

may well be scenarios, where this very difference is relevant, such as in a laboratory context, where small differences can be crucial when creating a new liquid solution.

Importantly, the difference between the last two glasses on the right can be relevant merely owing to the fine granularity enforced by the laboratory setting, irrespective of whether the first glass is also present. That is, the set of individuals (in our example: the set of glasses) forming the contrast set for a given judgement (in our example: whether the third glass contains more liquid than the second) does not determine whether a given difference (in our example: 2cm^3) counts as relevant in a context. Therefore, approaches to vagueness and granularity relying solely on comparison classes such as Burnett (2014)¹⁰ will not be sufficient to explain the full range of facts. A similar observation was made by Solt and Gotzner (2012), who bring experimental evidence for the insufficiency of a simple delineation semantics (i.e., one relying solely on contrast classes) for gradable adjectives like *tall* or *dark*.

Consider now the *finer than* relation \prec over granularity functions as defined by S&S which satisfies the following criterion (in our framework) for all scales $S \in \mathbb{S}$: $\gamma_S \prec \gamma'_S$, that is, γ_S is *finer than* γ'_S if and only if

(4.7) $\forall s \in \operatorname{carrier}(\mathcal{S})$:

 $\mathsf{num}(\sup(\gamma_{\mathcal{S}}(s))) -_{\mathbb{R}} \mathsf{num}(\inf(\gamma_{\mathcal{S}}'(s))) <_{\mathbb{R}} \mathsf{num}(\sup(\gamma_{\mathcal{S}}'(s))) -_{\mathbb{R}} \mathsf{num}(\inf(\gamma_{\mathcal{S}}'(s))).$

In words, a granularity function γ_S is finer than γ'_S if and only if the size of its units is smaller than the size of the units of γ'_S . Crucially, S&S themselves regard the unit sizes as definitive for granularity precision, rather than actual mappings, since they do not base the *finer than* relation on set inclusion (that is, they do not say that $\gamma_S \prec \gamma'_S$ iff the images of any point by γ_S is a subset of the image of the point by γ'_S).

In order to be able to refer to the unit size of granularities, let us introduce the following notion.

Definition 8 (Minimally distinguishable interval size)

Let us define the function $mdi: \mathbb{G} \to \mathbb{R}$ from the set of granularity functions to the set of real numbers such that $mdi(\gamma_{\mathcal{S}}) = num(sup(\gamma_{\mathcal{S}}(s))) -_{\mathbb{R}} num(inf(\gamma_{\mathcal{S}}(s)))$, where s is an arbitrary point of carrier(\mathcal{S}). $mdi(\gamma_{\mathcal{S}})$ is called the minimally distinguishable interval size (MDI, for short) by granularity function $\gamma_{\mathcal{S}}$.

¹⁰Burnett's (2014) Contrast Preservation principle requires the presence of a third individual if two individuals become indistinguishable, i.e., on our example, the first glass needs to be present in order for the last two to be distinguishable in one setting and indistinguishable in another.

The essence of granularity is reference to a unit size (or grain size), and on the present framework, this unit size is encoded as the minimally distinguishable interval size MDI. The MDI presents a limit on the difference between degrees which are indistinguishable at a given granularity γ : any interval of a size greater than $mdi(\gamma)$ is necessarily extended, because its infimum and supremum are necessarily mapped to different intervals by γ , and hence to different degrees by deg. This will be the formal basis for modelling granularity-based durativity changes in Section 4.2.

Comparison with other characterisations of a unit size. There are several independent strands of research bearing on the present notion of a MDI. On the account of Lasersohn (1999), the MDI corresponds to the size of the pragmatic halo associated with an expression in a given context, i.e., the difference (on some dimension) which counts as "pragmatically ignorable", although Lasersohn does not use an explicit measure of this ignorable difference in his formal account. Where the present proposal differs radically from that of Lasersohn (1999) is that it is a semantic account of granularity changes, rather than a pragmatic one. As such, it is immune to an objection of van Rooij (2011b) and Solt (2014) against pragmatic halos, noted (but dismissed as not crucial) by Lasersohn (1999, §6.1) himself: according to this objection to a pragmatic account of vagueness like that of Lasersohn (1999), it can very well happen that no statement with a predicate like arrive at 2 o'clock will come out as true (because it may so happen that no arrivals coincide with the point denoted by 2 o'clock to the exact nanosecond and even beyond nanoseconds), which is highly unintuitive. In contrast, such sentences can be true on a granularity functions-based framework, which is a semantic, rather than pragmatic account of vagueness.

Also, the pragmatic halo account needs to make some stipulations the present granularity function-based theory does not. For instance, Lasersohn (1999) has to stipulate that the *exact denotation* of an expression be the unique element which is *more similar* to itself than any others in its pragmatic halo. No such or similar stipulation is needed in our case because what are ordered (via the finer-than relation) are granularity functions on the basis of unit size, which is independently ascertainable, in contrast to an ordering of individuals based on *similarity*. We get the classical denotation when the granularity is set to "absolutely fine" (cf. Section 4.1.2.4 below), and the denotation at any other granularity assigned to the relevant expression is necessarily a superset thereof.

van Rooij (2011b) also proposes in informal terms that a specific minimal difference size is at the heart of granularity, saying that, for instance, x is taller than y in a partic-

ular context iff the height of x is higher than the height of y plus some fixed small real number ϵ . However, he, like Lasersohn (1999), does not incorporate this notion and difference measures into his formal account, and uses a supervaluation-like approach to granularity.

An account of granularity which does explicitly encode measures and unit size is that of Champollion (2010), who introduces a function ε which outputs 'true' for things that count as small given some measure. For instance, $\varepsilon(\lambda t[hours(t) = 1])(t')$ is true just in case t' is very small in comparison to one hour. This function is then part of the semantic representation of a sentence (involving, for instance, a *for*-adverbial such as *John waltzed for an hour*), and not a parameter of interpretation as the granularity functions of S&S that I employ. As such, it cannot be used to model the effects that different components of a sentence can have on the granularity parameter used in their interpretation, although such effects do appear to exist.

In particular, the granularity parameter used in the interpretation of different components of a sentence is not generally allowed to vary greatly. For instance, as Tatevosov (2008, p. 418) argued, in Russian (and it seems that in English, as well), modification of "long-term activities" like *rukovodil* 'supervise' is more felicitous with temporal adverbials which, in our terms, are associated with granularity parameters characterised by a unit size comparable to year-long (rather than minute-long) intervals, such as *v* 2004 godu 'in 2004'. Also, as Kearns (1991, p. 61) remarked, a mixing of granularity "produces absurdity, as in *The moment the market slumped John arrived for the meeting*". On the granularity function-based account building on S&S, such a restriction of *commensurable granularity* can be easily captured by a principle requiring the parameters of interpretation (and the granularity parameter among them) to be unchanged in the interpretation of a sentence — much as the variable assignment function is not generally allowed to change.

Also, while Champollion's account needs a specific measure such as the one hourlong interval to determine the unit of granularity, on our account building on S&S, the interpretation function always has a granularity parameter, whether or not there is a specific measure expression (such as *for an hour* in *John waltzed for an hour*) fixing it (cf. also the discussion in Section 4.3.3.3). As such, a "granularity functions as parameters of interpretation" approach is more general, allowing granularity to have relevance even in absence of explicit measures. It can offer an account of gradable adjectives like *full* and approximators like *exactly*, as well as an account of when a sentence like *You're exactly right* can be truthfully asserted (for details, see S&S), while it is not trivial how Champollion's (2010) account based on ε could be extended to these phenomena.

Tatevosov (2008) also models minimally accessible interval sizes, but he focuses exclusively on event duration. For him, non-durative events are those that are too short to match the minimal duration of an *observation event*. As such, his account of the durative/non-durative distinction is characterised by the same limited generalizability as Champollion's (2010) account in comparison with S&S's granularity functions approach. Because Tatevosov (2008) refers to the temporal length of observation events, it is specific to the temporal domain, while the notion of granularity functions and a MDI is domain-independent, and can be used to explain the extended/non-extended distinction in other dimensions, such as space or abstract property spaces (say, when determining whether a dress counts as redder than another one in some context).

In addition, a drawback of relying on the length of observation events as a limiting condition for durativity is that in many cases, it is arguably not simply the length of observation events that determines whether the agent regards an event as extended in time, but the goals of the agent (cf. the discussion on page 80 about the three glasses of water): the agent may be aware of the duration of, say, a cough (which presupposes that the observation events are smaller in length than the cough), but for the purposes of the discourse context, the objective durativity of that coughing event (and coughing events in general) is *irrelevant*. This is captured on the granularity functions-based account by having the granularity function be a parameter of interpretation, which is affected by a number of factors: the cognitive ability of the agent, the discourse context, as well as the linguistic expressions that are interpreted.

Outside of formal semantics, the MDI of an everyday granularity level on the temporal scale can be identified with the 2-3 second interval size (aptly called by Engelberg 2000 the *cognitive moment*), which several psychological studies (Pöppel 1978, Turner and Pöppel 1983, Feldhütter et al. 1990) have shown to be relevant in different areas of human perception and behaviour. Among others, this is the threshold above which events are perceived as durative, rather than instantaneous:

Beyond the two horizons of this [cognitive] moment exist the two periods which together constitute duration ... (Turner and Pöppel 1983, p. 297)

4.1.2.3 Overlapping intervals

Given the observation that for the purposes of granularity, what is relevant is unit size, I propose to adapt the granularity functions of S&S in such a way that the granular-

ity parameter of interpretation is defined by the MDI, rather than specific granularity functions. That is, the parameter of interpretation is not a granularity function, but a *set* of granularity functions. So let us define *granularity levels* as equivalence classes of granularity functions with the same unit size:

Definition 9 (Granularity level)

A granularity level is a set of granularity functions over the same scale which have the same MDI. Unless the context specifies otherwise, the granularity level that is the parameter of interpretation includes all granularity functions with the same MDI. The MDI of the granularity functions in a granularity level is called the unit of that granularity level.

Figure 4.4 illustrates the move from specific granularity functions to granularity levels. This move can be seen as an adaptation of the idea of Rothstein (2004), who



Figure 4.4: A granularity level as a set of granularity functions with the same unit size.

argues that although all activities have atomic parts, the atomic parts of regular activities (but not those of semelfactives) *overlap*. Similarly, although a single granularity function results in a set of disjoint blocks on a scale, there can be a great (perhaps even infinite) number of equally acceptable granularity functions at a particular granularity level, whose mappings are different, and their blocks *overlap* with those of other granularity functions. Figure 4.5 illustrates how the intervals in the range of a set of granularity functions belonging to the same granularity level (designated with $\gamma_1, \gamma_2, \gamma_3$) overlap just as the minimal parts of an activity like *run* (designated with P_{min}) overlap according to Rothstein (2004).



Figure 4.5: Analogy: Overlapping granularity functions and atomic parts of activities

Thus, γ_1 would map, say, the time point 17:14 to the interval 16:45–17:15, while γ_2 would map it into the interval 17:00–17.30. Both satisfy the criteria for granularity functions, and both have the same unit size (half an hour).

While the atomic function of Rothstein (2004) in itself outputs a set of overlapping intervals (eventualities), we need a *set* of granularity functions to achieve this end. Thus, if granularity *levels*, rather than granularity functions, are parameters of interpretation, we will get multiple, but, in a given context, equally acceptable, *interpretations* — similar to a supervaluationist account. And similarly to the idea of a sentence being *supertrue* on a supervaluationist account just in case it is true according to *all* classical valuations, we can now say that two points on a scale are distinguished at a particular granularity level just in case they are distinguished by *all* the granularity functions of that level.

In Figure 4.6, for instance, t_1 and t_2 are not distinguished at the relevant granularity level, because their distance is smaller than the unit of that level, and so there is a granularity function at that level (γ_2) that maps them to the same interval. In contrast, t_1 and t_3 are distinguished at the same granularity level, because their distance is greater than the unit of that level, and so no granularity function at that level maps them to the same interval.



Figure 4.6: Two points are distinguished at a granularity level just in case they are distinguished by all granularity functions at that level.

We can thus elude the problem of the higher-order Sorites paradox that S&S's original account faces.¹¹ If the distance between two points, t_1 and t_2 on a scale is less than the unit of the relevant granularity level Γ , then there will be at least one granularity function $\gamma \in \Gamma$ such that $\gamma(t_1) = \gamma(t_2)$, that is, one which cannot distinguish between t_2 and t_1 . However, if the distance of t_1 and t_2 is greater than one unit, then no such

¹¹Note that without the combination of granularity functions with the idea of multiple interpretations, the use of granularity functions would also result in rendering a dense scale, such as time, into a discrete one, which may have unwelcome consequences in our theory. But now we can correctly retain the "denseness" of the scale (even if merely a higher-order denseness, emerging across different possible interpretations), albeit at the price of a supervaluation-like move.

granularity function can be found at granularity level Γ : the two points are always distinguished at that level of granularity.

Importantly, though, the combination of multiple valuations and granularity functions does not solve the Sorites paradox in general. We still have sharp boundaries between the set of *intervals* that are extended at a particular granularity level, and those that are not (the boundary being the MDI). Thus, take S&S's example with 5 *metres* above. On their example, the granularity function γ_{mid} maps the degree 5m to the interval [4.75m, 5.25m]. Moving to granularity levels, there are several granularity functions with the same unit, all belonging to granularity level $\Gamma_{mid} = {\gamma_{mid}, \gamma'_{mid}, \gamma''_{mid}, \dots}$. The following granularity functions in Γ_{mid} , for instance, assign different images to the degree 5m:

 $\gamma_{\text{mid}}(5\text{m}) = [4.75\text{m}, 5.25\text{m}]$ $\gamma'_{\text{mid}}(5\text{m}) = [4.74\text{m}, 5.24\text{m}]$ $\gamma''_{\text{mid}}(5\text{m}) = [4.76\text{m}, 5.26\text{m}]$

A welcome aspect of moving to granularity levels, as mentioned above, is that 4.74m and 4.75m are indistinguishable at Γ_{mid} just as much 4.75m and 4.76m are, because two points are only distinguishable at a granularity level if all granularity functions at that level distinguish between them. On the other hand, while 4.75m and 5.25m are indistinguishable at Γ_{mid} , 4.74m and 5.25m are *still* distinguishable despite the negligible difference between the two pairs. This is because there is a specific unit size associated with a granularity level, and so any two points whose distance is greater are distinguished at that level, while any two points whose distance is smaller are not: the cut-off point is sharp.

However, we did relegate this unwanted sharp cut-off point to a higher level, doing away with some unwanted results (of say, two very close points, such as 4.74m and 4.75m, being distinguishable). So this is a general gain over the granularity *function*-based account of vagueness: although we cannot tackle the problem of the Sorites paradox, we do eliminate more unwanted distinctions.

4.1.2.4 Precisifications

Let us now elaborate on the notion of precisifications. Landman (1991, p. 139) argues that precisification, or in our terms, moving to a finer granularity, can happen indefinitely, and also that we cannot reach the minimal intervals (or points) of time through these. While I agree with him on the first point and at least leave open the possibility of

an infinite number of granularity levels over a scale, I suggest that it is and should be possible to access the actual points on a scale with them. In particular, we can model objective reality, as a limit point for all epistemic stances, with an *absolutely fine granularity*, which maps each point on a scale to itself. (This is a valid move given that points are intervals, even if degenerate ones.)

Definition 10 (Absolutely fine granularity)

In the set of granularity functions \mathbb{G} , there is an absolutely fine granularity for each scale. An absolutely fine granularity ρ_{S} ('reality') defined over a scale S is such that:

(4.8) $\forall s \in \operatorname{carrier}(\mathcal{S}) \colon \rho_{\mathcal{S}}(s) = \{s\}$

Note that there is only a single absolutely fine granularity function for each scale, while there are necessarily multiple granularity functions for almost all greater unit sizes. The absolutely fine granularity level is one of the two granularity levels that contains a *single* granularity function, the other being the level containing the *absolutely coarse* granularity function, which maps each point on a scale to the complete scale, i.e., it is the granularity function (and level) which is unable to make any distinctions on a scale.¹² The absolutely coarse granularity has no linguistic utility, though; it only has theoretical significance. When it is defined, it is the greatest element among the granularity functions defined over a scale with respect to the *finer than* relation, while the least element is the absolutely fine granularity.

Bearing in mind the slight superiority of granularity levels over granularity functions in attacking (though not solving) the Sorites paradox, in the discussion that follows I mostly only refer to granularity functions for the sake of simplicity unless the discussion requires explicit reference to granularity levels.

¹²Formally, though, an absolutely coarse granularity function is only defined for *bounded* scales given the characterisation of granularity functions in Section 4.1.1.

4.2 Granularity-dependent durativity and slow-motion readings

4.2.1 Granularity-induced changes in durativity

Given the right context (a context that involves what I call switching to a "fine granularity"), some non-durative predicates can be construed as describing durative events by virtue of the fact that the events in their extension are in fact not instantaneous in the physical reality. Examples include the following slow-motion readings of the progressive:

- (4.9) Look at the screen, the Challenger is exploding now. (Verkuyl 1989)
- (4.10) And now the subject is coughing. (Comrie 1976, p. 43, as uttered by a lecturer in an anatomy lecture commenting on a slowed down film about a single cough)

In (4.9), the event described by *explode*, which would be regarded as instantaneous at an everyday granularity, is presented as taking non-zero time at the finer granularity enforced by a slow-motion recording on television. For some predicates, a radical switch in granularity is needed for this effect, such as *cough*, which describe events that objectively take tenths of a second. In their case, more extreme settings, such as a scientific laboratory scenario is needed, as in (4.10).

Note that a switch to a finer granularity does not necessarily entail "zooming in" in the sense of, say, looking at something through a magnifier or a microscope. It merely means that differences which do not matter or are not recognised at a coarser granularity are foregrounded. Take the case of the bullet crossing a panel. A slow-motion recording thereof will not, at the same time, involve an objective magnification in space (as under a microscope), but by virtue of the slower progression of time, it becomes possible and relevant to discern spatial positions of the bullet with respect to the panel (such as the bullet entering the panel, being in the middle of it and then leaving it) inaccessible or irrelevant at a normal playback rate.

The idea that "something can be a durationless moment of time in the present context, but become an interval that has duration after all, when we shift the context to a more finegrained one" (Landman 1991, p. 138) has been recognised by several authors, but as Landman himself then pointed out, the semantic details of such context dependent granularity had never been worked out (though see Section 4.1.2.2 for a discussion of some previous proposals for this or related issues), and Landman did not do so, either. The present section aims to offer a model-theoretic analysis of granularity-based durativity changes in the framework detailed in Section 4.1 building on the theory of granularity functions of Sauerland and Stateva (2007, 2011).

4.2.2 Granularity-relative temporal trace and durativity

As a crucial step in modelling durativity changes, I *relativise* the temporal trace function to granularities. We will still preserve the ability to express the classical temporal trace of an event, because we can exploit the notion of an absolutely fine granularity to this end.

Let $G_{\mathcal{T}}$ be the set of granularity functions defined over the time scale $\mathcal{T} = \langle T, <_{\mathcal{T}} \rangle$. Then the definition of a granularity-relative temporal trace function is as follows.

Definition 11 (Temporal trace function—granularity-relative)

The temporal trace function $\tau: G_{\mathcal{T}} \times E \to \operatorname{Int}(T)$ is a function such that for all $e \in E$, if t_1 is the starting point of e in T and t_2 its final point, then the temporal trace $\tau(\gamma, e)$ of e at granularity γ over \mathcal{T} is defined as the interval given by the points to which the degreeifying function maps the start and endpoint of e at granularity γ :

(4.11)
$$\tau(\gamma, e) = [\deg(\gamma, t_1), \deg(\gamma, t_2)]$$

The classical notion of the temporal trace function is now captured by the notion of the absolute temporal trace function τ_{ρ} , which is the temporal trace function relative to the absolutely fine granularity function ρ over T:

(4.12)
$$\tau_{\rho}(e) \stackrel{\text{def}}{=} \tau(\rho, e)$$

That is, the temporal trace of an event at a specific granularity is determined by the location of its start and endpoints at that particular granularity (and the absolutely fine granularity will, of course, return the actual start and endpoints themselves). Note that this granularity-relative temporal trace is a representation of the conceptualisation of the runtime of an eventuality, while the objective link between eventualities and intervals is encoded in the absolute temporal trace function.

We are now in the position to account for how granularity changes can lead to a two-faced behaviour of event predicates with respect to durativity building on the modelling of conceptualisations of the same event under different granularities. Let $|\cdot|: \mathbb{B} \to \mathbb{R}$ from the set of all bounded intervals \mathbb{B} to the set of real numbers \mathbb{R} measure the length of an interval, such that for any interval $i \in \mathbb{B}$, the value of |i| is given by the distance $d_S(a, b)$ between the infimum a and supremum b of i, where d_S is the metric defined over the scale S that i is part of (cf. Section 4.1.1). Then, we can have two granularities such that:

$$(4.13a) |\tau(\gamma_{\text{fine}}, e)| > 0$$

$$(4.13b) \qquad \qquad |\tau(\gamma_{\text{coarse}}, e)| = 0$$

if the following holds for the starting point t_1 and endpoint t_2 of event e:

(4.14a)
$$\gamma_{\text{fine}}(t_1) \neq \gamma_{\text{fine}}(t_2)$$

(4.14b)
$$\gamma_{\text{coarse}}(t_1) = \gamma_{\text{coarse}}(t_2)$$

That is, the fine granularity distinguishes between the start and endpoints of the event (i.e., maps them to different intervals), while the coarse granularity does not. Graphically, this is illustrated in Figure 4.7. Generalizing the approach to granularity *levels*,



Figure 4.7: Duration switch through a granularity switch

we can say that an event *e* has zero duration at granularity level Γ iff there is a granularity function $\gamma \in \Gamma$, for which $|\tau(\gamma, e)| = 0$, that is, at that granularity level, some granularity function assigns zero duration to event *e*. Informally, an event is durative at some granularity level if and only if it is always possible to distinguish the start and endpoints of the event at that granularity level.

Then predicates describing event *e* behave as an accomplishment at the fine granularity γ_{fine} , and as an achievement at the coarse granularity γ_{coarse} —the two-faced behaviour we have been aiming at. In what follows, I will present how this abstract idea can be put to use in the case of concrete examples. **Instantaneous and durative readings.** Kearns (1991, p. 59ff) argued that a sentence like *Just as P, Q* requires *P* to be punctual. Let us inspect this claim in some detail. If *P* is an activity, then indeed the sentence appears infelicitous:

(4.15) #Just as Mary ran/was running on the treadmill, the meeting ended.

One may wonder if it is not the atelicity of *running on the treadmill* that is responsible for infelicitousness, but an accomplishment predicate results in an equally infelicitous sentence:

(4.16) #Just as Mary read the paper, the meeting ended. (Kearns 1991)

In contrast, an achievement predicate results in a felicitous sentence:

(4.17) Just as Mary read the note, the meeting ended. (Kearns 1991)

This pattern is specific to the *Just as...* construction (as well as the *As soon as...* and *The moment that...* constructions according to Kearns 1991): as Daniel Altshuler (p.c.) noted, substituting *just as* for *while*, we arrive at a construction that requires a durative predicate, cf.:

- (4.18) a. While Mary ran/was running on the treadmill, the meeting ended.
 - b. While Mary read the paper, the meeting ended.

On this construction, read the note also receives a durative interpretation, as in (4.19).

(4.19) While Mary read the note, the meeting ended.

Thus, the same predicate (*read the note*) can behave both like a durative and an instantaneous predicate, depending on the context — the two-faced behaviour cited extensively by Verkuyl (1972, 1989, 1993). But not only does a predicate like *read the note* have a potential for two-faced behaviour in different contexts, but one and the same event can be described by the same predicate under both a durative and an instantaneous interpretation. Consider the following sentence:

(4.20) Mary read the note in 5 seconds.

(4.17) and (4.20) can be made true by the same event of Mary reading the note, even though, as Kearns (1991) argues, the first clause of (4.17) has a non-durative interpretation, while the same predicate in (4.20) is used with a durative interpretation. But while the first one is evaluated at a coarse granularity at which a 5-second event is regarded as instantaneous, the granularity parameter of the second one is shifted to a finer one (due to the fine granularity introduced by the expression *5 seconds*, cf. the discussion on granularity commensurability in Section 4.1.2.2). So the clause with *Mary read the note* will have the interpretation in (4.21a) and (4.21b) for (4.17) and (4.20), respectively (irrelevant parameters of interpretation are suppressed). ∂ is the presupposition operator from Beaver (1995). I assume *just as* is responsible for the non-durativity presupposition — and so sentences such as (4.15) and (4.16) which violate this presupposition are odd.

(4.21a) [[Mary read the note [at t]]]^{$$\gamma_{coarse}$$} = $\exists e[read(e) \land theme(e, note) \land agent(e, mary) \land$
 $\tau(\gamma_{coarse}, e) = t \land (t <_{\mathcal{T}} now) \land \partial(|\tau(\gamma_{coarse}, e)| = 0)]$

(4.21b) [[Mary read the note in 5s]]^{$\gamma_{\text{fine}} = \exists e [\text{read}(e) \land \text{theme}(e, \text{note}) \land \text{agent}(e, \text{mary}) \land$ $\tau(\gamma_{\text{fine}}, e) = t \land (t <_{\mathcal{T}} \text{now}) \land \text{seconds}(\tau(\gamma_{\text{fine}}, e)) = 5]$}

As can be seen, the requirements on the described event are not contradictory in the case of the two sentences (since the same event might count as durative on a fine granularity, but instantaneous on a coarse one), and so the same event description can be used as a non-extended reference point in time for the clause *the meeting ended*, and modified with a duration-measuring *in*-adverbial.

The analysis above follows the idea of Kamp (1979/2013), who proposed that an instantaneous event in our model is not necessarily instantaneous in the "real world" (p. 91). But Kamp (1979/2013) achieved this by constructing a different time model for each sentence, each of which are individually connected to the real-world model of time, while there is no relation directly between the different models themselves. In contrast, on the present account, we do not need independent, arbitrarily constructed models of time, but instead, a tool with independent application potential, namely, granularity functions, is employed which produces the different models of time used in different contexts.

4.2.3 Constraining the granularity parameter and explaining slowmotion readings

Although as the foregoing discussion showed, we can successfully represent granularity-induced changes in durativity with reference to a granularity parameter of interpretation, the question arises if and how we can constrain this granularity parameter in a given case. In particular, can we predict that a coarse granularity is used in the interpretation of (4.17), but a fine one in the case of (4.20), or in the case of a slowmotion progressive like *Look at the screen, the Challenger is exploding*? The answer, I propose, is that we can, insofar as a distinction between a coarse and a fine granularity is concerned.

Take (4.20), *Mary read the note in 5 seconds*. This sentence is ambiguous between a "duration" and an "after" interpretation: what took 5 seconds is either the note-reading event itself (a reading that is highlighted given a continuation like *She is a fast reader*), or the time after which the note-reading event took place at what is conceptualized as an instant (a reading that is highlighted in a context in which a one-word note is pinned to the wall of a room, and researchers study how long it takes for different subjects to notice and read the note after entering the room). A "duration" reading arises if Mary's note-reading event in a specific scenario took 5 seconds. In this case, a fine granularity parameter is needed at which the event is extended and its temporal trace measures 5 seconds; a coarse granularity at which the event is instantaneous cannot be used to get a duration reading. In contrast, for an "after" interpretation, according to which an instantaneous note reading takes place *after* the 5-second interval, a coarse granularity is required at which the note reading event is instantaneous.

Of course, the question *exactly* how coarse or fine a granularity parameter is used in the case of the two interpretations is difficult, if not impossible, to answer. I believe that some form of *principle of economy* limits the choice of granularity. If any granularity level as coarse as or coarser than some specific granularity level, say, Γ_i is suitable to get the "after" interpretation in the case of (4.20), and no difference in the interpretation results from choosing one over the other, then the most economical decision is to use Γ_i as the parameter of interpretation. Conversely, if any granularity as fine as or finer than Γ_i is sufficient for the "duration" interpretation, then again Γ_i should be used.

Compare *read the note* with an achievement like *notice* which describes events that take much less time than note-reading events (and can be measured in hundredths of a second). (4.22) — in contrast to (4.20) — can only have an "after" interpretation,

because no matter how we choose our granularity parameter, no note-noticing event will be 5 seconds long, and so the only coherent interpretation is the "after" one.

(4.22) Mary noticed the note in 5 seconds.

Now take (4.17), Just as Mary read the note, the meeting ended. In this case, the *just as* construction presupposes that the note-reading event was instantaneous. Thus, if we take a *fine* granularity at which the note reading event is extended, we arrive at a presupposition failure. Therefore, the only option is to take a coarse granularity at which the endpoints of the note reading event are not distinguished and the event is thus instantaneous. This granularity would need to have a unit size of at least 5 seconds, which is a granularity level that is natural to use in everyday contexts. However, if we take the infelicitous example *#Just as Mary read the paper, the meeting ended* from Kearns (1991), and apply the same reasoning, then in order to avoid presupposition failure, the granularity parameter has to be coarse enough to have a unit size of at least an hour or so, as reading a paper takes minimally that much time. However, this is a granularity parameter that is at least an order of a magnitude coarser than those used in a natural everyday context, and so the presupposition in this case cannot be accommodated.

As for (4.23), we can again refer to a presupposition failure in constraining the granularity parameter.

(4.23) Look at the screen: the Challenger is exploding.

Given a semantics of the progressive as an operator involving partitivity over events (cf. Section 2.1.1), a reasonable applicability condition of the progressive is an input predicate denoting a set of *durative events*. Now, if *the Challenger explode*^{\emptyset} describes events that take less than, say, half a minute, then using any granularity level with a unit size that is greater than half a minute will result in an interpretation at which the progressive applies to a predicate of instantaneous events, violating the applicability condition. However, if a fine granularity with a smaller unit size is used, then the predicate *the Challenger explode*^{\emptyset} has events in its extension that are durative, and so the applicability condition of the progressive is satisfied. Because such a fine granularity is necessary to this end, a slow-motion interpretation results.

Given the level of the ease of accommodation of the required granularity parameter, we can also explain why some achievements are much less acceptable in the progressive: for instance, (4.24),

(4.24) #Peter is noticing Mary,

requires an extremely fine granularity level to avoid presupposition failure, one at which differences of a hundredth of a second in size are relevant, but such a fine granularity is much more difficult to accommodate in normal, everyday contexts than that required by *explode*.

In contrast to both (4.23) and (4.24), even an everyday granularity is sufficient in the case of progressive accomplishments, as in (4.25), to ensure that no presupposition failure occurs: the predicate *read a paper* denotes events that are extended at the levels of granularity used in everyday conversation.

(4.25) Mary is reading a paper.

Accomplishments and achievements. On the present framework, all predicates that apply to events which are in actual reality extended can function as either achievements or accomplishments, modulo the granularity parameter of interpretation. Of course, as we have seen in the foregoing, for some predicates, a durative interpretation requires an extremely fine granularity that is difficult to accommodate in most conversational contexts.

Following up on this idea, I propose that the way we can capture the difference between accomplishments and achievements in our framework is by appealing to the *use* of these predicates by speakers. In particular, for each predicate, we can identify a set of what we can call "normal" granularity levels such that the unit size of these granularity levels belongs to a certain range, and in most cases¹³ that the predicate is used, it is interpreted with respect to some member of this set of normal granularity levels.

¹³This characterisation is deliberately vague. We could set an arbitrary threshold for what counts as "most common" usage, for instance, 95% — as is common in statistical analysis — but I believe this is a matter for experimental research.

The normal levels of granularity are of course not randomly used by language users, but must intuitively be connected to the temporal magnitude of the events denoted by a predicate (which I will discuss in somewhat more detail on 112).

Then we can say that accomplishments are those predicates that in most cases are interpreted with respect to a granularity parameter at which they denote extended events, while achievements are those which denote instantaneous events at their normal granularity parameters:

- Accomplishments are predicates of change which describe, by vast majority, durative events at the levels of granularity most commonly used in the interpretation of the predicate.
- Achievements are predicates of change which describe, by vast majority, punctual events at the levels of granularity most commonly used in the interpretation of the predicate.

This is how the view promoted by authors like Filip (1993/99) and Engelberg (2000) can be captured, who say that although it is possible to have slow-motion scenarios in which a regularly punctual predicate functions as extended, these scenarios are not common and should not be taken to invalidate the basic, typical use of these predicates, which is punctual. Crucially, the present characterisation of the difference between achievements and accomplishments entails that this difference is not inherent in their *extension*, but is a property of the *linguistic predicates* and their use by speakers. At the same time, we do not preclude scenarios in which the granularity parameter is one which is seldom used in conversation, and in such cases, the durativity feature of an event described by a predicate, and by extension, the durativity feature of the predicate itself, can change, allowing us to cater for such scenarios in a model-theoretic semantics without sacrificing the divide between achievements and accomplishments.

Summary: a granularity parameter is sufficient for slow-motion readings. The goal of the present section was to show how slow-motion readings can be modelled. The crucial components necessary for this are the introduction of granularity functions as a parameter of interpretation and the *relativisation* of the temporal trace function to granularities. In this, I have followed the advice of Janssen (1997) who proposed incorporating as a new parameter of semantics anything that appears to influence meaning and would thereby rule out a compositional semantics. In the framework presented here, we can thus answer the qualms of Kearns (1991), who, like Verkuyl (1989, 1993), argues that there are no truly durationless events (but unlike Verkuyl, would wish to retain durativity as a linguistically relevant feature), and so (linguistic) durativity is "partly a matter of »grain size«", which, she says, is a "problem [...] for truth-conditional semantics" (*op. cit.*, p. 61).
Note that the analysis of slow-motion progressive readings does not require any form of coercion in the sense that no predicate is coerced into a different predicate. The predicate *the Challenger explode* $_{\emptyset}$ denotes the *same* set of events in the scope of the progressive (as in *The Challenger is exploding*) and outside the scope of such an operator (as in e.g., *The Challenger exploded at 16:03*). What changes is a parameter of interpretation: just as speakers aim to determine the most reasonable variable assignment function to use in the interpretation of a sentence, the granularity parameter is likewise narrowed down to one which results in a reasonable interpretation. Thus, slow-motion progressives are to be analysed through *underspecification* — namely, the underspecification of a parameter of interpretation can be derived from the fact that choosing a coarse granularity results in a presupposition failure (failure to satisfy the applicability condition of the progressive).

Crucially, the proposed account of slow-motion readings is compatible with any event-mereological analysis of the progressive, such as those of Landman (1992), Portner (1998), Bonomi (1999) or Varasdi (2010, 2014), because the progressive has in its scope a predicate of *durative* events in felicitous cases. As the discussion in Section 2.2 indicated, however, event-mereological accounts run into problems in the case of preliminary process readings like *Mary is reaching the top*, so in the next chapter, I will argue for a semantics of the progressive which is based on an additional formal tool, the *scales of change*. On that account, I will argue that it is the scale of change, rather than events, that are required by the progressive to be extended, that is, have non-degenerate proper parts.

So in the case of slow-motion readings of progressives like *The Challenger is exploding*, it is a scale of change, rather than an event, that is required to be extended and have proper parts. But importantly, just as the temporal trace is relativised to a granularity over time, so the scale of change will also be relativised to a granularity parameter over time. And given a change to a finer granularity over time at which an event is temporally extended, the scale of change associated with an event at that granularity will also be necessarily extended. Thus, given a temporal trace-based account of slow-motion readings of progressive achievements, such readings can also be modelled in the theory of the progressive referring to scales of change. Before presenting such a scale-based account of the progressive in Chapter 5, I discuss some implications and further uses of a granularity-relative notion of temporal extendedness.

4.3 Implications of the granularity-based account of durativity

4.3.1 Durativity and vagueness

On the present approach, what we see is that there is a connection between durativity (i.e., how we conceptualise objective temporal extent), granularity, and vagueness. In particular, the temporal trace function, and hence durativity, is relativised to a granularity parameter, which is originally used in the account of *scalar vagueness* by S&S. The fact that durativity is amenable to an analysis through granularity functions indicates that it displays this kind of vagueness. Importantly, however, "being durative" is not vague in the sense a vague predicate like *tall* is, which S&S call *epistemically vague*.¹⁴

The difference between the two types of vagueness, according to S&S, boils down to having or not having a precise interpretation. An *epistemically vague* predicate like *heap* does not have a precise interpretation (and is thus "necessarily" vague), while a *scalar vague* expression like *6 o'clock* does, in this case, a point in time. They argue that these two kinds of vagueness should be differentiated, and accounted for using different tools. A central argument of S&S for such a division is a similar division among *approximators* (which correspond more or less to what Lasersohn 1999 calls *slack regulators*), some of which can modify epistemically vague predicates (cf. *definitely a heap*), while others can only modify scalar vague expressions (cf. *exactly 6 o'clock*, but *#exactly a heap*). What I would here like to point out is that because scalar approximators only apply to predicates whose denotations refer to a point on a scale (cf. S&S's classification of them into "end-point" and "mid-point" oriented approximators), they cannot be used to test the scalar vagueness of predicates like "be durative" whose denotations are not single degrees on a scale.

But I propose that there is a further, conceptually trivial way to linguistically test (and thus support) the distinction between scalar and epistemic vagueness, namely, using an expression which presupposes the existence of a precise interpretation. Thus,

¹⁴I am not sure this is a fortunate choice of terminology for these expressions, since although some authors (such as Williamson 1994) do presuppose that there is a sharp boundary for predicates like *tall*, which is simply inaccessible epistemically to speakers, this is a contentious view, some authors being of the opinion that vagueness is inherent in the objective reality (cf. e.g., Tye 1990).

the expression *strictly speaking* is compatible¹⁵ with scalar vague predicates only (cf. (4.26a)), but not epistemic vague ones (cf. (4.26b)).¹⁶

- (4.26) a. Strictly speaking, Anna didn't arrive at **6 o'clock** sharp, but a few seconds later, but come on, don't be nitpicking!
 - b. ?Strictly speaking, this isn't a **heap**, it's not big enough to be one, but come on, it nearly is, don't be nitpicking!

Assuming that natural language expressions like *instantaneous* refer to the abstract concept of durativity analysed in Section 4.2, then as (4.27) shows, durativity ("being durative") belongs to the class of scalar vague predicates: it has a precise, strict sense, but under different granularities, can have a greater "semantic halo".

(4.27) Strictly speaking, the explosion was not **instantaneous**, but it was so rapid, it felt like it happened in a moment.

So, although "being durative" is not vague in the sense of *epistemic vagueness*, i.e., not vague like *heap* or *tall*, it is vague in the sense of scalar vagueness. So we should be able to construct a chain of *Sorites paradox* reasoning, which S&S hold to be the defining characteristic of *vagueness* in general. And I propose that we can.

Say, an explosion took a hundredth of a second. It would qualify as instantaneous in an everyday scenario (i.e., it would be possible to use an *at*-adverbial in the description of the explosion, say, *The house exploded at 12:00 sharp*, while an *in*-adverbial as

(i) A: Strictly speaking, this is not a red cardigan, it's puce. B: Come on, don't nitpick!

I believe this is a very interesting observation which shows that colour terms are *both* epistemic vague and scalar vague predicates. As (i) shows, it is possible to truthfully assert *the cardigan is red* in one context (the everyday context that speaker B assumes) and truthfully deny that the same cardigan is red in a more fine-grained context (the context that speaker A assumes). This is precisely the essence of scalar vagueness.

¹⁵It is important to restrict the interpretation of *strictly speaking* to a *more precise approximator* reading in S&S's terms. That is, the reading relevant is the one which concerns the *dimension* of the scale associated with the predicate it modifies. (Similar considerations apply to *exactly*, which is a scalar approximator according to S&S, cf. *#Red wine is exactly healthy*, but they note that under negation, it patterns like an epistemic approximator, cf. *Red wine isn't exactly healthy*.) Hence the addition of "it's not big enough to be one" in (4.26b): it indicates that what is at issue is not, for instance, the shape of a pile of sand, but the *amount* of sand particles.

¹⁶Peter Sutton (p.c.) noted that colour terms, which are epistemic vague predicates (there is no clear borderline between, say, orange and red) appear not to fit this observation, as they can felicitously appear with *strictly speaking* on the relevant dimension-oriented reading in the following discourse:

in *The house exploded in a minute* would not measure the duration of the event). If another explosion also took place which took just a millisecond longer, it would then still qualify as instantaneous (because a millisecond difference in length constitutes an *indistinguishability relation* between runtimes on an everyday scenario). But then taking a chain of events which are successively 1 ms longer than their predecessor, we can arrive at an explosion which took a minute, which would not qualify as instantaneous in a context in which 1-minute intervals are seen as extended: in this context, the *in*-adverbial in the description *The house exploded in a minute* would measure out the *duration* of this minute-long event, and an *at*-adverbial could at most refer to the beginning of the explosion (on the *punctual temporal focus* reading of Taylor 1988). But, just as in the case of a Sorites paradox in general, it is impossible to determine the cut-off point between events which are durative and those which are not.¹⁷

4.3.2 Endpoints and vagueness

An additional advantage of introducing granularity-relative temporal traces is that it now becomes possible to have the temporal trace of certain events be defined only at certain granularities. Then we can account for the vagueness of the location of the endpoints associated with many (perhaps all) eventualities. For instance, if John wrote a letter to Mary, then when exactly did the letter-writing event start (and, similarly, end)? When John started writing the very first character on the paper? Or at the moment he picked up his pen perhaps? But when did *that* event start? At the moment his fingers touched the pen? Or when his brain fired the first neuron which led to the grabbing movement? Such lines of argumentation could go on potentially indefinitely and there appears to be no objective criterion of delineation.

Similarly, Engelberg (2000) noted in passing that in sentences like the following "we cannot determine the temporal boundaries of the events referred to very precisely".

- (4.28) a. This year Jamaal broke his close ties with the Detroit Pistons.
 - b. Rebecca managed to break the deadlock.

Thus, the temporal boundaries of an event are vague just as the boundaries between tall and non-tall people, or between heaps and collections of grains not big enough to

¹⁷Note that, as mentioned above, this is a higher-order vagueness that the present account of granularity cannot capture, similar to other classical accounts of vagueness: since the MDI is precise, this will define the boundary of durative and non-durative events.

be heaps. This is problematic in a classical event-semantic framework such as Krifka (1992) on which predicates denote sets of eventualities and the temporal trace function maps an eventuality to a time interval. Time intervals are convex subsets of the set of time points, and in the case of bounded time intervals, their supremum and infimum is of course a single time point. This means that the boundaries (the supremum and the infimum) of the temporal traces of bounded events are sharp down to time points. This problem has not, to my knowledge, received any real attention in previous works, and the (classical) temporal trace function is simply taken for granted as specifying the runtime of any event.

A granularity-relative temporal trace provides a formal solution to the problem of vague event boundaries. The only assumption we need is that there may be a minimal granularity function or level in terms of the finer-than ordering over granularity functions and levels for each event such that the temporal trace of the event is not defined under that particular granularity.

Definition 12 (Maximally precise temporal trace)

For each event e, there is a granularity function $\gamma \in G_{\mathcal{T}}$ over time such that for all granularity functions $\gamma' \in G_{\mathcal{T}}$ over time, $\tau(\gamma', e)$ is defined iff $\gamma \preceq \gamma'$ (that is, γ is at least as fine as γ'), with γ possibly but not necessarily being the absolutely fine granularity ρ over time. $\tau(\gamma, e)$ is called the maximally precise temporal trace of e.

If γ is not the absolutely fine granularity ρ , then $\tau(\gamma', e)$ is not defined for granularity functions finer than γ . Suppose that $\tau(\gamma, e) = [t_1, t_2]$. Then for any strictly finer granularity γ' , there will be more than one $t, t' \in T$ such that $\deg(\gamma', t) \neq \deg(\gamma', t')$, but $\deg(\gamma, t) = \deg(\gamma, t') = t_1$. That is, the t_1 degree of granularity γ will correspond to more than one point at the finer granularity γ' , and so there is no principled way to select one over the other as the starting point of e at γ' (and a similar case can be made for t_2). Figure 4.8 provides an illustration.

As an example, take an event of Mary walking to the station from her house. We can now say that the temporal trace of this event is defined at a granularity level with, say, minute-long units, but it is definitely not defined at a granularity level with microsecond-long units. Thus, it makes no sense to ask at which microsecond exactly did Mary's walk to the station start.

Of course, assuming such a maximally precise temporal trace as in Definition 12 dedicates us to a non-classical logic with truth value gaps, since sentences like those in (4.28) then do not have a truth value with some fine granularity parameters of inter-



Figure 4.8: γ as the granularity of the maximally precise temporal trace.

pretation. Those who wish to retain classical two-valued logic can disregard Definition 12, and continue to assume that any sentence has a truth value at any granularity level. But I find such truth-value gaps welcome, because this is exactly what captures the fact that it does not make sense of talk about when events described by sentences like those in (4.28) started to the exact nanosecond.

We can generalize the notion of the granularity used in the maximally precise temporal trace of an event to event predicates. We could say that a maximally precise granularity function associated with an event predicate φ with extension $\{e_1, e_2, e_3, ...\}$ is a granularity function γ_i which is the granularity function of a maximally precise temporal trace of some e_i in the extension of φ , and no other granularity function γ_j of a maximally precise temporal trace of some e_j in the extension of φ is finer than γ_i . Thus, for instance, granularity functions with a microsecond unit cannot be used in the interpretation of a predicate like *walk to the station*.

This way, we now also have a means to explain why there is a discrepancy between the assertability of the progressive form of accomplishments and of achievements describing their culmination. As Hana Filip (p.c.) noted, although *arrive at the station* can describe the culmination of *walk to the station* (cf. the entailment from *Mary walked to the station* \Rightarrow *Mary arrived at the station*), only *Mary is arriving at the station*, but not *Mary is walking to the station* can be asserted at the point when Mary is just reaching for the handle of the door to the station. Because the temporal trace of events described by *arrive at the station* is by several magnitudes shorter than that of those described by *walk to the station*, it makes sense to suppose that the finest granularity that can be used in the interpretation of the former is much finer than that used for the latter. Thus, while small differences at the culmination point (being near the station) will be distinguishable when talking about arrivals, those very same differences are undefined and indistinguishable when talking about walkings to the station. Thus, it does not make sense to ask whether Mary has strictly speaking walked to the station or is still in the process of walking when she is reaching for the handle of the station door, but it does make sense to ask if she has strictly speaking arrived at the station or is still in the process of arrival.

However, a philosophical issue must be noted at this point. The granularity-relative temporal trace is meant to capture the different possible *conceptualisations* of the temporal boundaries of an event. But if we can restrict the temporal trace to granularity parameters above a granularity that is not the absolutely fine granularity ρ , then what have said about the physical reality? After all, given the realist approach that I endorse (cf. Section 3.3.1), there are actually events, and these must have a runtime, since (at least many) events are spatio-temporally located entities. Should this not entail that the temporal trace of all events is defined at the absolutely fine granularity?

There are several options left open for us to say about objective reality, just as in the case of vagueness in general (cf. Merricks 2001 or Sorensen 2013 for a discussion). We can say, as in Definition 11, that $\tau(\rho, e)$ is defined for each e, and adjust Definition 12 in a way that it specifies a maximally precise temporal trace that is *accessible* to human agents. This would correspond to an *epistemicist* approach to vagueness (according to which there is a sharp boundary between, say, individuals that are children and those that are adults, and we are simply ignorant of this boundary). Alternatively, we can accept that vagueness, including the vagueness of the boundaries of events, is objective and there is in fact no sharp boundary. In this case, we can model the objective runtime of an event with its maximally precise temporal trace according to Definition 12. Since in formal semantics, a modelling of the durativity of predicates is what is relevant rather than the objective runtime of events (given that we can describe objectively extended events with instantaneous predicates), I put this philosophical question aside.

4.3.3 Minimal parts and granularity

4.3.3.1 What is the minimal parts issue?

Based on ideas going back to at least Aristotle (cf. his Nicomachean Ethics, 1174a), states and activities are held to be (*sub*)*divisible*, that is, if they are true at a given inter-

val, then they are true at every subinterval thereof (cf. e.g., Vendler 1957, Bennett and Partee 1972, Dowty 1979, Bach 1981, Verkuyl 1989, Krifka 1989, Rothstein 2004). This is formally captured through the second-order property of *subdivisibility* (aka *divisive reference* as in Krifka 1989, or in an interval-based rather than event-semantic framework, the *subinterval property* as in Bennett and Partee 1972) defined formally in an event semantic framework as in Definition 13.

Definition 13 (Subdivisibility (aka divisive reference) (Krifka 1989, a.o.)) A predicate P is subdivisible (has divisive reference) iff $\forall x, y(P(x) \land y \sqsubset_E x \rightarrow P(y))$, where \sqsubset_E is the strict mereological part-of relation over eventualities.

In words, a predicate is subdivisible iff it applies to all the parts of an entity it applies to. Informally, any part of a running is also a running, but not all parts of running to the window are themselves a running to the window.

Note that some authors, such as Vendler (1957), Taylor (1977), Verkuyl (1989), Lascarides (1991), Link (1998, Ch. 8), Rothstein (2004), Zwarts (2005) or Winter (2006), use the term *homogeneity* — either in informal terms or by giving a formal definition as well — to express that a predicate applies to all parts of an individual or an interval that the predicate applies to, i.e., for the property of subdivisibility. However, *homogeneity* stands more generally, as in, e.g., Link (1983), Moltmann (1991), Piñón (1997), Filip (2000), Arche (2014), for the complex property of subdivisibility *and* cumulativity, where a predicate *P* is cumulative iff $\forall x, y(P(x) \land P(y) \rightarrow P(x \sqcup y))$ where $x \sqcup y$ is the event that is the join of *x* and *y*. This latter use is the one consistent with the terminology used in works on the mass/count distinction in the nominal domain (e.g., ter Meulen 1981, Bunt 1985, Pelletier and Schubert 2003) for which domain these terms were originally introduced (again, building on ideas of Aristotle, cf. his On Generation and Corruption), though the term "homogeneity" is also used ambiguously in the literature on the mass/count distinction (cf. Nicolas 2014, fn. 13).

A crucial difference between states and activities is that subdivisibility is too strong a property for the characterisation of activities: for instance, "there are parts of running events which are just too small to count as events of running" (Rothstein 2004, p. 11). It is therefore a weaker assumption that most authors today (since at least Taylor 1977, Dowty 1979) subscribe to, namely that there is a non-zero lower bound on the size of events in the extension of an activity predicate. E.g., waltzing is not true of events involving less than three steps, running is not assumed to be true of just a single step, and similar considerations apply to all activities. Thus, the received view today is that while states have divisive reference, activities are subdivisible only "down to some small intervals that are necessary to make the predicate true" (von Stechow 2009b). This issue affects the semantics of several constructions. For instance, temporal *for*-adverbials are often assumed to involve universal quantification over an interval, but because of the minimal parts feature of activities, quantification needs to be restricted to a set of contextually given relevant subintervals (cf. Dowty 1979, Moltmann 1991, Zucchi and White 2001).

The minimal parts issue is also a problem for many scale-based approaches to telicity. For instance, on Hay et al.'s (1999) account, telicity corresponds to a non-zero lower bound over the amount of change required by a predicate, while atelicity to an absence of such bound. As such, for activities, this view assumes that *any* amount of change satisfies the relevant predicate, i.e., even one step taken toward the bank should qualify as walking toward the bank, for instance. This correctly captures the intuitive idea about activities that these "are realized as soon as they begin" (Garey 1957, p. 106), but runs into the minimal parts problem. The problem that minimal parts pose also affects a scale-based approach (e.g., Beavers 2010) on which activities require a change to an unspecified degree, i.e., approaches assuming *existentially bound target degrees* for activities: if any degree as a target degree can satisfy the predicate, then, again, even one step toward the bank should count as walking toward the bank. So in order to assume some such scale-based account of activities, one needs to ensure that not any amount of change (say, one step) count as *change* (with respect to, say, walking) in a given context.

4.3.3.2 Previous accounts

Several solutions have been proposed for the minimal parts problem of activities, many of which exploit some form of restriction on parthood. Moltmann (1991), for example, restricts quantification over parts to contextually relevant parts through the notion of a *contextually specified part-of relation*. Moltmann (1991) therefore uses distinct part-of relations, since her contextually restricted parthood "has to be understood not as a part relation in a strict mereological sense, but rather as a contextually determined relation that may be coarser than the mereological part relation, as the relation 'is relevant part of'" (p. 633). Positing two different notions of parthood is a formally acceptable way of approaching the minimal parts problem. Indeed, such as solution has been adopted even in bioinformatics to tackle a minimal parts-like puzzle: Rec-

tor et al. (2005) introduced a transitive *determinate parthood* relation and an intransitive *granular parthood* relation in their semantics.

The drawback of such an approach is that the 'relevant part of' relation "lacks strict formal properties such as transitivity and closure" (Moltmann 1991, p. 633), which makes it more unconstrained than using the familiar mereological part-of relation with its well-known formal properties. Thus, while this is a valid approach, it would be better if we could keep in check the parameters of parthood that can vary across different contexts in contrast to simply using a contextually determined parthood relation as such. To bring an analogy, Deo (2009, p. 503) argued her account of the imperfective relying on a *contextually given minimal interval* is more advantageous over one simply restricting the domain of a universal quantifier, because "the context does not directly provide a set of relevant events or intervals, but rather places a condition on the length of the intervals quantified over by the universal". Like Deo (2009) in her analysis of the imperfective, I also believe that the only parameter of parthood that is needed to vary to account for the minimal parts problem is the *size* of the accessible parts.

The closest forerunner of what I will propose is Link (1998, Ch. 12), who introduced the notion of a *temporal granularity parameter of a type* θ , which is the minimal amount of time required for a process to evolve in such a way that it can count as θ (say, *walking*). He then restricted the homogeneity requirement on activities to subprocesses whose temporal trace exceeds the length of the temporal granularity parameter of the relevant type. I will adopt such a restriction on the subdivisibility (homogeneity) requirement in Section 4.3.3.4, but couched in a more general theory of granularity. In particular, my goal in this section is a generalization of Link's (1998) proposal, *decomposing* his notion of a temporal granularity parameter (a primitive) into the minimally distinguishable unit of time at the most precise one of the granularity levels used most frequently in the interpretation of a predicate. The advantage of this decomposition is that the "granularity parameter" I adopt from Sauerland and Stateva (2007, 2011) (which as a parameter of interpretation should not be confused with Link's (1998) "granularity parameter") has applications in the analysis of other phenomena.

There were other authors after Link (1998) who restricted subdivisibility to a specific (or a non-specific) interval. Winter (2006), for example, introduced the property of *weak downward monotonicity* as characterizing activity predicates. For Winter, *John walked between 2pm and 4pm* is true if *there is* a proper subinterval I' of [2:00, 4:00] such that John walked in all the subintervals I'' that satisfy $I' \subset I'' \subseteq$ [2:00, 4:00]. However, existential quantification may be too weak a requirement: for instance, *John ate at least 3 sandwiches between 2pm and 4pm* can also satisfy the weak downward monotonicity given an event of John eating 4 sandwiches between 2pm and 4pm, but this predicate this telic, not an activity (cf. Filip 2000 for the problem these kind of sentences pose for quantization-based approaches to telicity).

A similar proposal is the *stratified reference* of Champollion (2010), who builds his account on a function ε determining if some measurable entity counts as *small* given some specific measure. For instance, $\varepsilon(\lambda t[hours(t) = 1])(t')$ holds iff t' counts as "very small" with respect to one hour. In what follows, I inspect Champollion's (2010) account in more detail.

4.3.3.3 for-adverbials

I believe a shortcoming of the account of Champollion (2010) is the lack of connection between the activity predicate and the unit size which supplies the parts accessible for quantification in a particular case.

The goal of Champollion (2010) is to offer a uniform account of the minimal parts problem and an explanation of why *for*-adverbials can be used in the dimension of time and space (cf. *The crack widened for 5 meters*), but not speed or temperature (cf. **John drove for thirty miles an hour* and **The soup boiled for 100 degrees Celsius*). He argues that previous explanations of the latter phenomenon in terms of extensive measure functions (e.g., Krifka 1998) or monotonic properties (e.g., Schwarzschild 2002) are insufficient, as is a requirement of the subinterval property for atelics in light of the minimal parts problem.

Champollion (2010) links admissibility of *for*-modification to the availability of pseudopartitive constructions (*five hours of talks, five miles of railroad tracks,* but **five miles an hour of driving, *five degrees Celsius of water*), which, he argues, have the same requirement as *for*-adverbials. However, I am doubtful that this parallel is entirely exact. Dimensions like weight and volume allow pseudopartitives, as the examples *five pounds of rice* and *five litres of water* of Champollion himself show. But these, to my knowledge, do not allow *for*-adverbials:

- (4.29) a. *John dieted for five pounds.
 - b. *John poured (water) for five litres.

This indicates that *for*-adverbials and pseudopartitives do not have the same requirements, *pace* Champollion (2010). This is only important for our purposes inasmuch as one of the most compelling features of Champollion's (2010) account is the explanation of different linguistic phenomena with the same tools. But if, as it appears, the parallel between these phenomena breaks down, then it is not a disadvantage for an alternative account of, say, the minimal parts not to have an explanation for all of these diverse facts.

But let us inspect Champollion's (2010) proposal for *for*-adverbials. According to Champollion, *for*-adverbials presuppose *stratified reference*, not the subinterval property. To reproduce here his example, the predicate *waltz* has stratified reference with respect to dimension τ and granularity $\varepsilon(\lambda t[hours(t) = 1])$ iff the following holds ('*' is the "plural predicate" star operator of Link (1983) that operates on a one-place predicate *P* and generates all the individual sums of members of the extension of *P*; and *W* stands for the predicate described by *waltz*):

(4.30)
$$\forall e \Big(W(e) \to e \in *\lambda e' \big(W(e') \land \varepsilon(\lambda t[\mathsf{hours}(t) = 1])(\tau(e')) \big) \Big)$$

In words, he puts this as follows: every waltzing event consists of waltzing subevents whose runtimes are very small compared to an hour. This is the presupposition of *waltz for an hour*. But consider now the following sentence:

(4.31) John waltzed for just half a minute (and then tripped, so, red-faced, he left the ballroom).

On the theory of Champollion (2010), this sentence would have the following presupposition (cf. (4.30)):

(4.32)
$$\forall e (W(e) \rightarrow e \in *\lambda e' (W(e') \land \varepsilon (\lambda t[\mathsf{minutes}(t) = 0.5])(\tau(e'))))$$

(4.32) says that every waltzing event is made up of waltzing subevents whose runtimes are very small compared to half a minute. Suppose that waltzing events can be of 3 seconds in length, roughly the time it takes to perform three steps. (Note that nothing hinges on what specific length we consider for the purposes of the argumentation.) Therefore, intervals of 3 seconds should count as "very small" with respect to half a minute, because (4.32) requires that *all* waltzing events be made up of very small waltzing-events with respect to half a minute, i.e., including the 3-second waltzing events. But then if waltzing events are allowed to last 3 seconds, then it should be possible to utter the following:

(4.33) John waltzed for just 3 seconds (and then tripped, so, red-faced, he left the ballroom).

But this, in turn, would be required to have the following presupposition:

$$(4.34) \quad \forall e[W(e) \to e \in *\lambda e' (W(e') \land \varepsilon(\lambda t[\mathsf{seconds}(t) = 3])(\tau(e')))$$

This requirement says that every waltzing event is made up of waltzing subevents whose runtimes are very small compared to 3 seconds. This, however, is not satisfied by *waltz*, since waltzings cannot be made up of waltzing parts that are radically shorter then 3 seconds. This is a general algorithm to produce examples problematic for Champollion's (2010) account: since his analysis only takes the *for*-adverbial into account, by setting the *for*-adverbial to correspond to recursively smaller and smaller interval sizes allowed by the stratified reference requirement, after while, the resulting presupposition will (incorrectly) not be satisfied.

4.3.3.4 Minimal parts and minimally distinguishable intervals

Given the above recursive algorithm to produce a counterargument for an analysis of the minimal parts problem relying on a specific measure that is not dependent on the predicate, the advantage of Link's (1998) account, I believe, is that the minimal size of events available for quantification is relativised to predicates. However, instead of using Link's (1998, Ch. 12) dedicated function directly specifying a minimal interval size for a predicate, the tools adopted in Section 4.2 for an account of temporal extendedness are sufficient for an account of the minimal parts problem, as well.

Recall that temporal extendedness in the account proposed in Section 4.2 is relative to a granularity parameter of interpretation and extended parts of an event can only be at least as long as the MDI of the granularity parameter. I therefore propose that the subdivisibility requirement should apply to activity predicates down to the parts which are extended at a normal level of granularity associated with the predicate, where a "normal" level of granularity is used in the sense discussed on page 95 namely, these are the set of granularities with a unit size belonging to a given range such that in most cases when the relevant predicate is used, some member of this set of normal granularities is used.

The subdivisibility requirement of activities could then be modified as follows, building on Link (1998):

Principle 1 (Subdivisibility down to granularity-relative extended parts)

Let *P* be an activity predicate. Take granularity level Γ_m from the set of normal granularities used in the interpretation of a predicate *P* for which there is no finer granularity within this set. For any part e_i of any eventuality in the extension of *P*, *P* must apply to e_i if e_i is extended at Γ_m .

Note that this is not simply a restatement of the problem: the problem of minimal parts is that there is a lower bound on the *size* of events that can fall under an activity predicate, say, *walk toward the bank*. Principle 1 instead claims that an activity, say, *walk toward the bank*, applies to *any* subevent of an event it applies to if the subevent is extended at the finest granularity level in the set of normal granularities associated with the predicate. The size of the subevents down to which subdivisibility is required is encoded in an independently required parameter, the set of normal granularity functions associated with a predicate.

So, for example, *John ran for a while* is evaluated in 95% of the cases with respect to a granularity level in { $\Gamma_1, \Gamma_2, ..., \Gamma_n$ } over time, the finest one of which is Γ_m , with a unit size of, say, 5 seconds. In order for *John run*^{\emptyset} to qualify as an activity, it has to be subdivisible down to extended parts at Γ_m . So all parts of an event of John running that are accessible as extended parts at Γ_m should be events of John running. And indeed, all parts of John running greater than 5 seconds are events of John running, so subdivisibility is satisfied. At the same time, the event of John running still has parts smaller than 5 seconds, but they are only accessible for quantification when switching to a granularity finer than Γ_m .

The account of the minimal parts problem proposed here bears some connection to the solution that Chierchia (2010) put forth for the nominal domain, inasmuch as it employs a tool originally introduced for the analysis of *vagueness*. Chierchia uses supervaluations, and proposes to analyse the difference between count and mass terms and still retain the idea of the existence of minimal parts (atoms) in a given context by treating the predicate "be an atom" as vague. The present account achieves a similar result via the use of granularity functions also originally proposed by Sauerland and Stateva (2007, 2011) for analysing what they call scalar vague predicates — without, however, being dedicated to the vagueness of the notion of an "extended part" (at least not to its being *epistemically* vague, cf. Section 4.3.1).

for-adverbials revisited. On the present account, the size of the minimal parts of an activity is dependent on a normal granularity parameter (which is dependent on the predicate), rather than the granularity parameter used in the interpretation of a sentence (which is dependent on a number of things including the relevant predicate, the *for-*adverbial, other linguistic components, and the context). All we need of a *for-*adverbial is the usual requirement (in particular, presupposition) of the subdivisibility (but in the sense of Principle 1) of the predicate it modifies. Given this presupposition, *for-*adverbials do not need to have a universal quantificational analysis, but can be assumed to have the same asserted content as *in-*adverbials, namely, an event duration measuring interpretation. So the asserted content of (4.31) and (4.33) at granularity γ is as follows, respectively:

(4.35) a.
$$\exists e \exists t (W(e) \land \tau(\gamma, e) = t \land (t <_{\mathcal{T}} \mathsf{now}) \land \mathsf{minutes}(t) = 0.5)$$

b. $\exists e \exists t (W(e) \land \tau(\gamma, e) = t \land (t <_{\mathcal{T}} \mathsf{now}) \land \mathsf{seconds}(t) = 3)$

The universal quantification over proper parts follows from the subdivisibility of predicates admissible for modification by a *for*-adverbial. Assume that the unit size of the finest normal granularity Γ_m associated with *waltz* is 3 seconds. If *John waltzed for half a minute* is true, then he was also waltzing during each substantially great (i.e., minimally 3-second) part of the 30-second waltzing event. At the same time, *John waltzed for 3 seconds* can be true of an event, while it is not required to be true of smaller parts of that event, since subdivisibility only requires *John waltz* to be true down to the extended parts of waltzings at the normal granularity Γ_m of *waltz*, i.e., down to 3 second-long parts. (Note that this view is compatible with *waltz* being true of events shorter than 3 seconds: it is simply not *required* to be true of such events.) Making the requirement of the *for*-adverbial refer to the set of normal granularities associated with a predicate precludes the application of the recursive step of requiring smaller and smaller minimal sizes.

Open questions. Of course, it should be emphasized that this is by far not an exhaustive account of the minimal parts problem. For example, it suffers from a problem of sharp boundaries similar to that faced by many accounts of vague predicates, including supervaluations, scale-based approaches to gradable predicates like that of Kennedy (2007), and also granularity functions themselves (cf. Section 4.1.2). Given that we can always identify the smallest unit among those defined by the normal granularities associated with a predicate, I (just like Link 1998) predict that there is a specific interval size for each predicate such that the predicate is required to hold of subevents

down to that size. Importantly though, the subdivisibility requirement does not preclude the possibility that the predicate can apply to shorter events, it merely does not require it.

A further open question is how exactly the set of normal granularities come to be associated with a predicate, that is, *why* speakers use in the interpretation of a specific predicate by vast majority some granularity level whose unit size belongs to a limited range. This question is especially pertinent given that the minimal parts problem is not merely a linguistic issue, it is also an ontological one, and the present approach to this problem relies on the language use of speakers through reference to normal granularities. However, which contexts the speakers use a predicate in is dependent on its extension, and more specifically, it seems plausible to assume that the granularity parameters used most in the interpretation of a predicate are dependent on the size of the events in its extension.

A number of authors (e.g., Vanden Wyngaerd 2001, Tatevosov 2008) have argued for the need to associate some measure of an average, typical runtime with eventuality predicates. And once we accept that predicates are associated with a measure (of temporal duration), then they should be like any other measure expressions, and be associated with increasingly coarser granularities at measures of increasingly greater magnitude (for instance, *one million* is generally interpreted less precisely that *one hundred*). This is rooted in *Weber's law*, an influential principle of psychology, according to which the size of the difference threshold for measurable categories is proportional to their original size (to illustrate it with an example from Fults 2011, 7 and 8 are discriminated more or less as easily as 70 and 80, or as 700 and 800).

This psychological principle predicts that predicates with a significantly greater average runtime (such as *a planet cool*_{\emptyset}) tend to be interpreted at a coarser granularity. This seems to be borne out: for instance, in a narrative discourse, the time-lag tolerated between events described by the two sentences of the discourse (which intuitively is related to the *unit* of the granularity parameter of interpretation) is greater given predicates with substantially greater average runtime (cf. (4.36a)–(4.36b), but also examples by Dowty 1986, p. 46ff and Link 1998, p. 255).

- (4.36) a. Susan walked in. Peter left.
 - b. The Earth cooled. The oceans formed.

I believe that what the "normal" granularities associated with a predicate are is a matter of study for psychology rather than semantics. In this, I side with Dowty (1986),

who remarked that what counts as "immediately after" is "only determined by the hearer's understanding of the nature of events being described in the narrative, the overall degree of detail in which events are being described, and common knowledge about the usual temporal relationships among events" (*op. cit.*, p. 47).

What I wished to show here instead is that we can attack the problem of minimal parts with tools we need in our semantics in any case, and so we do not require additional tools (such as weak downward monotonicity) for specifically the minimal parts problem.

Chapter 5

PRELIMINARY PROCESS INTERPRETATIONS

This chapter offers a semantic analysis of the progressive in a novel scale-based framework with the goal of deriving the correct interpretations for progressive achievements. I first present a novel conception of the scale of change as a series of snapshot characterisations describing the change that is relevant in a discourse context, and introduce some auxiliary notions including cover events and primary and secondary scales of change. This forms the basis for a scale-based characterisation of aspectual classes, and a novel scale-based definition of the progressive. In the bulk of the chapter, I concentrate on preliminary process interpretations, and show how the unique behaviour of progressive culmination-achievements can be explained. The chapter will be concluded with a brief look at how the counterparts of the progressive in some languages other than English can be captured by varying some parts of the definition of the progressive.

5.1 Scales of change

As foreshadowed in Chapter 2, given the problems faced by different event semantic analyses of progressive culminations, I propose to omit reference to events in the extension of the input predicate of the progressive and offer a semantics of the progressive such that its truth conditions depend on *scales of change*, instead. As noted in Section 2.3, the range of predicates associated with a scale in previous scale-based theories of aspect does not fully correspond to those that are admissible in the progressive (e.g., *watch TV* or *rain* is not associated with a scale on any previous account, but they are felicitous in the progressive). This means that a novel conception of a "scale of change" is necessary if we want to relativise the truth conditions of the progressive to it, one which is more general and applies to all the predicates that can appear in the progressive. Also, my goal is to have scale mereology take the place of event mereology in the semantics of the progressive, but this means that a scale of change needs to be associated directly with eventualities and not only predicates, so that it is possible to express the condition that the scale of an ongoing eventuality is *part* of the scale of the predicate argument of the progressive operator. Thus, I need a notion of a scale of change that is essentially a perspective over an eventuality, and then generalize this notion to predicates based on the scales of change associated with eventualities in the extension of the predicate.

As a crucial further step, I need to model the dual-faced nature of culminations like *arrive*, which behave similarly to happenings in combination with temporal adverbials and other modifiers, but are, unlike happenings, felicitous in the progressive. The way this can be achieved in a scale-based framework is to associate them with *two* kinds of scales, one of which are the scales associated with events in the extension of the relevant culmination predicate, and a second kind which are the scales associated with events of *gradual change* that culminate in events in its extension. That is, a novel idea is to have a dual scale-based characterisation of a predicate, where one type of scale characterises its inherent properties (or asserted change), and the second type its relational properties (or presupposed change). The goal of this section is to introduce the above ingredients of the scale-based theory of aspect that will be employed in a new analysis of the progressive.

5.1.1 The scale of change as a series of snapshot characterisations

There are a number of different ways to implement an idea in a specific formalism.¹ The framework to be presented here—building primarily on the work of Beavers (2002, 2008, 2011, 2012, 2013), but still radically departing from it—lends itself well to accounting for the semantics of the progressive, but there are several possible alternative ways of formalizing the ideas presented here, the choice among which should be determined by a study of further phenomena.

Recall from the discussion in Section 2.3.2.4 that the framework I need has to be much more general than that of Beavers, applying to all predicates (with the possible exception of non-temporary, individual-level predicates which Klein 1994 calls 0-state

¹For instance, Beavers (2002, 2008, 2011, 2013) uses different formalisations of his framework, sometimes (as in Beavers 2013) even within a single paper.

predicates like 2 plus 2 is 4 or be a mammal). Thus, as a radical departure from former scale-based accounts, I assume that the *scale of change* does not correspond to an incremental argument, but is simply a series of sets of states. Given that non-scalar predicates like *watch TV* and *rain* cannot be uniformly and uniquely characterised with a series of states, I also depart from former scale-based accounts in assuming a "bottom-up" framework. In particular, I assume that a scale of change is defined for an *eventuality*, rather than directly for a predicate.

To this end, I draw on the concept of a *perspective* over a situation from Asher (1992) (cf. Section 2.2.5): a scale of change for an eventuality is a series of degrees capturing the change associated with an eventuality relevant in a discourse context. Where I depart from Asher's (1992) notion of a perspective is that

- *i*) instead of characterising a complete eventuality or situation as a *whole*, I characterise different temporal parts of the eventuality (as well as intervals directly preceding or following it) separately, drawing on the scalar tradition of modelling aspect, and
- *ii)* instead of using *propositions* to describe the contextually relevant aspects of an eventuality, I use sets of states (i.e., predicate extensions), though this is a purely arbitrary decision based on the idea that using a set of states instead of propositions to characterise event stages is closer in spirit to the event semantic framework I adopt. But every formal definition in what follows can be adapted to an approach closer to that of Asher (1992) on which the degrees of a scale of change are propositions.

In sum, in the present framework, a degree of a scale of change is a set of states which captures the contextually relevant properties of a particular "snapshot" of the relevant eventuality. In essence, the scale of change can be seen as a "scalarized perspective" over a situation, a perspective with temporal distinctions.

For the sake of illustration, we can conceive of a scale of change as a *film*: just as a film in fact consists of individual snapshots, i.e., frames or photos, which capture one static stage of the happening in the film, a scale of change consists of snapshots, i.e., sets of states, which capture one static stage of the ongoing change. But just as there are a number of things we are left in the dark about when watching a film (such as what the far side of objects in the film looks like), so the scale of change disregards certain properties of a particular event and only encodes the properties that are relevant in a given context.

And just as different films can be made about the same event depending on the goals of the director, the same event can be associated with different scales of change in different discourse contexts, depending on the goals of the interlocutors. To take an example, the same event of a game of chess can be presented as the winning party winning or as the losing party losing depending on the *focus* of the film or the discourse. This indicates close ties of the present proposal with theories of discourse coherence and attention — in fact, a connection between theories of discourse coherence and film directing has been recently proposed by Cumming et al. (2014), which indicates that there may be a deeper logical connection between discourse coherence, agent goals, the scale of change as a "scalarized perspective" over situations (and also the progressive, which I will analyse in terms of these scalarized perspectives), and the film analogy above.

Formally, the scale of change is defined as follows.² (Note that the granularity argument of $\sigma_{c,w}$ is a granularity function over the *time scale*, and not the scale of change itself.)

Definition 14 (Scale of change (of an eventuality))

 $\sigma_{c,w}: G_{\mathcal{T}} \times E \to \mathbb{S}_{\mathsf{stat}}$ assigns, for a context $c \in C$ and world $w \in W$, the scale of change to an eventuality at a particular granularity, where $G_{\mathcal{T}}$ is the set of all granularity functions over the time scale \mathcal{T} , E is the set of all events and $\mathbb{S}_{\mathsf{stat}}$ is the set of all scales whose degrees are sets of states.

For example, the scale of change assigned to an event of John's walking to the station in a particular discourse context can consist of sets $d, d', d'' \dots$ of states such that John has a specific proximity to the station in each state, with this proximity being greater in states in higher degrees of the scale, as illustrated graphically in Figure 5.1.

In a different discourse context, the same event may be associated with a different scale of change. The predicate used to describe the event is usually one of the key factors influencing which aspects, which properties of the event are relevant, but it is not the sole factor. For instance, when uttering *John walked to the station* in reply to the question, *Did John get any exercise at all today?*, then arguably the relevant aspect of the change going on during the walking event is the number of steps John took or perhaps the number of calories he burned as the event progressed.

²The decision to have the context and the world parameters as indices rather than as arguments of σ is based purely on considerations of simplicity and relevance: these two parameters will not be as important when talking about scales of change.



Figure 5.1: An illustration of the scale of change associated with an eventuality of John's walking to the station in a particular world and discourse context.

The scales of change associated with an eventuality *predicate* (such as *John walk*^{\emptyset} to *the station*) are then simply the set of all scales of change associated with all events in its extension, but see Section 5.1.3 for further details. The present theory can be seen a generalized version of Klein (1994), who captured the profile of change associated with different predicates through state schemata, cf. Table 5.1. But while Klein used binary distinctions (a state either holds or does not hold), a scale of change-based approach exploits finer degrees of differentiations.

Schema	Predicate type	Example
	0-state predicates	2 plus 2 is 4.
– – –	1-state predicates	The light was on.
$\dots + + + \dots$	2-state predicates	John built a house.

Table 5.1: The predicate schemata of Klein (1994)

States and scales of change. A departure from former scale-based approaches like those reviewed in Section 2.3.2 is that I assume, in the spirit of Klein (1994) and Jack-endoff (1983), that (state individuals and thereby also) state predicates are also associated with "scales of change". Even though this idea is not indispensable for an account of progressive achievements, it is a natural generalization of the scale-based approaches. According to the generalization I propose, the scale of change captures the profile of change associated with an eventuality or a predicate. The way it captures the profile of the change associated with states, that is, *lack* of change, is by associat-

ing them with a degenerate scale consisting of a single state predicate. Taking the film analogy, a state can be visually captured with a single frame of film, i.e., a photo.

This generalization was already the intention of Beavers (2002, §6), but he noted that given his definition of the Movement Relation (cf. Definition 3) and the extra restrictions he laid on it, unwanted predictions would result for states, namely, that they cannot be durative, which is why he forgoes this generalization. The restriction I lay on the mapping between an eventuality and its scale of change (see Appendix B) is different, and is compatible with states having extended temporal traces; thus, it effectively implements the desiderata of Beavers's (2002) theory without the unwanted consequence of prohibiting durative states.

Indeed, on the framework I propose, what would actually require an extra support would be to have state particulars (and hence state predicates) *not* be associated with a scale of change: since a scale of change is a *perspective* over an element of the set *E* of events, and since states are a subset of this set according to the model assumed in Section 3.3, then states must also be associated with scales of change. Indeed, Asher (1992) exploited only perspectives over states or situations (Asher used these terms interchangeably) in his theory of the progressive. Since the present notion of a *scale of change* is a generalization of Asher's (1992) perspective, it must apply to all cases which Asher's perspective did, including states.

Put in a broader perspective, a scale of change on the present account is associated with any eventuality particular — and one major property of eventuality particulars is that they are located in *time* (Parsons 1990). Thus, moving to the level of predicates, I propose that scales of change should be associated with all predicates which can be modified with *temporal adverbials* (which I take to be an indication that these predicates denote sets of eventualities, which are located in time). Because all dynamic predicates, as well as temporary states can be modified with temporal adverbials (cf. John was at home yesterday for an example with a temporary state), then, on this view, all of these predicates must be associated with scales of change. This foreshadows my proposal in Section 5.2 that some stative predicates, however, are not associated with scales of change according to this view, because they cannot be modified with a temporal adverbial. These predicates are individual-level states like *be a mammal*: cf. the infelicitousness of *#This animal was a mammal yesterday*. The only scenario on which such a sentence is felicitous is one on which the animal was a mammal yesterday, but is not a mammal today (cf. Altshuler and Schwarzschild 2013 on the cessation implica*ture* of past tense sentences), that is, in a context where *be a mammal* is a stage-level, rather than an individual-level predicate.

An extension of the scale-based semantics to states yields a closer parallel with the framework of Jackendoff (1983, 1996) given the following points of connection:

"[The abstract motion described by a verb is reduced] to the more primitive function BE – as is desired by those who wish to treat motion as a series of locational 'snapshots'." (Jackendoff 1996, p. 323)

"What distinguishes an Event from a State? Intuitively, an Event has a time-course, whereas a State 'just sits there'" (Jackendoff 1996, p. 327)

On the present approach, the scales of change associated with dynamic predicates contain at least two degrees, i.e., consist of several states ('snapshots'), while the scales of change of states are single-valued, i.e., a single 'snapshot', corresponding to the state itself.

"However, the reduction is in terms of a continuous function, thereby meeting the objections [...] to the 'snapshot' approach." (Jackendoff 1996, p. 323)
In the case of activity and accomplishment predicates, their scales of change are generally densely ordered, satisfying the continuity property mentioned by Jackendoff.

Given that I adopt Jackendoff's (1996) approach to dynamic predicates as a series of 'snapshots', the proposed representation of activities will be compatible with both a *homogeneous* view of these predicates (cf. Bennett and Partee 1972, Dowty 1979, Krifka 1992), as well as with one according to which activities consist of a continuous *series* of multiple, possibly an infinite number of changes from one state to another (John Beavers, p.c., and cf. also Pustejovsky 1991 for a view of activities as a sequence of events).

Note that the scale-based approach ensures that multiple "snapshots" do indeed correspond to "change" in the intuitive sense of the term, because no two degrees of a scale of change can be identical (because a scale is a linearly ordered set, and a set can only contain the same element once); and if different snapshots of the event are associated with different sets of states, a change of state is implied:

Principle 2 (Non-degenerate scales of change imply change)

For any c, w, γ, e , if the carrier set of $\sigma_{c,w}(\gamma, e)$ contains more than one degree, then e is associated with a (gradual or non-gradual) change.

Incremental arguments. While using a *scale of change* consisting of states, I also retain the notion of what is called a *scale of change* by Beavers (2012, 2013), which is the linearly ordered set that measures out an event in the extension of scalar/incremental predicates. I will call this scale an *incremental argument* based on Beavers (2013). This incremental argument (following authors like Tenny 1987, Dowty 1991, Jackendoff 1996, Filip 1993/99, Hay et al. 1999, Filip 2008, Beavers 2013, a.o.) unifies the incremental theme of creation and consumption verbs like *drink*, the path argument of directed motion verbs like *walk*, and the property scale of degree achievements like *cool*.

The Movement Relation of Beavers (2002, 2008, 2013) (reproduced above in Definition **3**), a mapping between the incremental argument and the event, is defined for scalar/incremental predicates, but not others like *rain* or *watch TV*. Given a homomorphic mapping between the incremental argument and the event, and a homomorphic mapping between the event and the scale of change, the incremental argument is also homomorphically mapped to the scale of change, constraining its structure. In what follows, I will not be concerned with the specific behaviour of scalar/incremental predicates, and will only focus on what I call the scale of change, which is a more general notion than the incremental argument to represent change.

5.1.2 The mapping to the scale of change

The scalar trace. I assume a mapping called the *scalar trace* to the scale of change $\sigma_{c,w}(\gamma, e)$ of an eventuality *e*. This mapping is similar in function to the Movement Relation of Beavers (2002, and elsewhere) — that is in turn built on the homomorphic mappings of Krifka (1992, and elsewhere). The precise details of the formalisation of the scalar trace are not directly relevant to the analysis of progressive achievements I will present, so I omit its definition here, and refer the reader instead to Appendix B, where it is characterised in detail. I have drawn on Beavers (2002) in characterising many of the formal properties of the scalar trace to ensure the same results: for instance, the scale of change associated with an eventuality respects temporal flow, i.e., progressing to *higher* degrees on the scale change corresponds to the characterisation of *later* temporal parts of the eventuality. (For instance, in the case of an event of John walking to the station, later parts of the walking event must be associated with a set of states in whose case John's proximity to the station is greater.)

Importantly for our purposes, just as Beavers's (2002) constraints on his Movement Relation guarantee that durative dynamic events are associated with *multi-valued* scales while non-durative dynamic events with *two-valued* scales, the scalar trace also guarantees such a correspondence between durativity and scale complexity, as elaborated in Appendix **B**.

An important departure from most of the scale-based approaches to aspect and in particular that of Beavers (2002, 2008, 2011, 2012, 2013) in characterising the mapping to scales of change is that I assume that the scale of change is not linked to eventualities directly, but their relation is mediated by another scale (usually time), as in Gawron (2005, 2009) and Deo and Francez (2013). The advantage of defining the scalar trace as a mapping from some independent ordered set like time rather than directly from the event itself is twofold:

- we can utilize the scale of change to explain not only phenomena restricted to the temporal trace of an event, but also phenomena involving intervals preceding and following it;
- by being able to vary the dimension of the mediating scale, we can have a uniform account of a wider range of phenomena.

Characterising achievements. If the scalar trace function can capture states of affairs outside the temporal boundaries of the eventuality, it is possible to characterise result states, as well as intervals leading up to an event in terms of the same scale of change that is used to describe the structure of the event itself.

This is particularly crucial in the case of events described by achievements. Endorsing the approach to duratvity of Beavers (2002, 2008, 2013), events in the extension of achievements are associated with a binary scale of change (for instance, for an event of John's arrival at the station, a set of states of John not being at the station and a set of states of John being at the station). In order for Beavers's generalized movement relation between event parts and scale parts to be defined, he has to assume that events in the extension of achievements are composed of *exactly two* parts (cf. Section 3.1.2).

On the present proposal, events in the extension of achievements can be literally instantaneous (i.e., their temporal trace is a single point in time, at least at the relevant granularity), and their scale of change still consists of two degrees. This is possible because the first degree can be the image of a point (if time is discrete) or all points in some contextually determined interval³ (if time is dense) immediately preceding the pointlike temporal trace of the event (in our example of John's arrival at the station, all points in an interval preceding John's arrival can be mapped to the set of states of

³Recall that the scale of change is context-relative.

John not being at the station), while the second degree of the scale of change can be the image of the point that is the temporal trace of the event (in our example, the time of John's arrival can be mapped to the set of states of John being at the station). Thus, like Piñón (1997), and Dowty (1979, §§2.3.2, 3.3), we evade the problem of instantaneous change by relegating the change information to outside the boundaries of the event (or in the case of Dowty, the interval at which BECOME ϕ is true for some proposition ϕ).

Varying the mediating scale. As for the second gain of adopting a mediating linear order, it enables us to accommodate cases where the relevant ordering is spatial or some other dimension. These are the so called (*spatial and abstract*) *extent* readings analysed extensively by Gawron (2005) and Deo and Francez (2013), and illustrated by the authors with examples like the following:

- (5.1) a. The road widens between San Francisco and San Jose.
 - b. The plot thickens in chapter three.
 - c. Fish ears grow with increased CO₂.

We can now accommodate the analysis of Gawron (2005) and Deo and Francez (2013), who argue for a mediating linear order in the case of degree achievements, whose identity can explain the observed (temporal, spatial extent, abstract extent) readings of a predicate like *widen*. Note that in our case, it is not only degree achievements, but all predicates which are assumed to have such a mediating structure. And I propose that it is welcome, since many other different kinds of predicates can display such readings when supplied with some comparative expression, cf.:

- (5.2) a. It rains more as you go farther south.
 - b. People eat more fish as you go toward the ocean.
 - c. You are more irritated, the less sleep you get.

As shown by (5.2c), or the famous description of Douglas Adams's *Mostly Harmless* as *The fifth book in the increasingly inaccurately named Hitchhikers Trilogy*, (abstract or physical) spatial readings are even available to states. And as these examples show, while the scale of change of states is necessarily a singleton when mapped to a temporal scale, they can have multivalued scales of change when mapping happens along another dimension, such as the scale consisting of the increasing number of books in the Hitchhikers series.

The difference between degree achievements and other predicates is that a comparative like *more, better* or *increasingly* is needed in the latter cases, and this is because only degree achievements lexicalize comparatives (cf. their received analysis based on an increase in degree, as in Hay et al. 1999), that is, only these verbs lexicalize their associated scales (cf. e.g., Rappaport Hovav 2008, Fleischhauer and Gamerschlag 2014) — on our account, their incremental argument which constrains the scale of change.

Note that such a comparative construction with, e.g., *more* is not available to all verbs, just as verb gradation with 'very' is not universally possible (Fleischhauer 2013). For one, *more* cannot be added to telic predicates, which is true in the case of degree achievements, as well: although *John cooled his soup more than Mary* is acceptable, **John cooled his soup to room temperature more than Mary* is not, the standard explanation being that the degree argument is already saturated. But I propose that many, if not all, of those predicates that do combine with *more* also show the same potential for a temporal/spatial/property dimension ambiguity as degree achievements.

In sum, it is expedient to assume an intermediate level of ordering mediating between an event and its associated scale of change for at least two independent reasons. However, in what follows, I will ignore cases when the mediating scale is not time, given that my goal here is an account of temporal aspect. Thus, "scales of change" and "scalar traces" are hence restricted to those which are relative to the time scale.

5.1.3 Predicate-level scales

Predicate-level scales of change should intuitively capture the change that is common to all events in the extension of a predicate. Therefore, just as a predicate extension is a set of events, I assume that predicate-level scales of change are sets of event-level scales of change. To take our film analogy, the scales of change associated with a predicate can be thought of as all the possible films that can be made about the same kind of event, for example, of a girl drinking a cup of water.

Formally, the set of the *primary scales of change of a predicate* is the set of all the scales of change associated with all the eventualities in the extension of an eventuality predicate:

Definition 15 (Primary scales of change)

If $\uparrow \varphi$ is an eventuality property (i.e., $\llbracket \uparrow \varphi \rrbracket^{M,w,g,c,\gamma} \colon W \to \wp(E)$), then the set $\mathsf{PrSoC}(\uparrow \varphi)$ of the scales of change associated with $\land \varphi$ is the set of all scales assigned to all possible φ events, that is, $\mathsf{PrSoC}(\uparrow \varphi) \stackrel{\text{def}}{=} \{\sigma_{c,w}(\gamma, e) \mid \gamma \in G_{\mathcal{T}} \land c \in C \land w \in W \land e \in \llbracket \varphi \rrbracket^{M,w,g,c,\gamma} \}.$

Note that the definition refers to eventuality properties, i.e., a natural language predicate is characterised by the scales of change of all events in its extension at all possible worlds.

I will also talk about the scales of change of predicate φ at a particular granularity γ , which is the set of all scales $S \in PrSoC(^{\varphi})$ such that the granularity argument in S is γ . This set of scales can be designated as $PrSoC_{\gamma}(^{\varphi})$.

The primary scales of change of a predicate capture the properties of the events in its extension. As such, all of Vendler's achievements (happenings and culminations alike) are associated with two-valued primary scales of change, because events in their extension are instantaneous dynamic events, and are as a result associated with twovalued scales. On the film analogy, to visually capture an achievement, it is enough to have to snapshots, two photos, for instance, of John not being at the station and of John being at the station (for *John arrive*_{\emptyset} *at the station*), or two photos of John not seeing Mary and of John seeing Mary (for *John notice*_{\emptyset} *Mary*).

However, the idea about culminations like *arrive* going back to at least Carlson (1981) is — continuing with the film analogy — that they are also implicitly associated with longer "films", for instance, a film about walking to the station. This is often captured as culminations *presupposing* the occurrence of a preceding activity (cf. Section 3.2): this activity is not part of their asserted content (cf. Heyde-Zybatow 2008 for an explicit account), and events in the extension of culminations are not required to be temporally extended (for an alternative view, see Filip 1993/99, Altshuler and Filip 2013, discussed in Section 2.2.2). Instead, events in the extension of culminations stand in a particular *relation* to extended events of gradual change. In what follows, my aim is to determine the nature of this relation in order to exploit it in the characterisation of culmination predicates.

5.1.3.1 Cover events

In order to analyse the difference between happenings and culminations, I will make use of the auxiliary notion of a cover event. Where these two kinds of achievement appear to differ is that culminations like *arrive* (but not happenings) describe events which are typically preceded by a gradual change (cf. Section 3.2). The term *cover event* refers to the event of this gradual change whose final degenerate part is the event described by a culmination predicate (i.e., the cover event contains the culmination point as well as the "process part").

But as the discussion of Rothstein's (2004) event-mereological solution in Section 2.2.4.2 revealed, the simple characterisation "final degenerate part" is insufficient for our purposes: The set E of events forms a join-semilattice, and so the join of two events e, e' is always an event itself, as well (cf. Section 3.3.1). Thus, the join of any arbitrary event of gradual change (say, an event of Mary walking toward John) with an instantaneous dynamic event (say, an event of Mary noticing John) is also an event, but we do not want to call their join a cover event of the latter. We have to refer to something more substantive than simple event mereology, and this is where scales of change can be put to use.

To this end, let us define the cover event relation as a relation between an extended event and its final degenerate part such that the change associated with the extended event is an elaboration of the change associated with the degenerate event. In terms of our contextually determined scales of change, this means that the sets of states making up the degrees of the scale of the extended event (say, an event of walking to the station) must be subsets of the sets of states making up the degrees of the scale of the degenerate event (say, an event of arrival at the station). Formally, the cover event relation is defined as follows.

Definition 16 (Cover event)

For any $e, e' \in E$ such that $e \sqsubset_E e'$, e' is a cover event of e iff for some context $c \in C$, world $w \in W$ and granularity function $\gamma \in \mathbb{G}$,

- $\sigma_{c,w}(\gamma, e)$ is binary and $\sigma_{c,w}(\gamma, e')$ is multi-valued and
- there is a mapping cov from $\sigma_{c,w}(\gamma, e')$ to $\sigma_{c,w}(\gamma, e)$ such that
 - 1. the maximal degree, and only the maximal degree, of $\sigma_{c,w}(\gamma, e')$ is mapped to the maximal degree of $\sigma_{c,w}(\gamma, e)$, and
 - 2. all non-maximal degrees of $\sigma_{c,w}(\gamma, e')$ are mapped to the minimal degree of $\sigma_{c,w}(\gamma, e)$, and
 - 3. $\forall d \in \sigma_{c,w}(\gamma, e'): d \subseteq cov(d)$, that is, all degrees (as sets of states) of the scale of change of the cover event e' are subsets of the degree of the scale of change of event e to which they are mapped.



Figure 5.2 provides an illustration of the cover event relation between events.

Figure 5.2: The cover event relation between events. (Parameters not important for the illustration are suppressed for simplicity.)

The key part of the definition is the final requirement. This ensures that not only is e a mereological part of e', but the change happening throughout e' is an elaboration of the change associated with e. For example, take an event of John walking to the station and its final part, the event of John's arrival at the station in that scenario. Then the scale of change associated with the instantaneous event of John's arrival consists of i) the set of states of John not being at the station, and ii) the set of states of John being at the station. The scale of change of John's walking event in turn consists of sets of states of John being located closer and closer to the station (as in Figure 5.1 on page 118), with the maximal degree consisting of states of John being located at the station. Then the non-maximal degrees of the scale of the walking event are all different subsets of John not being at the station is identical to (and hence a subset of) the maximal degree of the scale of John's walking event is therefore the cover event of John's arrival in this scenario.

In contrast, take the following scenario also with an extended event and its final degenerate part. Let *e* be an event of Mary noticing John, and *e'* a composite event of Mary walking toward John and noticing him. The scale of change of *e* is an ordered pair of the set of states of Mary not being aware of John and the set of states of Mary being aware of John. But as for *e'*, the series of states characterising the change that is going on while Mary is walking toward John are not elaborations of the set of states characterised by Mary not being aware of John: they are sets of states of Mary being located at a given distance from John. These are *not* subsets of the set of states of Mary not being aware of John in many states of being located at a distance from him at which point she is not yet aware of him in the given

scenario). Therefore, e' is *not* a cover event of e, because the final requirement on the cover event relation is not satisfied. Even if the set d of states of Mary not being aware of John were an element of the scale of change of e', any other degree d' of the scale of e' would need to be different from this set (which follows from treating scales as linearly ordered sets, cf. the discussion in relation to Principle 2 above). But there are no graded differences in Mary's awareness of John during e', and so d' cannot be a subset of d, contradicting the final requirement on cover events.

As the example reveals, the cover event relation differentiates between instantaneous events in the extension of culminations and instantaneous events in the extension of happenings, exploiting the idea that the former, but not the latter are typically characterised by a gradual change leading up to them (cf. Section 3.2). To foreshadow the analysis of progressive culminations in Sections 5.3–5.4, I will assume that a progressive culmination like *John is arriving at the station* can be true during the runtime of the cover event of an event in the extension of the culmination. The scale-based definition of the cover event will then exclude progressive happenings like *Mary is noticing John* to be true in the interval leading up them.

An interesting case is presented by events in the extension of the predicate *die*. Given that a progressive culmination is true during a cover event, the famous aphorism, *From the day we are born, we are dying* can also be accounted for. In this case, the gradual change that goes on from the moment humans are born until they die is their being progressively older. Thus, the complex event that is the life of a human is in this special context associated with a scale consisting of degrees with states of being of a progressively greater age. These are all subsets, and what is more, different subsets of states of being *alive*, which is the set of states that is the first degree of the scale associated with a dying event in this context. The event corresponding to the life of an individual is therefore a cover event of the event corresponding to the death of the individual. The progressive of the predicate describing the dying event will therefore be predicted to be true during the greater event of being born, living and dying in this special discourse context.

From events to predicates. Although it will not be needed for the formal analysis, we can in some cases also talk about type-level cover event predicates in the following sense. An accomplishment predicate ψ is a *cover event predicate* of achievement predicate φ just in case for all worlds w and some context c and granularity γ , $\forall e' \in \llbracket \psi \rrbracket^{M,w,g,c,\gamma} : \exists e \in \llbracket \varphi \rrbracket^{M,w,g,c,\gamma}$ such that e' is the cover event of e. That is, whenever an event belonging to the extension of accomplishment predicate ψ occurs, then

its final non-extended part belongs to the extension of the culmination-achievement φ . Thus, the extension of the achievement φ is a superset of the set of events that are the final non-extended parts of events belonging to the extension of the accomplishment ψ .

The cover event predicate relation can be linguistically detected, for instance, by entailments in the perfective: e.g., right after culmination, (5.3a) entails (5.3b).

- (5.3) a. Mary has (just) climbed to the top of Mount Everest.
 - b. Mary has (just) reached the top of Mount Everest.

No corresponding entailments with an accomplishment can be found for happenings.

Note that the way a cover event is defined, a culmination predicate may have several cover event predicates. For instance, there may be different ways of arriving somewhere (on foot, by plane etc.), and so *walk to the station* and *drive to the station* are both cover event predicates of *arrive at the station*. Not all culminations are associated with cover event predicates in a language. For instance, for culmination predicate *win* or *die*, there appears to be no accomplishment predicate describing a gradual change culminating in winning or dying other than some lengthy paraphrase or a complex predicate including the predicate *win* or *die* itself.

5.1.3.2 Secondary scales of change

Given the notion of a cover event, we now define the notion of a *secondary scale of change*. A secondary scale of change serves to capture the intuitive idea that culminations are characterised not only by their own asserted instantaneous change, but also by the presupposed gradual changes that lead up to events they describe. On our film analogy, all films about, say, walking to the station (i.e., films about complete walks to the station) are necessarily also all films about arriving at the station, so *arrive*₀ *at the station* can be presented as a film about a walk to the station equally as well as two photos of not being at the station and of being there.

Admittedly, a dual characterisation with a primary and a secondary scale is stipulative, rather than following from some more basic principles, but it can be used to explain the ambivalent behaviour of culminations. That is, using two-valued primary scales and multi-valued secondary scales is a formalisation of the idea of Filip (1993/99) and Altshuler and Filip (2013) that culminations like *arrive* are characterised by *both* an instantaneous *and* a gradual change, and different linguistic phenomena are sensitive to one or the other. Thus, while culmination predicates are true of instantaneous events and are thus compatible with, e.g., time point adverbials and are not admissible as complements of aspectual verbs, they can appear to behave like accomplishments in the scope of an operator having the ability to access their secondary scales — and as I will argue in Section 5.3, the progressive is such an operator.

So just as primary scales of change are the set of all scales of change associated with events in the extension of a predicate, let us now define the secondary scales of change of culminations as the set of all scales of change associated with cover events of events in their extension:

Definition 17 (Secondary scales of change (of culminations))

The set of secondary scales of change SeSoC($^{\phi}\varphi$) associated with an event property $^{\phi}\varphi$ contains the set of scales of change of the cover events of events in the extension of φ at all possible worlds.

Formally, if $\uparrow \varphi$ is an eventuality property (i.e., $\llbracket \uparrow \varphi \rrbracket^{M,w,g,c,\gamma} \colon W \to \varphi(E)$), then $\operatorname{SeSoC}(\uparrow \varphi) \stackrel{\text{def}}{=} \{ \sigma_{c,w}(\gamma, e') \mid \exists e \in \llbracket \varphi \rrbracket^{M,w,g,c,\gamma} \colon e' \text{ is a cover event of } e \text{ at } \gamma \in G_{\mathcal{T}}, c \in C, \text{ and } w \in W \}.$

Note that the way Definition 17 is formulated, secondary scales are currently only defined for culminations, and serve no other purpose. However, I leave open the possibility that the definition could be modified and extended to other cases. For instance, it might be possible to capture the temporariness of a stage-level state like *John be happy* (or, in general, of any 1-state predicate, in the terminology of Klein 1994) by assuming that its secondary scales contain multi-valued scales of change describing John's level of happiness or sadness throughout his life — whether this is a viable route to take is a subject for future research.

The idea that a predicate can be associated with two kinds of scales is a radical departure from all former scale-based approaches. Scale-based analyses of both adjectival and verbal phenomena so far have assumed that each predicate is associated with a single scale and the properties of this single scale determine the properties of the predicate. The present proposal opens up a wide topic of how two scales can *interact* with each other — which has not received much attention in scale-based works, as yet.

Intuitively, while a primary scale of change captures the internal change associated with a predicate, a secondary scale of change captures a *relational* component of the predicate. Thus, PrSoC and SeSoC should not be seen as components of a predicate in the sense of lexical decompositional approaches (e.g., Dowty 1979, Pustejovsky 1991, Levin and Rappaport-Hovav 2005, Ramchand 2008), i.e., they do not refer to different parts of the event structure associated with the predicate. Instead, they are different aspects of a predicate's meaning: the primary scale capturing inherent, the secondary scale capturing relational properties thereof (put differently, PrSoC and SeSoC can be seen as capturing the asserted and presupposed contents, respectively, in the two-layered framework of Heyde-Zybatow 2008 and Malink 2008).

As an analogy, take *relational nouns* like *mother* which "describe objects as standing in a certain relation to others" (Löbner 1985, p. 292).⁴ These are generally analysed as denoting two-place relations (Löbner 1985, de Bruin and Scha 1988, Partee and Borschev 2003, Barker 2012, Partee and Borschev 2012). But as noted already by Löbner (1985), relational nouns can also be used as *sortal nouns*, referring to individuals rather than relations (as in *She is a good mother*) if their additional argument is existentially bound.⁵ Formally, if relational nouns denote two-place relations (e.g., $\lambda y \lambda x$. mother_of(y, x)), then by applying to them the *detransitivization type-shifter* Ex = $\lambda R \lambda x. \exists y R(y, x)$ of Barker (2012), we get a sortal noun (a noun denoting a set of individuals, e.g., $\lambda x. \exists y$ mother_of(y, x)).

Culmination predicates can be seen in analogy to *detransitivized* relational nouns: they denote events of instantaneous change which are related to a separate event of gradual change. A substantial difference between culminations and relational nouns is that the relation in the case of culminations is much weaker: as argued in Section 3.2, not all events falling under a culmination predicate need be related to an event of gradual change. (In contrast, all mothers have to have a child.) This is why the complex notions of a cover event and secondary scales of change are needed: they serve to characterise the complex relation culminations stand to events of gradual change.

As regards the scales of change associated with different predicate types, the following is now the case:

⁴Löbner (1985) actually distinguishes *functional nouns* like *mother* from *relational nouns* like *daughter* (the mother of an individual is necessarily unique, in contrast to a daughter). However, in later works, the more common practice is to assume that "the *basic* types of nouns are sortal nouns, proper names and relational nouns, and for the most part, functional nouns are simply an accidental subclass of the relational nouns" (Partee and Borschev 2012, p. 445f).

⁵That relational nouns can refer to individuals rather than relations in a context might not even be simply a *possibility* for them, at least in German: in an extensive corpus study, Horn (in preparation) found that relational nouns in German are used as non-relational nouns in more than two-thirds of the cases (in a corpus of diverse kinds of texts including, among others, newspaper articles and narratives).

- Accomplishments (like *drink a bottle of wine*) are associated with multi-valued scales of change (at their normal granularities).
- Happenings (like explode and flash) are associated with binary scales of change (at their normal granularities).
- Culminations (like *arrive* and *die*) are associated with both binary and multivalued scales, where the multi-valued scales belong to their secondary scales of change.

Thus, we now have enough tools in our framework to make the above three-way distinction among telic predicates that we needed to set apart based on their different interpretations in the progressive.

5.1.3.3 Scales of satisfaction

Before presenting a scale-based characterisation of different aspectual classes, I introduce two further auxiliary notions. These will not be relevant for my analysis of progressive achievements, but are nonetheless useful in distinguishing different aspectual classes — in particular, *telic* and *atelic* predicates, for the distinction of which we have to make reference to the satisfaction conditions of a predicate.

For each predicate and scale of change, we can define a final subscale of a scale of change, such that if any point of this subscale is reached, the relevant eventuality predicate is satisfied. In the case of *run to the bank*, this part of the scale would be a single point, namely, the state of being at the bank, since any running event with an endpoint on a path before or beyond it would not qualify as running to the bank. (Running to a point beyond the bank would be running to *that* point, rather than to the bank.) This — possibly degenerate — final subscale of the scale of change will be called the *scale of satisfaction* for that particular eventuality predicate. The complement of this subscale — that is, the complete initial part of the scale of change at which the predicate is not satisfied, is called the scale of *pre-satisfaction*. In other words, any primary scale of satisfaction associated with a predicate can be partitioned into two blocks such that the boundary of the blocks is the point where the predicate gets satisfied: Figure 5.3 provides an illustration.

Formally, a scale of satisfaction is defined by collecting all the degrees which are target states of some event in the extension of the relevant predicate: these are the stages of the change where the relevant predicate is satisfied:


Figure 5.3: Illustration of the scales of satisfaction and pre-satisfaction

Definition 18 (Scale of satisfaction)

The scale of satisfaction $\operatorname{Sat}({}^{\diamond}\varphi)(S)$ for eventuality property ${}^{\diamond}\varphi$ and a scale of change $S \in \operatorname{PrSoC}({}^{\diamond}\varphi)$ is that subscale of S which contains all and only those points $p \in \operatorname{carrier}(S)$ such that there is a $c \in C, w \in W$ and $\gamma \in G_{\mathcal{T}}$ for which there is an $e \in \llbracket \varphi \rrbracket^{M,g,w,c,\gamma}$ such that $\sigma_{c,w}(\gamma, e) \subseteq_{\operatorname{ini}} S$ and $\max(\sigma_{c,w}(\gamma, e)) = p$ (where $\subseteq_{\operatorname{ini}}$ is the initial subscale defined in Definition 7 and max is the maximum).

Note that the scale of satisfaction is not the image of the temporal trace of an eventuality in the extension of the predicate: instead, it is the image of the set of all *possible endpoints* for the eventuality predicate in a given situation.

Then we can define the scale of pre-satisfaction as the complement of the scale of satisfaction:

Definition 19 (Scale of pre-satisfaction)

The scale of pre-satisfaction $\operatorname{Presat}(^{\varphi})(S)$ for an eventuality property $^{\varphi}$ and a scale of change $S \in \operatorname{PrSoC}(^{\varphi})$ is the complement of $\operatorname{Sat}(^{\varphi})(S)$ in S.

5.2 Scale structures by aspectual class

5.2.1 Primary scale structures

In a scale-based framework, the difference between aspectual classes is modeled with reference to their scale structure. I will follow Beavers (2008, 2013, cf. Section 2.3.2.4) in maintaining that the aspectual properties of predicates depend on two scalar properties:

- *i) The specificity* of the degree that needs to be reached to satisfy the predicate determines telicity.
- *ii)* The mereological complexity of the scale determines durativity.

As an account of telicity is orthogonal to the topic of the present dissertation, I do not go into the complexities of choosing one account over another. I here simply follow Hay et al. (1999) and assume that a telic predicate φ specifies a (non-zero) lower bound on the amount of change; in other words, in each situation, a specific non(-zero) degree needs to be minimally attained on the scale for φ to be satisfied. This idea can be translated in the present framework using the notion of a scale of satisfaction:

Definition 20 (Telicity based on scale structure)

An eventuality predicate φ is telic iff for all w, g, c, γ , in all $S \in PrSoC(^{\varphi})$, there is a specific, non-minimal degree $d \neq \min(S)$ such that d is the minimal element of the scale of satisfaction, that is, $d = \min(Sat(^{\varphi})(S))$.

As for durativity, as already noted, I follow Beavers (2002, 2008, 2013), and assume that binary scales correspond to non-gradual change, and are associated with nondurative predicates, while multi-valued scales correspond to gradual change, and are associated with durative predicates. I extend Beavers's account of mereological complexity in the way intended by Beavers (2002, §6) to statives. Table 5.2 summarizes the complexity of the scales I assume to be associated with different kinds of predicates.

States. Let us now inspect the general structure of the change associated with different aspectual classes. As for states like *be happy*, since they involve no change, I assume that the scale of change associated with them is necessarily a singleton, that is, it contains a single degree corresponding to a single state.⁶ Given the constraints de-

⁶Of course, the single degree on the scale of change can also be a degree on some orthogonal multivalued scale — a property/location scale — to which modifiers such as *almost* and spatial extent read-

No. of deg.	Change	Aspectual class
2<	Gradual change	Accomplishments, activities
2	Non-gradual change	Vendler's achievements
1	No change	1-state stative predicates
N/A	No potential for change	0-state predicates

Table 5.2: Scale complexity, properties of change and aspectual classes

tailed in Appendix B, the primary scales of change $PrSoC(^{\phi})$ associated with a state predicate φ are indeed all singleton, because all state individuals are associated with singleton scales. As regards the temporal extent of states, I endorse the most common conception that states hold at non-degenerate intervals and at any instant of the interval at which they hold.

Note that given the focus of the present dissertation on achievements, I disregard complexities pertaining to other classes, including states: I will not go into the question of how to differentiate non-temporary, *0-state predicates* (in the terminology of Klein 1994⁷) like *The pen is red* and temporary, *1-state predicates* like *The light is on*, or how to the differentiate *momentary* states like *be asleep* and *be a mammal* from *interval* state predicates like *sit* (in the terminology of Dowty 1979, Ch. 3). I here list some ways in which I believe the 0- vs. 1-state predicate distinction might be captured on the present framework:

- a) 1-state predicates are associated with singleton primary scales of change; 0-state predicates are associated with an empty scale of change (note that this requires lifting the nonemptyness requirement imposed on scales), i.e., their primary scale of change is the empty set Ø.
- b) Both 0-state and 1-state predicates are associated with singleton primary scales of change; but 1-state predicates are in addition associated with multi-valued secondary scales of change (cf. Section 5.1.3.2).

ings as in Section 5.1.2 have access. E.g., *be almost black* is associated with a scale of change consisting of the single degree of being of that particular colour, but that particular colour is also a degree on a multivalued scale of colours.

⁷The reason I use the terms 0- and 1-state predicates rather than "stage-level" and "individual-level" predicates (as in the literature building on Carlson 1977) or "temporary" and "non-temporary" states (as in Bach 1981) is that Klein's (1994) terminology is ideally suited to capture the essence of these predicates on the present framework: 1-state stative predicates are characterised by scales containing 1 degree, while 0-state predicates are not even associated with 1-degree scales, cf. Table 5.2.

- c) Only 1-state predicates are associated with a singleton primary scale of change; 0-state predicates are not associated with a primary scale of change.
- d) Only 1-state predicates are predicates of eventualities; 0-state predicates are predicates of intervals (and as such, are not associated with primary scales of change).

Here, I opt for the solution that 0-state predicates are predicates of intervals, for the reason that 0-state predicates appear to pattern widely differently from all other predicates (cf. e.g., Kratzer 1995, who also assumes on this basis that individual-level predicates, in contrast to other predicates including stage-level states, do not hold of eventualities). For instance, they do not appear to be amenable to modification with spatial or temporal adverbs (#John was a male at noon/in London), indicating that they do not denote sets of spatio-temporally located entities. In addition, a prime reason Davidson (1967) proposed the introduction of events into semantics (and note that he himself did not regard any state predicates as having an event argument) is to cope with the problems that the stacking of modifiers like *slowly*, *deliberately*, *with a knife*, etc. onto predicates poses. But 0-state predicates do not typically allow such modifiers, only sentential modifiers like *probably*, which arguably operate on propositions, not event predicate denotations. However, a careful consideration of which, if any, of the above solutions to distinguishing 0- and 1-state predicates is suitable is left for further research. Unless indicated otherwise, by state, I refer to (state particulars, i.e., elements of E_{stat} , or) 1-state stative predicates.

Dynamic predicates. For dynamic predicates, the following considerations apply:

- given the constraints on dynamic events detailed in Appendix B, accomplishments and activities are associated with multi-valued primary scales of change, and Vendler's achievements with a binary one (cf. Section 5.1.2);
- accomplishments and achievements are telic, that is, there is a specific non-zero point at which their scale of satisfaction starts, whereas activities are satisfied by any change.

Based on this, the generic scale structure of the primary scale of change for predicates belonging to different aspectual classes is presented in Figure 5.4.⁸

⁸A generic scale structure is an abstraction over all scales of change associated with a predicate. The scale structure for accomplishments is a generalization over regular accomplishments and those of the "quantization puzzle" (cf. Filip 2000) such as *eat at least 3 sandwiches*. Note that the approach of Hay et al. (1999) does not share the shortcoming of most scale-based approaches to telicity, such as the closed-scale approach of Kearns (2007) and Kennedy and Levin (2008), which make wrong predictions



Figure 5.4: The primary scale structures for different aspectual classes. The scale of satisfaction is encoded in green, while the scale of pre-satisfaction is encoded in red. Endpoints that are included in an interval are displayed with a full bullet, while those excluded are indicated with an empty bullet.

5.2.2 Differentiating types of Vendler's achievements

As detailed in Section 5.1.3.2, the way the culmination/happening distinction can be made is by reference to *secondary scales of change*: culminations, but not happenings are associated with multi-valued secondary scales of change. This forms the basis for predicting a different interpretation for the combination of different modifiers and operators with happenings, culminations and accomplishments.

As for the question of how semelfactives and non-semelfactive happenings are differentiated, I argue that their difference does not lie in *structural* properties of their associated scales of change. Instead, it depends on the specification of the temporal duration of the states described by the two degrees on the scale of change of the respective predicate types. For non-semelfactive happenings like *recognise* and *explode*, the states corresponding to the minimum (not being aware of someone, and being whole, respectively) and the maximum (being aware of someone, and not being whole, respectively) are specified to hold at a *non-degenerate interval*, i.e., to take some minimally specified time. For semelfactives like *The light* $flash_{\emptyset}$, the states corresponding to the minimum (the light not being illuminated) and the maximum (the light being illumi-

about the predicates of the quantization puzzle, just as Krifka's (1989, 1992, 1998) quantization and homomorphism-based account (cf. Filip 2011b). As an example, given an event during which John ate 6 sandwiches, the scale of satisfaction for *eat at least 3 sandwiches* contains all degrees equal to or higher than the state of John having eaten the third sandwich.

nated) are *not* specified to hold for more than an instant (at the particular granularity). This *lack* of a constraint on the duration of either their pre-state or their post-state is what makes semelfactives "full-cycle" predicates (in the terminology of Talmy 1985, 1988) and enables them to be iterated easily. Of course, nothing then precludes these states from holding at a longer interval, which is the most general case for their prestate (usually, a light is assumed not to be illuminated for some period before flashing), but it is also possible for their result state (the light being illuminated), as the following example shows:

(5.4) Then, it flashed and remained illuminated.⁹

The same observation was made by Beavers (2013), whose example was *I* hit the car (once), but my hand got stuck so *I* never pulled it away.

5.2.3 Verbs and verb phrases

Based on the foregoing, the following summarizes the kind of scales that are associated with English non-stative *verbs* (whose the asserted content may be represented in any way, such as through a decompositional analysis building on Dowty 1979, or in a Neo-Davidsonian framework as in Parsons 1990, etc.) at their natural granularity levels:

- Activities (e.g., *walk*): these are associated with multi-valued scales, because the events in their extension are extended, and so the scales of change associated with these events are necessarily multi-valued in all contexts.
- Accomplishments (a marginal class, e.g., recover): these are also associated with multi-valued scales (but with a specific degree at which their scale of satisfaction starts), because the events in their extension are extended. The result state of these verbs holds for an extended period of time.
- Telic scalar/incremental verbs (e.g., cross): these are associated with binary and multi-valued scales alike, as events in their extension include both instantaneous events (e.g., events in the extension of *John crossed the border at noon*) and extended events (e.g., those in the extension of *The army crossed the border in 2 hours*). The result state of these verbs holds for an extended period of time.

⁹http://www.ptcruiserlinks.com/forum/general-pt-cruiser-discussions/ 27896-mil-check-engine-light-without-dtc.html

- Culminations (e.g., arrive): these are associated with binary scales of change, as events in their extension are non-extended and hence associated with binary scales in all contexts; but they are also associated with multi-valued secondary scales of change. The result state of these verbs holds for an extended period of time.
- Non-semelfactive happenings (e.g., explode): these are associated with binary scales of change, as events in their extension are non-extended and hence associated with binary scales in all contexts. The result state of these verbs holds for an extended period of time.
- Semelfactives (e.g., *flash*): these are associated with binary scales of change, as events in their extension are non-extended and hence associated with binary scales in all contexts.

For instance, the primary scales of change of the culmination *arrive* include the following, where γ is any natural granularity function for *arrive* (for brevity, I omit the model *M* and the variable assignment *g* parameters):

$$\begin{split} \mathsf{PrSoC}_{\gamma}(^{arrive}) &= \{ \langle \cup \{\llbracket \mathsf{not_at}(j,z) \rrbracket^{w,\gamma,c} \mid w \in W \}, \cup \{\llbracket \mathsf{at}(j,z) \rrbracket^{w,\gamma,c} \mid w \in W \} \rangle, \\ \langle \cup \{\llbracket \mathsf{not_at}(j,z) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{at}(j,z) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ \langle \cup \{\llbracket \mathsf{not_at}(m,x) \rrbracket^{w,\gamma,c} \mid w \in W \}, \cup \{\llbracket \mathsf{at}(m,x) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ \langle \cup \{\llbracket \mathsf{not_at}(m,x) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{at}(m,x) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ \end{split}$$

At the verb phrase level, at which the agent (and often the goal) of *arrive* are specified, many of these scales are eliminated, leaving only those scales which are associated with events in the extension of the *verb phrase*, e.g.,

$$\begin{aligned} \mathsf{PrSoC}_{\gamma}(^{J}Ohn\ arrive_{\emptyset}\ at\ the\ station) &= \\ & \{ \langle \cup \{\llbracket \mathsf{not_at}(j,\mathsf{station}) \rrbracket^{w,\gamma,c} \mid w \in W \}, \cup \{\llbracket \mathsf{at}(j,\mathsf{station}) \rrbracket^{w,\gamma,c} \mid w \in W \} \rangle, \\ & \langle \cup \{\llbracket \mathsf{not_at}(j,\mathsf{station}) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{at}(j,\mathsf{station}) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \ldots \}. \end{aligned}$$

The secondary scales of change of *arrive*, in turn, include the following (where z_i, z_j are different from z and are on a path extending to z, and likewise for x_i, x_j and x):

$$\mathsf{SeSoC}_{\gamma}(^{arrive}) = \{ \langle \cup \{ \llbracket \mathsf{at}(j, z_i) \rrbracket^{w, \gamma, c} \mid w \in W \}, \cup \{ \llbracket \mathsf{at}(j, z_j) \rrbracket^{w, \gamma, c} \mid w \in W \}, \\ \dots, \cup \{ \llbracket \mathsf{at}(j, z) \rrbracket^{w, \gamma, c} \mid w \in W \} \rangle, \langle \cup \{ \llbracket \mathsf{at}(m, x_i) \rrbracket^{w, \gamma, c} \mid w \in W \},$$

$$\cup \{ \llbracket \mathsf{at}(m, x_j) \rrbracket^{w, \gamma, c} \mid w \in W \}, \dots, \cup \{ \llbracket \mathsf{at}(m, x) \rrbracket^{w, \gamma, c} \mid w \in W \} \rangle \}.$$

Again, at the verb phrase level, only those scales are kept which are associated with events in the extension of the verb phrase.

As for the happening *explode*, it is only associated with primary scales of change, namely, those binary scales of change that are associated with events in its extension:

$$\begin{split} \mathsf{PrSoC}_{\gamma}(^{explode}) &= \{ \langle \cup \{\llbracket \mathsf{whole}(x) \rrbracket^{w,\gamma,c} \mid w \in W \}, \\ &\cup \{\llbracket \mathsf{disintegrated}(x) \rrbracket^{w,\gamma,c} \mid w \in W \} \rangle, \\ \langle \cup \{\llbracket \mathsf{whole}(x) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{disintegrated}(x) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ \langle \cup \{\llbracket \mathsf{whole}(y) \rrbracket^{w,\gamma,c} \mid w \in W \}, \cup \{\llbracket \mathsf{disintegrated}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ \langle \cup \{\llbracket \mathsf{whole}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{disintegrated}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ \langle \cup \{\llbracket \mathsf{whole}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{disintegrated}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \ldots \}. \end{split}$$

The scales of change of a semelfactive like *flash* have the same structure as those of non-semelfactive happenings, e.g.,

$$\begin{aligned} \mathsf{PrSoC}_{\gamma}(^{flash}) &= \{ \langle \cup \{\llbracket \mathsf{dark}(x) \rrbracket^{w,\gamma,c} \mid w \in W \}, \cup \{\llbracket \mathsf{illuminated}(x) \rrbracket^{w,\gamma,c} \mid w \in W \} \rangle, \\ & \langle \cup \{\llbracket \mathsf{dark}(x) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{illuminated}(x) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ & \langle \cup \{\llbracket \mathsf{dark}(y) \rrbracket^{w,\gamma,c} \mid w \in W \}, \cup \{\llbracket \mathsf{illuminated}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ & \langle \cup \{\llbracket \mathsf{dark}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{illuminated}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ & \langle \cup \{\llbracket \mathsf{dark}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \}, \cup \{\llbracket \mathsf{illuminated}(y) \rrbracket^{w,\gamma,c'} \mid w \in W \} \rangle, \\ \end{aligned}$$

But the pre- and result states (represented by the first and second degree on each scale, respectively) of semelfactives are not specified to hold for more than an instant, as opposed to those of *explode*-like happenings.

It is again to be emphasized that these predicate-level scales of change are inherited from events, and so the *extension* of a predicate at different possible worlds is what ultimately determines the scales associated with it. As additional information is added (complements, modifiers etc.), the extension shrinks (e.g., the extension of *John arrive*_{\emptyset} *at the station* is a subset of the extension of *arrive*), and along with it, the set of associated scales of change also shrinks (because the scales of change associated only with events that are not in the extension of the more complex predicate are not scales of the complex predicate).

5.3 The progressive

5.3.1 Integrating a definition of the progressive into the scale-based framework

The amount of existing work on the semantics of the progressive is immense and intricate, ranging from temporal viewpoint accounts (such as Bennett and Partee 1972, as well as Klein 1994) through modal approaches (such as Dowty 1979, Landman 1992, Asher 1992, Portner 1998, Bonomi 1999) and accounts which rely on the progressive as a primitive (such as Parsons 1990, Gendler Szabó 2004, 2008) to analyses relying on informational, indicative links (Varasdi 2010, 2014). I will not offer a comprehensive semantics of the progressive here aiming to address all the intricate issues which have formed the centre of debate for the past decades, but will instead focus on its viewpoint component, and argue that it should be about *change* (i.e., something qualitative), rather than *runtime* (i.e., something quantitative) — as opposed to, for instance, temporal adverbials like *at noon* or *in an hour* which *are* about runtime. As such, the analysis offered below is a simplification, aiming only to capture one aspect of the relevant operators, but I will hope to show the at least *prima facie* tenability and potential advantages of a scale-based approach.

A further limitation to keep in mind is that the present discussion, just as the thesis itself, is geared toward English. At the same time, I subscribe to the view that the basic semantics of aspectual operators are similar across languages (or at least, across some set of languages), with language-specific variations captured by parametric differences — in the fashion of Altshuler (2014), who presented a typology of partitive aspectual operators parametrizing language-specific differences through a handful of criteria. Section 5.6 will therefore be devoted to a brief look at how the present account of English aspectual operators could be extended to aspect in other languages.

As for English, I here propose to conceive of the progressive as expressing that a change of a particular kind is going on, and there is still a potential to satisfy the corresponding predicate at a later stage. In terms of the scale of change, this means having the possibility to continue to higher degrees to become an eventuality falling under the relevant predicate. There are a number of ways in which to make this informal idea explicit, the most straightforward solution being to embed the scale-based idea into an existing progressive theory in lieu of temporal or event mereological relations. To this end, I will briefly (and without any exhaustivity) discuss some considerations for selecting a theory from the available theories of the progressive.

The scale of change-based approach naturally lends itself to a Landman-style characterisation, for instance. The degrees on the scale of change can be seen as corresponding to the boundaries of *stages* in the account of Landman (1992). We may say, building on Landman (1992) (but without using his sophisticated machinery involving continuation branches, for simplicity), that $[PROG(^P)(e)]^{M,w_0,g,c,\gamma} = 1$ iff there is a *P*-event e' occurring in a world w which is a reasonable option for e in the world w_0 of evaluation, whose scale of change is a proper superset of that of e, that is, $\sigma_{w_0,c}(\gamma, g(e)) \subset$ $\sigma_{w,c}(\gamma, e') \wedge e' \in [P]^{M,w,g,c,\gamma}$. The problem with such an existential quantification-based characterisation is that it is known to be vulnerable to cases where only a disjunctive progressive statement is true, such as the coin-flipping scenario that prompted Dowty to modify his possibility-based account (Dowty 1977) to a necessity-based one (Dowty 1979), or the multiple-choice paradoxes described by Bonomi (1999).

In particular, the proposed possibility-based scalar account of the progressive will result in *false positives*. For example, to take the coin-flipping scenario, the proposed definition, just like that of Dowty (1977), would incorrectly predict that all of (5.5a)–(5.5c) is true when a coin has just been flipped (because the change going on is compatible with all three predicates), even though speakers would only regard (5.5c) as true:¹⁰

- (5.5) a. The coin is coming up heads.
 - b. The coin is coming up tails.
 - c. The coin is coming up heads or tails.

One way to overcome this problem is to change existential quantification over reasonable worlds to universal quantification over continuation *trees*, as suggested in Gendler Szabó (2004). Another option is to go with the approach in Dowty (1979) who employs universal quantification over *inertia worlds* (worlds in which the course of events develops inertially), which is the choice of Deo (2009), for instance.

However, there have been several objections raised against the notion of inertia worlds from as early on as Vlach (1981), who argues that in a scenario when Max is hit by a bus while crossing a street, close to the moment when Max is hit, in no

¹⁰Landman's (1992) account predicts that exactly one of (5.5a)–(5.5b) is true, which is also an incorrect prediction. The reason the proposed Landman-style scale-based definition patterns with that of Dowty (1977) instead of Landman (1992) is that it does not refer to a single continuation branch constructed based on similarity between worlds but uses existential quantification over reasonable worlds instead. The definition could be made more faithful to Landman (1992) by restricting existential quantification to reasonable worlds on the continuation branch, but that would not eliminate the problem of a false positive.

inertia worlds does he get across the street, which would incorrectly predict that at that moment, *Max was crossing the street* is false. Indeed, similar problems might apply to the notion of a "reasonable option" in Landman (1992). Take the sentence

(5.6) Samantha was crossing the minefield.

from Asher (1992), which he mentions as a potential counterexample to his own progressive theory referring to "normal" worlds. As Asher notes, people are not normally able to cross minefields, so we can assume that successfully crossing the minefield is also not a *reasonable option* for Samantha, so Landman (1992) (whether employing existential or universal quantification), like Asher (1992), would incorrectly predict (5.6) to be false — a *false negative* due to the notion of reasonability.

So given these very brief considerations, I here opt for an account of the progressive based on Varasdi (2014), who employs quantification over event predicates, instead of possible worlds. Informally, and with some simplification, on his account, PROG(P)(e) holds iff, out of a contextually determined set of disjoint predicates including P, it is only P-events into which e can develop. In this case, the observed characteristics of e are *indicative* of P with respect to the contextually given set of alternative predicates.

5.3.2 The components of the progressive

I now turn to adapting Varasdi's (2014) to my scale of change-based framework. We need a definition according to which a progressive is true whenever a change of the relevant kind (and only the relevant kind out of a set of alternatives) is happening, and there is still a possibility for it to continue. What we aim to capture with a scale- and alternatives-based definition of the progressive is that the way we categorize ongoing eventualities in terms of complete types is by matching the *change schemata* of the types to the actual change and select the one that fits out of a set of alternatives. Thus, in what follows, let us collect the components needed for an adequate scale-based definition of when $PROG(^{\phi})(e)$ is true.

A non-final subscale component. The most crucial part of a scale-based definition of the progressive has to take the place of those components of event-mereological definitions of the progressive that refer to a part-of relation between events. The goal of the whole scale-based framework is to enable us to have the progressive refer to

a part-of relation over scales, instead. So instead of a (non-final) part-of relation over events ($e \ \Box_E e'$), our progressive definition will have to include a (non-final) partof relation over scales of change. To this end, the non-final subscale relation defined in Definition 6 will be used ($S_i \ \subseteq_{nf} S_j$). The rationale behind the non-final subscale requirement is based on the intuitive view of the progressive as referring to the interior part of an event (cf. Section 2.1.1). But by using scales, instead of events, we can ensure that culmination predicates, which are associated with multi-valued secondary scales, are admissible in the progressive. Thus, the superscale (S_j) of the non-final subscale relation has to be *either* a primary scale of change associated with the input predicate, or a secondary scale of change associated with it. As for the scale that we require to be its subscale, it must be associated with the eventuality of which the progressive is predicated: it is the ongoing event that has to realize a part of the change associated with the predicate. Thus, a non-final subscale component of the progressive definition will be $\sigma_{w,c}(\gamma, e) \subset_{nf} S$ for some $S \in (\PrSoC(^{\phi}) \cup SeSoC(^{\phi}))$.

Such a non-final scale requirement directly ensures that the culmination point is not part of the event that makes the progressive true. And it also ensures that the predicate φ need not get realized in the actual world: because the primary and secondary scales associated with a predicate contain the scales of change of all eventualities the predicate applies to at *all* possible worlds, the scale of change whose non-final subscale is realized by an actual event may never be realized in the world of evaluation. The scale of change in question might be associated with an event that is only realized at a different possible world.

The non-final subscale requirement also precludes the progressive from applying to a state predicate *P*, which is associated with singleton primary scales of change (cf. Section 5.2). Because there are no degrees below a single degree, the scale of change of *e* in the definition cannot be a non-final subscale of any of the scales of *P*. So states cannot satisfy the non-final subscale requirement of the progressive. Note that it is well known (cf. Dowty 1979, Bach 1981) that states (more precisely, 1-state predicates) *can* sometimes appear in the progressive, in which case they take on a transitoriness meaning component (cf. the contrast between Bach's 1981 pair of examples, *I live in Massachusetts* and *I'm living in California*: the latter, but not the former implies that the place of residence in question is temporary). The way this combinability with the progressive can be modelled in the present framework is to have 1-state predicates be associated with activity-like secondary scales of change, as proposed in Section 5.1.3.2, and then the progressive could be made true by such a *multi-valued secondary scale of*

change. Given the non-final subscale requirement, this would imply that the state can change in the future, which may be responsible for the implication of transitoriness.

A contrastive component. At this point, however, the requirement on the progressive is too weak, because it involves existential quantification over scales (just as we have seen in the initial reasonable options-based definition in Section 5.3.1). We need a way to exclude false positives in multiple-choice scenarios like the coin-flipping case mentioned in Section 5.3.1. This is where the account of Varasdi (2014) can be directly put to use: we need to refer to a contextually determined set Θ_c of alternative predicates that capture the outcome possibilities relevant in the given context. Then, just as Varasdi (2014), we can require that exactly one of these alternatives (namely, the predicate that is the argument of the progressive) satisfy the requirements of the progressive — in our case so far, that the scale of an actualized event be a non-final part of a primary or secondary scale of change of the predicate. Now, when a coin is flipped, the contextually relevant predicates are *the coin come*⁰ *up heads* and *the coin come*⁰ *up tails*, and the change going on is characteristic of both elements of this set of alternatives to refer to a set Θ_c of alternatives in the definition of the progressive, see Varasdi (2014).

Complexity requirement. The components of the scale-based definition discussed so far still overgenerate true progressives. In particular, *e* can be a state associated with a singleton scale of change, and then nothing precludes it from satisfying the non-final subscale and potentially the contrastive components of the definition for a happening predicate φ (e.g., *notice*). In particular, if the scale of *e* is a singleton, it can be identical to the initial degree of a binary scale of φ , and so a false positive would arise: e.g., *Mary is noticing John* would be incorrectly predicted to be felicitous at an everyday granularity.

What we need is to ensure that a *change* of the relevant kind is going on. A state cannot be *indicative* of an achievement (or an activity or an accomplishment, for that matter) in the terminology of Varasdi (2014); or in the terminology of Landman (1992), it cannot be a *stage* of one. Landman (1992) refers to "stages" instead of simply parts in his definition of the progressive for precisely this reason: while a state may be part of an event, it cannot be a stage of it, because a stage has to be "developed" enough to be recognizable as a stage of some event. Using the idea put forth in Section 5.3.1 that the boundaries of stages correspond to the degrees on a scale, we can translate Landman's stage-based requirement into a requirement on scale complexity: the scale

of the ongoing event *e* has to be developed enough to be of the *same complexity* as the scale of the predicate whose non-final part it is. Thus, an activity or an accomplishment can only be in progress if an extended change is happening, because the ongoing event *e* has to be associated with a multi-valued scale which in turns entails that *e* has to be extended.

In contrast, take an achievement predicate Q, and take its primary scales of change. $PrSoC(^Q)$ for such a predicate consists of binary scales, i.e., scales of the structure $\langle d_1, d_2 \rangle$. Then, given the non-final subscale requirement, the progressive of φ could only be true of e if the scale of change of the eventuality e is a *singleton*, consisting of d_1 . But then the complexity requirement is violated, because $\sigma_{w,c}(\gamma, e)$ is not a binary scale. Unless the achievement Q is associated with a multi-valued secondary scale — which is true for culminations, but not happenings (cf. Sections 5.1.3.2 and 5.2.2) — the progressive of Q can never be true.

If a progressive can never be true for a predicate, then it is reasonable to exclude the application of this operator to that predicate. Thus, the applicability condition of a multi-valued input scale is just as much a reasonable criterion for the scale-based definition of the progressive as the durative input requirement for temporal or event mereological definitions of the progressive. So I assume the following principle:

Principle 3 (Applicability condition of the progressive)

Do not apply PROG to φ if φ is not associated with mostly (primary or secondary) multivalued scales of change.

The output of the progressive. What remains now is to take a stand on what kind of predicate the output of the progressive is, although this question is actually orthogonal to our focus, because a scale-based definition could be formulated with respect to any view we take about the output of the progressive. As our characterisation stands so far, the output of the progressive is a *process*, in line with the view of Mourelatos (1978): it is true of extended events and it is not telic, as it holds of all extended parts of an event that it holds of. If we want our definition to adhere to the *stativizer* view of the progressive as in Vlach (1981) and Moens and Steedman (1988), we can have the progressive apply to a state *s* that is a mereological part of the event *e* we have been discussing so far. Alternatively, the progressive PROG($^{\phi}$) could also be a predicate of *time intervals*, as in Dowty (1979), Portner (1998) or Deo (2009) in the way that

 $PROG(^{\varphi})$ is true of an *i* such that $i = \tau(\gamma, e)$ for the event *e* we have been discussing so far.

A careful consideration of the behaviour of the progressive must ultimately decide between the approaches outlined in the previous paragraph. Just as I did not go into a detailed survey of which former progressive account to embed in my scale-based framework and decided on the basis of a handful of rough preliminary considerations, I here do not consider detailed arguments for or against one or the other view of the progressive. One reason to chose the stativizer view is that as remarked by, for instance, Dowty (1986), statives and progressives tend to behave similarly in terms of their discourse properties (but Dowty remarks that progressives differ from states in not typically allowing an inceptive reading¹¹). Note that the parallel between the behaviour of statives and the progressive holds only if we restrict our attention to temporary states, but not when we take into account individual-level states like John is a male. But a progressive is clearly not an individual-level property (or put more generally, it is not what Klein 1994 calls a 0-state predicate), which is shown by its combinability with a time point adverbial: John was walking at noon is just as felicitous as John was happy at noon (a stage-level/1-state predicate), in contrast to #John was a male at noon (an individual-level/0-state predicate). Thus, on a stativizer view of the progressive, the output of the progressive operator must be a 1-state predicate. And given a commitment to 0-state predicates being predicates of time intervals (cf. Section 5.2.1), this is borne out, as the progressive outputs a predicate of state tokens, and so can only be a 1-state predicate.

No perfect progressive. A more interesting case can be made for the stativizer view of the progressive based on the perfect progressive that is also pertinent to the focus of the dissertation. It has been noted in previous works (cf. e.g., Mittwoch 1991, Rothstein 2004) that even achievements which may combine with the progressive are generally infelicitous (cf. (5.7b)) or bizarre and ironic (cf. (5.7c)) in the perfect(ive) progressive, in contrast to accomplishments (cf. (5.7a)), as the following examples from Rothstein (2004, p. 44) illustrate.

- (5.7) a. She has been cooking dinner (for half an hour).
 - b. #Fred and Susan have been leaving.
 - c. ?Fred and Susan have been leaving for an hour.

¹¹Though it seems to me that, for instance, *Suddenly, he was singing* most naturally has an inceptive reading.

As noted by several authors (e.g., Comrie 1976, Dowty 1979, Kamp and Reyle 1993, Portner 2003, de Swart 2007), while the perfect of telic predicates describes their result state, the perfect of statives such as *John has lived in New York for three years* has a continuative or persistent reading (beside an experiential perfect reading) on which the relevant state is understood to be still ongoing. If we take the view of the progressive as a stativizer (i.e., it outputs states), then what is predicted is a *continuative interpretation* when the input to the perfect is a progressive. (And conversely, we predict that the progressive cannot have a perfect in its scope if we also take a stativizer view of the perfect.) This prediction is borne out, as illustrated by a perfect progressive like *John has been writing the letter for 3 hours* which has a continuative, rather than a result interpretation.

Now, since a progressive achievement is generally synonymous with an *about to* construction and so takes very little time (see, e.g., Kearns 1991, Rothstein 2004), it is generally odd to assert that it was going on for the relatively long time that a reference interval in everyday scenarios entails. Thus, the ironic nature of (5.7c) arises from the expectation that Susan and Fred's state of leaving an hour before was to be over quite soon thereafter, and so they cannot seriously have been leaving at that time.

So I propose that the oddness of achievements in the perfect progressive is rooted in a similar non-compatibility of interval sizes as is the reduced combinability of nonsemelfactives with the iterative reading of the progressive (cf. Section 2.1.2). But just as there are non-semelfactive iteratives, there are also non-ironic perfect progressive achievements:

- (5.8) a. ... the world has been winning the race between population growth and food production¹²
 - b. Like a man who has been dying for many days, a man in your city is numb to the stench.¹³

Importantly — and in line with the present proposal — the achievements that can appear in such a (non-iterative) perfect progressive construction are typically those in whose case the imminency meaning component is often *absent*: *win* and *die* are well known to be entirely acceptable in situations when the culmination (the actual death

¹²http://www.voanews.com/content/global-population-predictions-reportunited-nations-science/2454734.html

¹³https://www.goodreads.com/quotes/554034-like-a-man-who-has-been-dyingfor-many-days

and the culmination of the race) is not, yet, imminent. The example below illustrates such a situation:

(5.9) [H]e was one hundred and seventy days dying and not yet dead (from *The Stars My Destination* by Alfred Bester)

But for achievements like *arrive* in whose case a progressive like *John is arriving* is synonymous with *John is about to arrive*, we do observe the incompatibility between the imminency of the culmination and the natural length of a reference interval for a perfect progressive. As such, in their cases, a pluractional (iterative or habitual, cf. Bertinetto and Lenci 2012) reading of the perfect progressive can ensue only, as in (5.10).

(5.10) ... spring has been arriving weeks early for years¹⁴

Thus, if we take the stativizer view of the progressive, then a continuative reading of a perfect progressive is correctly predicted, and the incompatibility with progressive achievements implying imminency of the culmination can also be explained.

The definition of the progressive. Given the foregoing discussion, I propose the following definition of the progressive, according to which $PROG(^{\varphi})(s)$ holds iff the state denoted by *s* is a part of some event *e* such that the scale of change associated with event *e* is a *non-final part* of a primary or secondary scale of change of φ , and *only* φ out of the contextually given alternatives (defined as in Varasdi 2014), with the proviso that the complexity of the scale of *e* is the same as that of φ whose non-final part it is:¹⁵

Definition 21 (The progressive in a scale-based account) $[PROG(^{\phi})(s)]^{M,w,g,c,\gamma} = 1$ iff

1. (Progressive as a stativizer:) $g(s) \in E_{\text{stat}}$ and there is an eventuality e occurring in w such that $g(s) \sqsubseteq_E e$ (where \sqsubseteq_E is the non-strict part-of relation over eventualities) and

¹⁴http://www.theguardian.com/environment/blog/2014/mar/20/spring-equinoxgoogle-doodle-season

¹⁵The contrastive requirement will be slightly revised below in Section 5.3.3.

- 2. (Non-final subscale requirement:) there is a scale $S \in (PrSoC(^{\phi}) \cup SeSoC(^{\phi}))$ such that $\sigma_{w,c}(\gamma, e) \subset_{nf} S$ (where \subset_{nf} is the non-final subscale relation defined in Definition 6, and PrSoC and SeSoC are the primary and secondary scales of change of a predicate as defined in Definitions 15 and 17, respectively), and
- 3. (Complexity requirement:) $\sigma_{w,c}(\gamma, e)$ is of the same complexity as S (where the three levels of scale complexity are those listed in Table 3.1) and
- 4. (Contrastive requirement:) there is no $\psi \neq \varphi \in \Theta_c$ such that $\sigma_{w,c}(\gamma, e) \subset_{nf} S'$ for some $S' \in (PrSoC(^{\psi}) \cup SeSoC(^{\psi}))$, where Θ_c is a set of mutually disjoint predicates that includes φ , given by context c.

Figure 5.5 provides an informal graphic rendering of the scale-based definition of the progressive (disregarding the contrastive requirement).

$$\begin{array}{c} \mathsf{PrSoC}(^{\wedge}\varphi) \cup \mathsf{SeSoC}(^{\wedge}\varphi) \\ & \cup \\ & \mathcal{S} \\ & \vdots \\ \sigma_{w,c}(\gamma, e) \\ & \cup \\ & \cup \\ \sigma_{w,c}(\gamma, e) \\ & \cup \\ & \cup \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\ & | \\$$

Figure 5.5: Illustration: the progressive in a scale-based account

It is important to emphasize that the progressive does *not* operate on scales even in this framework. PROG operates on an eventuality property and outputs an eventuality predicate. It is in determining the *truth conditions* of a progressive sentence that we refer to scales.

Event-parts vs. scale-parts. Where Definition 21 differs from that of Varasdi (2014) is that it focuses on scale inclusion rather than event inclusion and thus offers a way to account for progressive culminations via reference to their multi-valued secondary scale. Indeed, culminations are the only case where the two progressive accounts differ. In the case of accomplishments, the $\sigma_{w,c}(\gamma, e) \subset_{nf} S$ part of the definition holds just in case *e* is a non-final part of some $e' \in \llbracket \varphi \rrbracket^{M,w',g,c,\gamma}$ for some possible world *w'*, that is, if it is a non-final part of some (possible) φ -event.

On Varasdi's (2014) account, the progressive of activities is made true by the eventuality occurring in the world of evaluation itself (corresponding to *e* in Definition 21), as he uses the *non-strict* part-of relation \sqsubseteq_E over eventualities. So activities also constitute a minor point of difference, but not in the truth conditions of the progressive, because what makes activities activities is that they can be potentially continued infinitely (cf. the characterisation of them by Beavers 2013 as involving a change to an *existentially bound* value on their scale of change), that is, there is always a possible event falling under the activity predicate with a more developed scale of change than the maximal one occurring in the world of evaluation. So whenever a progressive activity holds on Varasdi's definition, it holds on mine, and *vice versa*.

As for achievements, in Section 5.4 I will inspect in more detail how the theory proposed here fares. But first, I present some potential complications to Definition 21 of the progressive, one of which will prompt a slight revision in the contrastive requirement.

5.3.3 False positives and negatives and the role of the contrast set

No false positives. The question may arise whether the essentially disjunctive criterion referring to primary and secondary scales $(PrSoC(^{\phi}) \cup SeSoC(^{\phi}))$ in Definition 21 is not subject to the same problems as a disjunctive definition of the progressive referring to either the asserted or the presupposed content that I tentatively suggested in Section 2.2.2 as a solution for accounts of culminations as presupposing their activity part. However, the problem for that proposal was that there are all kinds of presupposed contents (an example was *manage to open the door*, which presupposes trying to open the door), and so allowing the progressive to access these contents heavily overgenerates the cases when a progressive sentence is predicted to be true. In contrast, secondary scales of change are specific to the actual change we want to associate with a predicate. Thus, false positives like *x is managing to open the door* while *x* is simply trying to open the door are excluded on the present disjunctive definition of the progressive.

However, there is a class of predicate which raises problems for the present account as well, namely those involving aspectual verbs like *stop* or *finish*. These present a problem for a theory of the progressive according to which the progressive can access the presupposed content of a predicate—because *stop* ϕ -*ing* presupposes the prior truth of ϕ -ing, and so, for instance, *John is stopping walking* is incorrectly predicted to be true during an event of John walking. These cases also present a problem for a coercion account like that of Rothstein (2004) — because, in an exactly parallel way to how Rothstein's analysis was shown in Section 2.2.4.2 to incorrectly predict *Mary is recognising John* to be true during Mary's walk toward John prior to her noticing him, *John is stopping walking* is also incorrectly predicted to be true at the time John is walking, because *stop walking* can be successfully coerced into a derived accomplishment in Rothstein's (2004) sense with its activity part being the process of John walking.

While the present account successfully excludes the progressive of happenings like *notice* or *recognise* from being true during some process preceding an event in their extension through the requirement on *cover events* to the effect that, informally, the different stages of the process have to be *elaborations* on the pre-state of the happenings (cf. Section 5.1.3.1 for details), the problem is that the cover event relation is formally satisfied by, say, a walking-event and an event of stopping walking. Informally, the pre-state of stopping walking is a state of being in the process of walking (a progressive state), and the contextually relevant aspects of different stages of the walking process are of course elaborations on being in the process of walking (say, having taken one step, lifting one's leg for the second time, etc.). Thus, the scales of change associated with walking-events will count as secondary scales of *stop walking*, to which the progressive then can have access. So, given simply a definition of the progressive involving partitivity over scales does not exclude the false positive *x is stopping walking* being true during a walking-event.

However, Definition 21 of the progressive has further conditions on the truth of the progressive, and in particular, the *contrastive requirement*. As seen already above in Section 5.3.2, a contrastive component of the progressive is necessary to exclude some false positives, namely, the incorrect truth of *The coin is coming up heads* in the coin-flipping scenario. This same component also successfully excludes the truth of *x is stopping walking* during the event of *x* walking. Arguably, for the evaluation of a progressive like *x is stopping walking*, the relevant alternative outcomes are {*stop walking, continue walking*}. But then although the scale of change associated with a walking-event (describing different stages of walking) is a non-final subscale of a scale associated with the activity *continue walking*, contradicting the contrastive requirement. Thus, *x is stopping walking* is — correctly — not predicted to be true during *x*'s walk according to Definition 21.

Excluding false negatives. The problem with the definition, as it stands, is that it allows for *false negatives*. In particular, in the very same scenario of x walking, x is con*tinuing to walk* is also—this time incorrectly—predicted by the Definition 21 to be false, because the case is exactly parallel to x is stopping walking: the set of alternative outcome predicates includes x stop walking, and so the contrastive requirement is again not satisfied. The problem is that the *cover event* relation as defined in Section 5.1.3.1 is only sensitive to the relation between the gradual change associated with the cover event and the initial phase of the non-gradual change associated with an instantaneous event, and does not model the intuitive idea that the gradual change is not only the elaboration of the *pre-state* of an achievement predicate, it also brings the state of affairs closer to the *post-state* of the achievement being realized: in the case of arrive, for instance, movement towards the goal brings the theme closer to being *at* the goal. In contrast, walking does not bring the agent closer to stopping walking. However, I see no way of modelling this idea in a classical semantic framework. We could of course simply say that culmination predicates like arrive (but not predicates like *stop*) are specified in the lexicon to be associated with a specific set of secondary scales, but this would be merely a representation of the state of affairs, and would not follow from the properties of the *model*—which would be the requirement of a model-theoretic analysis (cf. the discussion in Section 3.3.1).

The solution I propose is instead to integrate a further line of research into our analysis, namely, one which refers to a notion of "significant difference" among alternatives. Such a notion has been proposed for example by Fara (2000) for the semantics of gradable adjectives like *expensive* (according to which *the car is expensive* means that the car costs significantly more than is typical) and by Lassiter (2011a) for the semantics of the adjectives *probable* and *likely* (according to which *x* is probable is true if the probability of *x* is greater than a value that is significantly greater than the average likelihood of the contextually relevant alternatives).

More pertinent to the present analysis, Martin (2015) introduced such an element of significantly greater difference into Varasdi's (2014) progressive definition in order to explain the behaviour of *defeasible causatives* like *explain*. Defeasible causatives are predicates whose encoded change of state can be entirely denied, e.g., *The teacher explained the theorem to the students* is compatible with the truth of *But the students failed to understand the theorem*. Martin (2015) shows how Varasdi's (2014) progressive theory relying on the properties of a situation being *indicative* of exactly one outcome predicate out of a set of alternatives can be used to explain the behaviour of defeasible causatives if his definition is modified in the following way (my italics): Given a set of alternatives $\Theta_C = \{\theta_1, \theta_2, \dots, \theta_n\}$ and an event *e*, we say that the set of properties $\mathcal{F} = \{P \mid P(e)\}$ of event *e* indicate θ_k w.r.t. Θ_C iff *e* realizes significantly more sine qua non conditions of θ_k than of any other alternative in Θ_C : $|\mathcal{F} \cap \text{Sqn}(\theta_i)|$ is significantly bigger if i = k than otherwise (i.e., if $i \neq k$).

I propose to integrate Martin's (2015) modification of Varasdi's (2014) account into my own definition of the progressive. In particular, we can say that the contrastive requirement of Definition 21 for the truth of $PROG(^{\phi})(s)$ requires the following:

• (*Revised contrastive requirement:*) the ratio of the number of the scales in the set $PrSoC(^{\alpha}\theta_k) \cup SeSoC(^{\alpha}\theta_k)$ that satisfy the rest of the requirements of the progressive and the number of scales in $PrSoC(^{\alpha}\theta_k) \cup SeSoC(^{\alpha}\theta_k)$ that do not do so is significantly greater for $\theta_k = \varphi$ than for any $\theta_k \neq \varphi$ in the set of contextually relevant alternatives Θ_c .

Since each event of walking can be continued and stopped alike, the number of scales of *continue walking* and of *stop walking* whose source is a walking-event is the same. But these are all secondary scales of *stop walking*, which describes a non-gradual change, and whose *primary* scales of change are therefore all two-valued. These two-valued primary scales of change cannot satisfy the requirements of the progressive (cf. the discussion on the interplay of the non-final subscale and complexity requirements in Section 5.3.2), and so a significantly greater percentage of the scales associated with *continue walking* satisfy the conditions of the progressive in such a scenario than those associated with *stop walking*.

That the progressive has such a graded contrastive requirement is useful in our account of progressive achievements, but as Martin's (2015) proposal shows, these do not constitute the sole motivation for it. In fact, Károly Varasdi (p.c.) has suggested that progressive accomplishments probably also require reference to such "significant difference" instead of a hard requirement on the contrast set as in Varasdi (2014) for the following reason: consider a scenario when a ball dropped into the top part of a chute rolls into a box in 99.5 percent of the cases, while in half a percent of the cases, it accidentally (by a design flaw) gets diverted near the bottom and rolls into a puddle instead; while the ball is in the top part of the chute, although the course of events is compatible with both outcomes, it seems that *The box is rolling into the box* is true, or at least truthfully assertable. This claim is testable and must be experimentally verified with native speakers of English; but if it is indeed so, then the revised contrastive requirement is needed for our semantics of the progressive in any case, and exploiting

it further to exclude false negatives like *continuing walking* only provides further support for its necessity.

5.4 Culminations and happenings in the progressive

5.4.1 Deriving progressive culminations and happenings

Progressive culminations and happenings. As for culminations, Definition 21 correctly predicts that their progressive is true during the process parts of their cover events. Because the type-level scales of culminations include the multi-valued scales of change of their cover events as secondary scales, then whenever there is an eventuality *e* realizing a non-final part of the scale of change of some possible cover event e', that is, $\sigma_{w,c}(\gamma, e) \subset_{nf} \sigma_{w,c}(\gamma, e')$, then it also realizes a non-final part of a secondary scale of change of the culmination ($\sigma_{w,c}(\gamma, e) \subset_{nf} S$ for some $S \in SeSoC(^{\phi})$).

For instance, *John is arriving at the station* is true just in case there is an event of John approaching the station whose scale of change in the given discourse context is a subscale of a secondary scale of *John arrive*[®] *at the station*. Formally, if

$$\begin{aligned} \mathsf{SeSoC}_{\gamma}(^{\mathsf{John arrive}} at \ the \ station) &= \{ \langle \cup \{ \llbracket \mathsf{at}(j, z_i) \rrbracket^{w, \gamma, c} \mid w \in W \}, \\ &\cup \{ \llbracket \mathsf{at}(j, z_j) \rrbracket^{w, \gamma, c} \mid w \in W \}, \cup \{ \llbracket \mathsf{at}(j, z_k) \rrbracket^{w, \gamma, c} \mid w \in W \}, \ldots, \\ &\cup \{ \llbracket \mathsf{at}(j, z) \rrbracket^{w, \gamma, c} \mid w \in W \} \rangle, \ldots \}, \end{aligned}$$

where z_i , z_j , z_k are points on a path to the station with an increasingly greater proximity to the station whose location is z, then PROG(^*John arrive*₀ at the station) is true (in the evaluation world w_0) of a state that is part of an event e whose scale of change is

$$\sigma_{w_0,c}(\gamma,e) = \{ \langle \cup \{ \llbracket \mathsf{at}(j,z_i) \rrbracket^{w,\gamma,c} \mid w \in W \}, \cup \{ \llbracket \mathsf{at}(j,z_j) \rrbracket^{w,\gamma,c} \mid w \in W \}, \\ \cup \{ \llbracket \mathsf{at}(j,z_k) \rrbracket^{w,\gamma,c} \mid w \in W \} \rangle.$$

A shortcoming of the present analysis, as it stands, is that it falls short of explaining why a progressive culmination like *John is arriving at the station* is true only at the *very last stage* of the process leading up to it, that is, it fails to account for the imminency meaning component — just like other analyses of progressive achievements, cf. Section 2.2. This will form the topic of Section 5.5.1 below.

In contrast to culminations, however, happenings do not have secondary multivalued scales of change (cf. Section 5.2), so their progressive can never be true at an everyday granularity. They require a fine granularity scenario at which their scale of change is extended (cf. Chapter 4). One exception is when a happening predicate holds of an event that is extended at even a normal, everyday granularity. This is the case for the following example from Dowty (1979, p. 180):

(5.11) John is gradually realizing that you are right.

In this case, the progressive is felicitous, because *gradually realize that* x *is right* is a predicate that is true of events that are extended at everyday levels of granularity, and so its progressive can be true if there is a change corresponding to the nonfinal subscale of a (primary) scale of change associated with this complex predicate. Of course, the same event could also be described without *gradually*, as well, in which case the happening *realize that* x *is right* is automatically understood in a restricted sense as referring to the kind of realization-events that are extended in time, that is, as synonymous with *gradually realize that* x *is right*, because otherwise the sentence could never be true.

Note also that if no change corresponding to that of a cover event is going on, then even the progressive of a culmination predicate is not true. Thus, to take Piñón's (1997) example of a spy committing suicide: prior to shooting herself, the spy cannot be said to be dying, as there is no change matching any cover event of *die*, that is, there is no actual (nonstative) *e* such that $\sigma_{w,c}(\gamma, e) \subset_{nf} S$ for any $S \in SeSoC(die)$.

The advantage of a scale-based approach. The essence of the present account is that the input criterion for the progressive refers to scales of change, rather than aspectual types. On such a framework, we can predict the different readings of different progressive achievements, as seen above. A further advantage of a scale-based account is that by excluding an instantaneous to durative event type coercion, it is unproblematic to retain standard explanations of achievement behaviour relying on non-durativity, such as the infelicitousness of modification by adverbs of completion (cf. Section 5.5 below).

It is conceivable that whether the progressive has access to such a "secondary scale" is a *parameter* which can differ by language. The English progressive does have such access, while, as I will argue in Section 5.6.2, the progressive in Hungarian does not, that is, the only possible reading of a progressive achievement is the slow-motion, fine

granularity one. The *être en train de* progressive in French similar to Hungarian, being tendentially incompatible with achievements on the preliminary process reading (Bertinetto 2000).¹⁶ Italian has a progressive form (*stare* + GERUND) which allows for a preliminary process progressive with achievements (i.e., it has access to secondary scales), and another progressive (*stare a* + INFINITIVE) which does not (Bertinetto 2000).

A further advantage of a scale-based approach to the progressive is that we can account for the non-uniform behaviour of different progressives with a uniform semantics for the progressive. Kearns (2003) and Martin (2011) have suggested that achievements are incompatible with the "standard progressive", because their progressive is true during their preliminary process, rather than their component process. In contrast, on the scale-based account proposed here, we have a unified semantics for the progressive, namely, that of an internal viewpoint over change (rather than temporal parts).

At the same time we can still account for the observation of Martin (2011) (and a related observation of Kearns 2003) to the effect that progressive (culmination-)achievements are special inasmuch as they (as opposed to progressive accomplishments) do not entail the existence of a part of an event which can possibly satisfy the base predicate. For example, no partial arrival is implied to occur by *Mary is arriving at the station*, while *Mary is reading the book* does imply that a partial book-reading event has already taken place. This is because on the present account, the progressive is true during the change possibly culminating in an arrival or the reading of a book, respectively, but while a complete event of reading the book contains partial book-reading events as parts, an event described by *arrive* is still merely the culmination, lacking any part structure. We have no durative *arrive*, it is only the output, the atelic progressive predicate *arriving at the station* that is durative. In the following section, I inspect in more detail the related issue of modification by adverbs of completion, which forms an important topic in an analysis of achievements.

5.4.2 No partial completion

There are two distinct, but closely related observations about achievements having no partial completion. One is the observation that a progressive culmination-achievement

¹⁶In contrast to Hungarian, though, this incompatibility is not absolute, cf. e.g., the felicitous *Il est en train de gagner intelligemment la partie* 'He's winning the game cleverly' (from Martin 2011), but such constructions are generally avoided according to Bertinetto.

does not imply that the achievement itself is partially realized, which I have already discussed above.

The other observation relates to modification by *adverbs of completion* like *halfway, halfway through, completely, partly.* I will here only focus on *halfway (through),* primarily because it syntactically takes as its argument an *-ing* form. All cases without an *-ing* form (such as the infelicitousness of *#Mary has partly arrived at the station*) are easily explained, because in our scale-based semantics, there are no durative achievements at normal granularity levels, and so such achievements cannot be modified by adverbials requiring duration.

What is often assumed (cf. e.g., Piñón 1997, Rothstein 2004) is that achievements cannot be modified by *halfway through*:

(5.12) #Mary is halfway through arriving at the station.

The present analysis leaves the received, duration-based reasoning about this intact. If *halfway through* takes an eventuality predicate as its argument, then in the case of culminations, that is either a non-durative event predicate (in (5.12), arrive-at-the-station), and hence the unacceptability, or the complex progressive predicate (in (5.12), PROG(arrive-at-the-station)), which again is problematic for the following reason. It is generally accepted that adverbs of completion (*halfway through* at the least) can only modify telic predicates (cf. e.g., Vanden Wyngaerd 2001, Rotstein and Winter 2004, Kennedy and McNally 2005, Piñón 2005, but similar observations were already made by Tenny 1987, p. 170), cf. the unacceptability of (5.13).

- (5.13) a. #John is halfway through pushing the cart./#John has halfway pushed the cart.
 - b. #John is halfway through living in New York./#John has halfway lived in New York.

Since progressives are arguably atelic predicates¹⁷ (irrespective of whether we accept the stativizer view of authors like Vlach 1981 and Moens and Steedman 1988, or the process as output view of Mourelatos 1978), we should expect them to be infelicitous in combination with adverbs of completion like *halfway through*. It is important to

¹⁷Standard tests of atelicity (Vendler 1957, Dowty 1979) confirm this. For instance, progressive sentences can be felicitously modified with a temporal *for*-adverbial like *It seemed to me that Robinson Jeffers was building that house for years and years* from http://www.rain.org/campinternet/ backcountry/weeklies/jeffers-eastwood.html.

bear in mind that the *-ing* form can not only signal the progressive, but has several other functions, as well, including nominalization (as in *Eating the cake was not a nice thing to do*), in which case the *-ing* form denotes sets of complete events of the relevant type.

In the case of *halfway through eating the cake, halfway through* can be argued to be the main predicate which syntactically takes an *-ing* complement. This is underpinned by the fact that

- *i) halfway through* can take a nominal complement as in *halfway through the tour* (which indicates that when it combines with an *-ing* form, that is arguably a nom-inalization), and
- *ii)* halfway through can appear without any complements as in *I'm* halfway through (which indicates that it is the main predicate). Although such examples might perhaps be argued to be a case of ellipsis, such ellipsis is not possible in the case of partly and completely, which are genuinely adverbs (#*I'm* partly).

We may therefore conclude that all adverbs of completion modify the base predicate rather than the progressive, even *halfway* (*through*) which requires an *-ing* form. But as already noted, the base predicate in the case of culminations is non-durative in the present framework, and so infelicitousness is expected.

On the other hand, we do find occasional examples of such modification of achievements with adverbs of completion, that is, at least for some speakers, at least some cases are acceptable. For instance, we find examples with happenings like *explode*:

- (5.14) a. The bomb, halfway through exploding, turned itself into a beam of energy...¹⁸
 - b. Then with expert skill she did a somersault through the half-open sliding door and crawled behind two round barrels that looked as though they had frozen halfway through exploding.¹⁹

This is unsurprising, because happenings are expected to behave like accomplishments in a suitably fine-grained scenario.

But there are also cases with *halfway through* modifying a culmination. However, as an inspection of the interpretation of *halfway*-modification will indicate, these adverbs do not refer to the same interval as (preliminary process) progressive culmina-

¹⁸http://www.ihoz.com/spandex-p17.html

¹⁹http://www.kidpub.com/book-page-or-chapter/awoken-chapter-one-nick-1893129929

tions. They either make reference to the complete interval of change leading up to the culmination (i.e., the complete cover event), as in (5.15a), or to a smaller interval corresponding to the actual change described by the achievement, which enforces a slow-motion reading, as in (5.15b).

- (5.15) a. **Halfway through arriving**, I'm told that we're all gonna sit down and watch a slideshow about my brother.²⁰
 - b. The zoom-in-freeze-shot of the guy who killed you is hilarious cos' sometimes it catches the guy **halfway through dying**.²¹

Slow-motion readings like (5.15b) are straightforward, as — given a switch to a fine granularity — these cases are durative telic event predicates, just as in examples with happenings like those in (5.14). But what about cases like (5.15a)? Note that we find analogous felicitous examples with *halfway*, which syntactically does not take an *-ing* complement contrary to *halfway through*:

- (5.16) a. If the view at this point was great how much greater could the view be from the top? Some people are satisfied with their current view without realizing that **they have only halfway arrived**. If this is good what would happen if you went higher?²²
 - b. I am no doubt more than **halfway arrived** at insanity...²³

This lends support to the proposal above that *halfway (through)* modifies the base predicate rather than the progressive. And it follows that in order for the modification of achievements (which are non-durative) with adverbs of completion like *halfway (through)* to be acceptable, these adverbs must themselves either coerce achievements into a durative predicate, or — preferably, to retain homogeneity in our semantics — access the achievement's secondary scale of change.

Note that adverbs of completion can combine with both adjectives (*halfway empty*) and verbs (*has halfway eaten X*). Piñón (2005) offered a comprehensive, degree-based account of these two uses, arguing that adverbs of completion make reference to the *degree* to which a situation type is realized. So these adverbs of completion are sensitive to and refer to degrees, i.e., a scale (instead of an event runtime), irrespective of

²⁰http://discussions.redstaffapps.com/viewtopic.php?f=108&t=715

²¹http://forums.taleworlds.com/index.php?topic=178785.90

²²http://willc1day.blogspot.de/2014/08/keep-climbing.html

²³http://www.sandiegoreader.com/news/2009/nov/04/tgif-fear-cold-and-dark/

whether it is a scale of change or an abstract scale such as emptiness, and the extendedness of this scale is a reasonable applicability condition, just as for the progressive.

We can therefore say that for some speakers (e.g., Piñón and Rothstein), such adverbs, unlike the progressive, do *not* have access to a secondary scale of change (and hence *#halfway through arriving* is infelicitous, as it does not satisfy the criterion of an extended scale), while for some other speakers, they do.

5.4.3 Complements of aspectual verbs

Another central characteristic of achievements that any theory thereof should account for is that achievements are infelicitous as complements of aspectual verbs (Dowty 1979, Mittwoch 1991, Piñón 1997, Kearns 2003, Rothstein 2004) — as in (5.17).

(5.17) ?Mary started/finished/stopped/continued arriving at the station.

Let us take the aspectual verb *finish*. Examples with *finish* + an accomplishment predicate are felicitous (cf. (5.18a)), and note that this is so despite the fact that the progressive outputs an atelic predicate, and atelics — again, as with *halfway through* — are not as felicitous as complements of *finish* as accomplishments (cf. Dowty 1979, p. 60, and (5.18b)).²⁴

- (5.18) a. John finished building the house.
 - b. ?John finished pushing the cart/being in the room.

So we can conclude, that just as in the case of *halfway through* modification, the *-ing* form here is arguably not the progressive. Having made this point for one aspectual verb, we can generalize this to all the others (such as *stop, start*) as well, since a convincing argument should be made if we would want to maintain that one aspectual

²⁴Note that there are some felicitous examples, but these appear to me to be of two special kinds. On the one hand, an activity predicate as the complement of *finish* is often contextually understood to be associated with some culmination point. For instance, *I finished working* conveys in an office setting that my work day has ended, i.e., it does not simply convey that I just stopped working and went home.

On the other hand, take a restaurant setting in which you can tell a waiter that *I've finished eating*, even though half of your meal is still on your plate. This is, I would say, a different use of *finish*, which is only compatible with *incremental predicates*, and informally means something like "an event-part of the relevant kind has been completed, and no greater event-part of the same kind will be". On this use, it is incompatible even with accomplishment predicates that are not incremental: *I've finished repairing your computer* cannot be uttered truthfully if the repair is not complete (cf. *I've finished eating my soup*, in a restaurant setting, which can be uttered even though half of the soup is still in the bowl).

verb takes the base predicate as its argument, and another, the progressive. Now, if we do not have coercion of an achievement predicate into an accomplishment predicate in our semantics at all, we have no problem drawing on the non-durativity of achievements in explaining their unacceptability as complements of aspectual verbs.

It must be noted here that happenings are unlike culminations and are admissible as the complement of *start* when the event in question is conceived of as extended, which indicates that they can behave like accomplishments:

(5.19) One of the ammo piles was hit and **started exploding**. (from *Blood on the Risers* by John Leppelman)

The same applies to the complement position of *continue*:

(5.20) The ammunition ship began to blow up, and it **continued exploding** until about 11 a.m. (from *The GI's War: American Soldiers in Europe During World War II* by Edwin P. Hoyt)

Although according to Smith (1991, p. 57), even predicates which we classify as happenings do not typically combine with *finish* or *stop*, we do find examples where they do:

- (5.21) a. Father, of course, went off like a bomb and hasn't stopped exploding yet.
 (from A Land Fit for Heroes: Miss Kirkwood's Class of Heroes by Sullatober Dalton)
 - b. The ominous and rapidly-growing bulge aside, there was plenty to keep geologists on their toes during that two and a half week lull beginning in mid-April 1980. The mountain may have **stopped exploding**, but on April 20th, seismicity hit a two-week high.²⁵
 - c. And when the universe has **finished exploding** all the stars will slow down, like a ball that has been thrown into the air, and they will come to a halt and they will all begin to fall towards the centre of the universe again. (from *The Curious Incident Of The Dog In The Night Time* by Mark Haddon)

²⁵http://blogs.scientificamerican.com/rosetta-stones/prelude-to-acatastrophe-our-best-judgement-of-risk/

d. Why the Population Time Bomb Hasn't Finished Exploding²⁶

In contrast, examples with culminations like *arrive* in the complement position of aspectual verbs seem to require extended, mass or plural themes, resulting in a durative predicate (just as happenings are used duratively in the complement position of aspectual verbs):

- (5.22) a. My dividend cheques **stopped arriving** after only two years²⁷
 - b. There is no way to know whether data has **finished arriving**, or whether more might arrive in a millisecond (from *Python Programming On Win32: Help for Windows Programmers* by Mark Hammond & Andy Robinson)
 - c. The audience had only barely **finished arriving** at the venue before the tickets for the show had sold out.²⁸

It must be noted though that the topics of modification by adverbs of completion and appearance in the complement position of aspectual verbs are both extremely intricate and far-reaching, and it has been impossible to truly do justice to them here. I merely wished to show what I hold to be the core observations and the routes of explanation that the present framework offers. In what follows, I turn to some further important characteristics of culminations and argue that they are all connected in some way to the imminency meaning component of progressive achievements.

5.5 Imminency and other characteristics

Like event coercion accounts of the progressive (e.g., Moens and Steedman 1988, Rothstein 2004, Martin 2011, cf. the discussion in Section 2.2), the account presented here does not in itself account for the constraint of imminency, but as I will argue, this meaning component presents more complications than is customarily assumed, anyway. More importantly, I will discuss characteristics that are specific to culminationachievements, and argue that many of them are related to or interact with the impli-

²⁶http://www.smithsonianmag.com/smart-news/why-the-population-time-bombhasnt-finished-exploding-6752037/?no-ist

²⁷http://www.telegraph.co.uk/finance/personalfinance/money-saving-

tips/jessicainvestigates/11363172/My-dividend-cheques-stopped-arrivingafter-only-two-years.html

²⁸http://www.thedailystar.net/shout/the-jazzperiment-performance-imranahmed-quintet-74799

cation of the imminency of the culmination — similarly to the way I argued in Section 5.3.2 that the lack of a perfect progressive is connected to this implication.

5.5.1 Implication of imminency

Imminency is a well-known meaning component of *arrive*-like progressive achievements (see, e.g., Kearns 1991, Rothstein 2004): a progressive achievement is roughly similar in meaning to a construction with *about to*. Thus, we have the equivalence in (5.23) for these achievements, but the non-equivalence in (5.24) for accomplishments.

- (5.23) a. The flight is now arriving at Gate 10.
 - b. \Leftrightarrow The flight is about to arrive at Gate 10.
- (5.24) a. John is building a garden shed right now.
 - b. \Leftrightarrow John is about to build a garden shed right now.

It is only preliminary process readings which display this characteristic, while slow-motion readings do indeed appear to behave like accomplishments and do not imply that their culmination is close, as evidenced by the following hypothetical but acceptable conversation in a scientific scenario similar to one described by Comrie (1976, p. 42):

(5.25) (A scientist is explaining to her audience a slow-motion recording of a subject's brain as the subject is recognising a face on a photo.)Here you can see the subject recognising a face. It will take quite a long time, as this is an extremely slowed down recording to enable us to study the neurological mechanism behind recognition.

And the progressive of a happening is not synonymous with an *about to* construction, either, as the latter, but not the former, is compatible with the (slow-motion) process not having started yet, as in (5.26).

- (5.26) a. The Challenger is exploding now.
 - b. \Leftrightarrow The Challenger is about to explode now.

In contrast, there is no readily conceivable situation in which only one of *Mary is* arriving at the station and *Mary is about to arrive at the station* is true. It appears

that *arriving at the station* carves out the very final part of its cover event, a journey, and cannot be true before it.

A point of complication, noted already in Section 5.3.2 above, is that some culmination predicates, such as *win* and *die*, do *not* imply imminency in the progressive. Thus, a doctor can truthfully say that *John is dying*, *but I don't know how much time he has*. The imminency meaning component appears to be characteristic of and restricted to those predicates which describe the culmination of some type of physical movement (*arrive*, *reach*). But because the set of monomorphemic achievements in English is quite limited (Filip 2008) — and the set of culmination predicates, in particular, appears to be a closed class — it is difficult, if not impossible, to make well-founded generalizations about how we can characterise the set of culminations that imply imminency.

For this reason, I adopt the idea of *lexicalization* from Piñón's (1997) proposal about progressive achievements (cf. Section 2.2.3). But crucially, I do not believe that it is progressive achievements that are lexicalized; instead, I believe it is the lexical specifications of a culmination predicate itself that should be responsible for the imminency implication. On the present framework, the way this can be achieved is by assuming that some culminations (*arrive, reach*) are not characterised by the scales of their *cover events*, but by the scales of a *final part* of their cover events. That is, it is lexically specified for culminations like *arrive* and *reach* that their secondary scales are the scales of final parts of events of moving to the goal.

5.5.2 Probability of success

Rothstein (2004, p. 143–4) drew attention to the observation (which was noted in passing already by Comrie 1976, p. 47) that reaching the endpoint is a "good possibility" in the case of progressive achievements — as opposed to accomplishments — that is, they "cannot be used with an explicit doubt as to whether it is possible to reach the culmination". Rothstein introduced a principle about progressive achievements encoding precisely this observation, which she *also* invoked in the explanation of the imminency meaning component. However, I here wish to argue that while the imminency of the culmination is virtually always implied by a progressive culmination like *arrive*, the probability of success is a less stable implication.

In particular, there are examples of progressive achievements akin to the progressive accomplishment *Samantha was crossing a minefield, when she was blown up* (Asher 1992, p. 492), which constitute counter-examples to the relevant culmination being probable. Consider a situation in which a group of terrorists aim to blow up a train just before it arrives at the station. Then it is perfectly acceptable for them to say at the crucial moment,

(5.27) The train is arriving at the station, activate the bomb!

although it is not a "good possibility" for the train to arrive at the station under these circumstances, and the speaker is explicitly aware of this fact.

While not incorporating it into our semantics of progressive achievements, we can explain and endorse Comrie's and Rothstein's intuition of the "good possibility" of success in the case of progressive culminations through the fact that these are true at a much smaller interval in comparison to accomplishments. Therefore, there is simply less opportunity for them statistically to be disrupted. Note that this hinges on the objective length of events of the relevant kind, which means that we expect slowmotion progressives to also display this feature, and in an even more pronounced way (their progressive being true at an interval which under normal circumstances is conceived of as punctual). This, intuitively, appears to be true, as sentences like *The bomb is exploding* or *The subject is recognising the stimulus* do appear to strongly imply that the disruption of the process is highly unlikely.

However, because "probability of success" is at most only a very weak implication even in the case of achievements like *arrive*, it is difficult to establish if different achievement predicates,²⁹ or indeed, if accomplishments and achievements, do in fact differ in this respect.

5.5.3 Imminency instead of stages

An assumption put forth by Rothstein (2004, p. 146) is that "progressive achievements are not seen as stages in other events", by which she aims to capture the observation

²⁹As for *win* and *die* (which do not have an imminency implication), they are also compatible with the culmination not having a great probability, as I think the following examples indicate:

 ⁽i) a. Mariota is winning the race right now, but several players are right on his tail. (http://drewtroutman.sportsblog.com/posts/1218752/week_10_heisman_watch.html)

b. And Spiderman is dying now, but he's not going to stay that way. (http://www.comicvine.com/barry-allen/4005-22804/forums/is-thereturn-of-barry-allen-a-complete-failure-619700/)

that (5.28a) can be used to describe Mary crossing the border at Maastricht, and then continuing to Amsterdam, while (5.28b) cannot.

(5.28) a. Mary is running to the Netherlands. In fact she is running to Amsterdam.b. Mary is arriving in the Netherlands. In fact she is arriving in Amsterdam.

Rothstein (2004, p. 146) asserts that this is because the activity part of accomplishments derived from achievements, as opposed to that of regular accomplishments, is lexically empty, and thus "it makes no sense to assert that the BECOME event associated with one derived accomplishment is part of the BECOME event associated with a different derived accomplishment", and (5.28b) therefore "can be used only if the event verifying *Mary arrive in the Netherlands* and *Mary arrive in Amsterdam* are the same event".

I propose that there is a different, simpler explanation for the difference between (5.28a) and (5.28b), which goes as follows. We have established that *arrive* in the progressive implies that the culmination point is close. Now, the arrival in the Netherlands and the arrival in Amsterdam cannot both be a short way away if Mary is running and is crossing the border at Maastricht. In fact, both culminations being a short way away can only happen if Mary is arriving by airplane or by ship (the former because a plane covers huge distances in a very short time; the latter because Amsterdam itself lies on the coast). In contrast, the accomplishments in (5.28a) make no such constraints, and both can therefore be made true if the appropriate conditions hold when Mary is crossing the border at Maastricht.

5.5.4 Modification by agentive adverbials

Achievements are typically not able to combine with agentive adverbials like *deliber-ately, carefully, attentively* (Dowty 1979), and this feature is carried over to their progressive form, as well. Piñón (1997), for example, attributes this behaviour to the intuitive assumption that any sort of intentional activity or act takes time, if only a short time, and so achievements, being instantaneous, lack the temporal extent required for intentional activity. Note that this route of explanation presupposes that achievements (but not semelfactives, which even on a single-event reading can easily be modified with agentive adverbials, cf. *cough deliberately*) are indeed true at exactly one *objec-tive* time point. Considering the problems that an instantaneous view of change poses (cf. Section 3.1), however, one may wish to endorse the idea that there are no objectively instantaneous dynamic events. Also, Piñón's (1997) explanation does not in itself account for the fact that agentive adverbials also fail to modify progressive achievements, which, in whatever way we derive them (or even if progressive achievements are simply lexicalized) are *durative*, and so should on his account be able to combine with agentive adverbials.

I myself only have the following rather weak account of why achievements are incompatible with agentive adverbials, which however does not face the problems of Piñón's (1997) proposal. The conversational goal of an assertion of a sentence with a culmination-achievement like *arrive* or *reach* appears to be the conveying of the temporal location of the culmination (including focusing on its *occurrence*). But it is not really possible for an agent to have control over the temporal location of the culmination in relation to the *reference time* of an assertion.

However, some culminations can, in fact, be quite natural even without the progressive with such adverbs of intention:

- (5.29) a. The way the Courier explained it Fort William had **deliberately won** the toss to get a home draw.³⁰
 - b. I ended up getting Trick Room back off and **carefully won** the game by double targeting the Pokémon he chose not to Protect for three turns in a row.³¹
- (5.30) a. I told a priest this in confession and he told me that if I **died deliberately** (by suicide) I would probably go to hell.³²
 - b. Many have **died carelessly** by swimming here and getting sucked under the rock bowls.³³

But there are several restrictions to these cases. Firstly, we do not really find analogous cases for the progressive. That is, if an achievement predicate in the progressive is modified by an agentive adverbial, it appears to be invariably a case of a pluractional reading, as in (5.31).

³⁰http://keepingoutofthed.blogspot.hu/2009/07/mary-chooses-currachd-ruadhfor-wedding.html

³¹http://nuggetbridge.com/reports/press-x-to-bubble/

³²http://www.chastitysf.com/q_death.htm

³³http://www.virtualtourist.com/travel/North_America/United_States_of_

America/Hawaii_State_of/Hawaii_Big_Island/Hilo-772311/Off_the_Beaten_Path-Hilo-Waterfalls-BR-1.html
(5.31) He was **deliberately winning** games at very close scores to keep the other guy thinking he could win the next game.

Second, the set of predicates admitting this modification is limited, and appears to more or less coincide with those (e.g., *win* or *die*) which do not imply the imminency of the culmination. Of course, since the achievements in these cases are not in the progressive form (cf. the first restriction noted above), an implication of imminency or lack thereof cannot be directly used in the explanation of this phenomenon, but there is a quite distinct difference between exactly those sets of predicates which do and those which do not have this implication in the progressive. Exceptions involving felicitous cases of an agentive adverbial modifying a culmination like *arrive* include examples akin to the following:

(5.32) John deliberately arrived at the party late.

In these cases, however, one might argue that what *deliberately* scopes over in the example is *late*: i.e., it is the lateness of the arrival that was deliberate, not the arrival itself, which is underpinned by the fact that *some* adjunct appears to be necessary in these cases. This is quite parallel to what happens in "pseudocausal sentences" like *The delayed departure caused much consternation* according to Eckardt (2000): *caused* (in our case, *deliberately*) is sensitive to which expression is (silently) focussed and thus provides *alternatives* for interpretation.

5.6 Beyond English

Although many accounts of progressive operators focus on a specific operator in a specific language (very frequently, the English progressive), an implicit or explicit goal of most, if not all theories is to account for cross-linguistic similarities and differences via varying a limited number of parameters (cf. e.g., the enterprises of Deo 2009 or Altshuler 2014). The present section serves as a brief and rough inspection of how the proposed scale-based aspectual framework might be extended to some languages other than English, but without any goal of exhaustiveness or in-depth analysis.

5.6.1 The Russian imperfective and the Hindi perfective

Altshuler (2014) offers a typology of what he calls *partitive operators*, which includes not only progressive and imperfective operators of different languages, but some perfective operators, as well, notably, the Hindi perfective (signalled morpho-syntactically by *-yaa*). He argues that both the Russian imperfective and the Hindi perfective can describe events which are either complete or incomplete with respect to the input accomplishment predicate. Examples cited by Altshuler are as follows:

(5.33)	a.	Ja dočit-yva-l		poslednie stročki pis'ma.				
		I read.up-yva(I	MPF)-PST last		lines	letter		
		'I (have) read the last lines of the letter.'						
	b.	xotja 1 even.though 1	ne do-čita-l not read.up-PS	iх Г ther	do n unti	konca. l end		
		'even though I did not finish it.'					(Russian)	

(5.34) maayaa-ne biskuT-ko **khaa-yaa** par use puuraa nahiin khaa-yaa Maya-ERG cookie-ACC eat-yaa(PFV) but it-ACC finish not eat-yaa(PFV) 'Maya was eating the cookie, but did not finish it.' (Hindi)

Compatibility with both culmination and non-culmination can be captured in the present framework by using the simple non-strict subset relation \subseteq as opposed to the non-final subscale relation \subset_{nf} in the semantics of the aspectual operator. That is, while the English progressive would require there to be a strictly higher degree than that mapped to the endpoint of an event e on the scale of change for the relevant predicate, the Russian imperfective and the Hindi perfective allow for the event e itself to satisfy the relevant predicate. This is a direct adoption of the event-based analysis offered by Altshuler (2014), as well as Filip (1993/99, p. 214), who use $e \subseteq e'$ in the semantics of the Slavic imperfective and $e \sqsubset e'$ in that of the English progressive.

So I propose the following tentative definition, which aims to capture the Russian imperfective (and we may assume something similar for the Hindi perfective, cf. below):

Definition 22 (The imperfective in a scale-based account) $[IMPF(^{\phi})(e)]^{M,w,g,c,\gamma} = 1$ iff

1. (Subscale requirement:) there is a scale $S \in (PrSoC(^{\varphi}) \cup SeSoC(^{\varphi}))$ such that $\sigma_{w,c}(\gamma, g(e)) \subseteq S$, and

- 2. (Complexity requirement:) $\sigma_{w,c}(\gamma, g(e))$ is of the same complexity as S and
- 3. (Contrastive requirement:) there is no $\psi \neq \varphi \in \Theta_c$ such that $\sigma_{w,c}(\gamma, g(e)) \subseteq S'$ for some $S' \in (PrSoC(^{\psi}) \cup SeSoC(^{\psi}))$, where Θ_c is a set of mutually disjoint predicates that includes φ , given by context c.

The analysis of the difference between the (Hindi) perfective and the (Russian) imperfective, in turn, can, again, be adopted from Altshuler (2014) and Filip (1993/99, Ch. 4). In particular, Filip (1993/99) argues that the (Slavic) perfective applies to maximal ("total") events, and Altshuler (2014) argues that the perfective (but not the imperfective) describes the *maximal stage* of an event only. In our scale of change-based approach, this can be translated as the Hindi perfective being true only of an event whose endpoint is associated with the maximally attained degree in its context on the relevant scale of change. That is, an addition to Definition 22 for the Hindi perfective would be the following:

♦ (*Maximality requirement:*) there is no e' occurring in w such that $\sigma_{w,c}(\gamma, g(e)) \subset \sigma_{w,c}(\gamma, e')$ and $\sigma_{w,c}(\gamma, e') \subseteq S''$ for some $S'' \in (PrSoC(^{\varphi}) \cup SeSoC(^{\varphi}))$.

In effect, the present section showed that former insights about aspectual operators encoded in event-mereological accounts can be incorporated into the scale-based account of the progressive, as well.

5.6.2 The restricted progressive in Hungarian

The Hungarian progressive is more restrictive than English in many respects. For one, the Hungarian progressive does not allow a futurate reading, but only a partitive interpretation: a sentence like *I'm going to the cinema tomorrow* cannot be expressed with the progressive construction. I will here focus on another limitation: the Hungarian progressive does not have a preliminary process reading. But first, it is necessary to offer a brief introduction to the Hungarian progressive in general.

The Hungarian progressive. The Hungarian progressive, unlike the English progressive, is not wholly productive, and is subject to numerous restrictions. For one, its interpretational opportunities are restricted, and may only have the traditional, "viewpoint" reading, i.e., of viewing an event "from within" (cf. Bennett and Partee 1972). This restriction is related to the restriction that the Hungarian progressive needs to be anchored to some specific time—either the speech time, or to an overt or implicit but contextually recoverable event. The temporal trace of the eventuality which a progressive describes is understood to properly include that time. (Csirmaz 2008, drawing on Jespersen, calls this the *framing effect*.)

Also, the progressive in Hungarian is traditionally assumed to be restricted to agentive predicates (Kiefer 2006). However, this is not a strict criterion, as there are plenty of acceptable non-agentive progressive sentences, particularly sentences describing natural phenomena, such as the Hungarian equivalent of *The Sun was setting* in (5.36) below. In fact, the agentive requirement of the Hungarian progressive may actually be disappearing diachronically.

Moving to restrictions on the surface form, the Hungarian progressive has no morphological marking, as it is only signalled by word order (particle-verb inversion) and a distinctive suprasegmental form (accenting each postverbal constituent, including the verbal particle). For example (indicating stress with '), 'Józsi 'le-ment 'Joseph went down' is perfective, while 'Józsi ('épp) 'ment 'le 'Joseph was going down' is progressive. As a result, focus, which results in the same inverted word order and a specific suprasegmental pattern (postverbal deaccenting), neutralizes the difference between the perfective and the progressive, making it difficult to test whether a specific sentence is progressive. In what follows, therefore, my examples will include the particle épp 'just/just now', which serves to signal progressivity.

Since the progressive in Hungarian is signalled by particle-verb order inversion, it is expected that verbs without a particle can be ambiguous with respect to their grammatical aspect. This is, in fact, borne out: for instance, *telefonál* 'talk with/call someone over the phone' can be interpreted both perfectively and imperfectively.

The Hungarian progressive is also not wholly productive in terms of which particles allow a progressive form, and the most general telicizing particle *meg*- is not generally acceptable in a progressive construction. Particles typically allowing the progressive are directional ones, unless the meaning of the particle+verb complex is noncompositional, as in *be-néz* 'fail' (colloquial, lit.: 'in-see'). Therefore, in what follows, I will use verbs with such particles where the meaning of the particle+verb complex is at least partly compositional (and so some similar verb complexes are acceptable in the progressive). **No preliminary process progressives.** Hungarian achievement verbs are not generally compatible with the progressive aspect in Hungarian (Kiefer 1992, 2006). In particular, culminations are genuinely incompatible with a preliminary process reading of the progressive which is their most natural progressive reading in English. The following sentence is thus infelicitous when the train is not actually rolling into the station:

(5.35) ?A vonat épp **érkezett be** az állomásra. The train just arrive.PAST PRT(in) the station.SUBL 'The train was arriving at the station.'

Because the Hungarian progressive is often assumed to require that its predicate argument be agentive (cf. Kiefer 2006), and achievements are sometimes regarded as non-agentive (cf. Pustejovsky 1991), their incompatibility appears to be explained trivially. However, quite aside from the issue of whether or not achievements can in general be regarded as non-agentive, there are, in fact, acceptable non-agentive progressives in Hungarian, such as the following:³⁴

(5.36) Épp **kelt fel** a nap. just rise PRT(up) the sun 'The sun was rising.'

So the agentivity requirement in itself is not enough to explain the unavailability of preliminary process progressives in Hungarian. Moreover, a slow-motion reading *is* available to achievements, even non-agentive ones, cf.:³⁵

(5.37) ... épp **esett le** a gyerek a pelenkázóról just fall PRT(down) the child the baby changing table-ABL '... the child was falling off the baby changing table.'

This seems to indicate that while the progressive in English has access to secondary scales of change, the same is not true of the Hungarian progressive. That is, the definition of the Hungarian progressive only has $PrSoC(^{\phi})$ in its clauses, rather than the disjunctive $PrSoC(^{\phi}) \cup SeSoC(^{\phi})$ as the English progressive. As such, the applicability condition of the progressive prohibits its application to culminations in Hungarian, unless their scale of change is multi-valued via a switch to a finer granularity.

³⁴http://iroklub.napvilag.net/iras/22226

³⁵http://magyarnarancs.hu/lelek/keresik-a-kifogasokat-78161

Chapter 6

CONCLUSION

The aim of this dissertation was to show that viewing time and change as scales and adopting various tools from scale-based theories for an analysis of durativity, granularity, aspectual classes and viewpoint aspect, a number of different puzzles relating to achievements and verb semantics in general cease to be puzzling for our semantics.

One innovation of the dissertation is the relativisation of the temporal trace function to granularity functions, originally developed for an analysis of scalar vague predicates. This, I argued, can be used in an analysis of slow-motion durative uses of achievements. Some further uses of a granularity-relative temporal trace — and thereby a granularity-relative notion of extendedness — include approaching the wellknown minimal parts problem of activities without the need for extra formal tools, as well as an analysis of the lesser known, but equally puzzling problem of vague event boundaries.

A further innovation of the dissertation is a novel notion of a scale of change that integrates the scale notion of scale-based approaches to aspect and the notion of a perspective over a situation from previous work on the progressive. Building on this concept of a scale of change, I approached both aspectual classes and the semantics of the progressive in a new scale-based framework, which differs from previous scalebased approaches to aspect in that it applies to a greater range of eventuality predicates, instead of only those with incremental arguments. The differences between aspectual classes, and in particular, different achievements of Vendler, are modelled and explained as differences in the structure of the scale of change associated with such predicates:

- Culmination-achievements like *arrive* are associated with both their own binary scales of change and the multi-valued scales of change of their cover events, which are events of gradual change culminating in an event in the extension of the culmination-achievement. As a result of this characterisation, the progressive interpretation most natural to these predicates is that of a preliminary process progressive, because the gradual change the progressive can access is the change leading up to the culmination point.
- Non-semelfactive happenings like *recognise* are associated with a binary scale of change at everyday granularity levels and multi-valued scales at suitably fine granularity levels, and their result states are required to hold for more than an instant. As such, the progressive interpretation most natural to them is a slowmotion one (rather than an iterative or a preliminary process one).
- Semelfactives like *flash* are associated with binary scales of change with no temporal specification for either their preliminary or result states. As a result, their most natural progressive interpretation is an iterative one.

I have argued that once we model the structures of change associated with different predicates with scales and assume a scale-based approach to durativity and the progressive, we can systematically derive the two progressive readings specific to achievements (those that I called the preliminary process and slow-motion uses) and their behavioural characteristics, without needing to assume "durative achievements" as input to the progressive operator or a non-uniform semantics for the progressive. Under the present proposal, the applicability condition of the progressive concerns the scale, rather than the eventuality type, which, I argued, leads to a more constrained account that can explain the different readings of the progressive when applied to predicates of different types (accomplishments, culminations, happenings alike) while at the same time retaining a unified semantics for this aspectual operator.

The thesis contains ideas that are relevant for scale-based accounts in formal semantics, in general. First, it unifies two different lines of research in scale-based literature: one focusing on granularity and the vagueness of scalar expressions, and another focusing on aspectual phenomena. And in contrast to what is the received practice in scale-based approaches, the framework presented here allows a single predicate (in particular, culmination predicates) to be associated with two different kinds of scales (in particular, binary and multi-valued scales). The proposal also aims to draw attention to the fact that the scale-based approach can be generalized and used to explain a greater range of phenomena by exploiting extreme cases satisfying the formal definition of a scale. The notion of a scale of change itself is a case in point: it is simply a series of states with an arbitrary ordering over them (which is of course required to respect temporal flow, but is not necessarily a "natural" ordering like a scale of dryness).

However, a number of questions have been left unanswered and innumerable details have not have been worked out in the present thesis. A question relating to progressive achievements that has been left for further research is for example how the imminency meaning component of progressive achievements can be derived (or whether it can be derived at all instead of simply being a matter of lexical specification as I here assumed), but I did attempt to show that it is an important factor in the semantics of culminations over and above implying synonymity with an *about to* construction. I also did not tease apart the lexical content of verbs and aspects of their meaning that are defined at the sentential level and/or in a specific context. And while I brought several different arguments for a minimally three-way distinction among Vendler's achievements, I did not inspect a wide range of predicates belonging to these classes, but focussed instead on predicates which have formed the centre of the debate on progressive achievements, and I also did not look for further potential classes of nondurative predicates.

In this light, the work done in the present thesis appears rather limited in its scope and results, but it adds to the ongoing discussion in the topics of the progressive, aspectual classes, achievements, and scale-based semantics.

BIBLIOGRAPHY

- Abusch, Dorit (2010). Presupposition triggering from alternatives. *Journal of Semantics* 27(1), 37–80.
- Altshuler, Daniel (2014). A typology of partitive aspectual operators. *Natural Language & Linguistic Theory* 32(3), 735–775.
- Altshuler, Daniel and Hana Filip (2013). Aspect: From verb to discourse. ESSLLI 2013 Lecture Notes.
- Altshuler, Daniel and Roger Schwarzschild (2013). Moment of change, cessation implicatures and simultaneous readings. In E. Chemla, V. Homer, and G. Winterstein (Eds.), *Proceedings* of Sinn und Bedeutung 17, Paris, pp. 45–62.
- Arche, María J. (2014). The construction of viewpoint aspect: the imperfective revisited. *Natural Language & Linguistic Theory* 32(3), 791–831.
- Asher, Nicholas (1992). A default, truth conditional semantics for the progressive. *Linguistics and Philosophy* 15(5), 469–508.
- Bach, Emmon (1981). On time, tense, and aspect: An essay in English metaphysics. In P. Cole (Ed.), *Radical Pragmatics*, pp. 63–81. Academic Press.
- Bach, Emmon (1986). The algebra of events. Linguistics and Philosophy 9, 5–16.
- Bale, Alan Clinton (2008). A universal scale of comparison. Linguistics and Philosophy 31, 1–55.
- Bale, Alan Clinton (2011). Scales and comparison classes. *Natural Language Semantics* 19, 169–190.
- Barker, Chris (2012). Possessives and relational nouns. In C. Maienborn, K. von Heusinger, and P. Portner (Eds.), *Semantics: An International Handbook of Natural Language Meaning*, Volume 3, pp. 1109–1130. Berlin & New York: Mouton de Gruyter.
- Beaver, David (1995). *Presuppositions and Assertions in Dynamic Semantics*. Ph. D. thesis, University of Edinburgh.
- Beaver, David and Bart Geurts (2013). Presupposition. In E. N. Zalta (Ed.), *The Stanford Encyclopedia of Philosophy* (Fall 2013 ed.).
- Beavers, John (2002). Aspect and the distribution of prepositional resultative phrases in English. *LinGO Working Paper* (7).
- Beavers, John (2008). Scalar complexity and the structure of events. In J. Dölling, T. Heyde-Zybatow, and M. Schäfer (Eds.), *Event Structures in Linguistic Form and Interpretation*, pp.

245–265. Walter de Gruyter.

- Beavers, John (2010). The structure of lexical meaning: Why semantics really matters. *Language 86*, 821–864.
- Beavers, John (2011). On affectedness. Natural Language and Linguistic Theory 29, 335–370.
- Beavers, John (2012). Lexical aspect and multiple incremental themes. In V. Demonte and L. McNally (Eds.), *Telicity, Change, and State: A Cross-Categorial View of Event Structure*, pp. 23–59. Oxford: Oxford University Press.
- Beavers, John (2013). Aspectual classes and scales of change. *Linguistics* 54, 681–706.
- Bennett, Michael and Barbara H. Partee (1972). Toward the logic of tense and aspect in English. Technical report, Systems Development Corporation, Santa Monica. Reprinted in *Compositionality in Formal Semantics: Selected Papers by Barbara H. Partee*, (2004), Blackwell Publishing, pp. 59–109. Page numbers refer to the reprint.
- Bertinetto, Pier Marco (2000). The progressive in Romance, as compared with English. In Ö. Dahl (Ed.), *Tense and Aspect in the Languages of Europe*, pp. 559–604. Berlin, New York: Mouton de Gruyter.
- Bertinetto, Pier Marco, Karen H. Ebert, and Casper de Groot (2000). The progressive in Europe.In Ö. Dahl (Ed.), *Tense and Aspect in the Languages of Europe*, pp. 517–558. Berlin, New York: Mouton de Gruyter.
- Bertinetto, Pier Marco and Alessandro Lenci (2012). Iterativity vs. habituality: On the iterative interpretation of perfective sentences. In R. I. Binnick (Ed.), *The Oxford Handbook of Tense and Aspect*. Oxford University Press.
- Bill, Cory, Jacopo Romoli, Florian Schwarz, and Stephen Crain (2014). Indirect scalar implicatures are neither scalar implicatures nor presuppositions (or both). Poster presented at the 27th Annual CUNY Conference on Human Sentence Processing, Columbus, USA.
- Bonomi, Andrea (1999). The progressive and the structure of events. *Journal of Semantics* 14, 173 205.
- Bunt, Harry C. (1985). Mass Terms and Model-Theoretic Semantics. Cambridge University Press.
- Burnett, Heather (2014). A delineation solution to the puzzles of absolute adjectives. *Linguistics and Philosophy* 37(1), 1–39.
- Carlson, Greg N. (1977). A unified analysis of the english bare plural. *Linguistics and Philoso-phy* 1(3), 413–457.
- Carlson, Lauri (1981). Aspect and quantification. In P. Tedeschi and A. Zaenen (Eds.), *Tense and Aspect*, Volume 14 of *Syntax and Semantics*, pp. 31–64. New York: Academic Press.
- Champollion, Lucas (2010). *Parts of a whole: Distributivity as a bridge between aspect and measurement*. Ph. D. thesis, University of Pennsylvania.
- Chemla, Emmanuel (2009). Presuppositions of quantified sentences: experimental data. *Natural Language Semantics* 17(4), 299–340.
- Chierchia, Gennaro (2010). Mass nouns, vagueness, and semantic variation. *Synthese* 174, 99–149.

- Comrie, Bernard (1976). Aspect: An Introduction to the Study of Verbal Aspect and Related Problems. Cambridge University Press.
- Cresswell, Maxwell John (1976). The semantics of degree. In B. Partee (Ed.), *Montague grammar*, pp. 261–292. New York: Academic Press.
- Csirmaz, Anikó (2008). Particles and a two component theory of aspect. In K. É. Kiss (Ed.), *Event Structure And The Left Periphery*, Volume 68 of *Studies In Natural Language And Linguistic Theory*, pp. 107–128. Springer.
- Cumming, Sam, Gabriel Greenberg, and Rory Kelly (2014). Spatial coherence in visual narrative. Talk presented at Heinrich-Heine-Universität Düsseldorf.
- Dahl, Östen (1985). Tense and Aspect Systems. Basil Blackwell.
- Davidson, Donald (1967). The logical form of action sentences. In N. Rescher (Ed.), *The Logic* of *Decision and Action*. University of Pittsburgh Press.
- de Bruin, Jos and Remko Scha (1988). The interpretation of relational nouns. In *Proceedings of the ACL*, Buffalo, NY, pp. 25–32.
- de Swart, Henriëtte (2007). A cross-linguistic discourse analysis of the Perfect. *Journal of Pragmatics* 39, 2273–2307.
- Deo, Ashwini (2009). Unifying the imperfective and the progressive: partitions as quantificational domains. *Linguistics and Philosophy* 32, 475–521.
- Deo, Ashwini and Itamar Francez (2013). From change to value difference in degree achievements. In *Proceedings of Semantic and Linguistic Theory (SALT)* 23, pp. 97–115.
- Dini, Luca and Pier Marco Bertinetto (1995). Punctual verbs and the linguistic ontology of events. Paper presented at Fact and Events Conference, Trento.
- Dowty, David R. (1977). Toward a semantic analysis of verb aspect and the english 'imperfective' progressive. *Linguistics and Philosophy* 1(1), 45–77.
- Dowty, David R. (1979). Word Meaning and Montague Grammar. Dordrecht: D. Reidel.
- Dowty, David R. (1986). The effects of aspectual class on the temporal structure of discourse: semantics or pragmatics? *Linguistics and Philosophy* 9(1), 37–61.
- Dowty, David R. (1991). Thematic proto-roles and argument selection. Language 67, 547-619.
- Ebert, Karen H. (2000). Progressive markers in Germanic languages. In Ö. Dahl (Ed.), *Tense and Aspect in the Languages of Europe*, pp. 605–653. Berlin, New York: Mouton de Gruyter.
- Eckardt, Regine (2000). Causation, contexts and event individuation. In J. Higginbotham, F. Pianesi, and A. C. Varzi (Eds.), *Speaking of Events*, pp. 105–121. Oxford University Press.
- Engelberg, Stefan (2000). The magic of the moment: What it means to be a punctual verb. In *Proceedings of the 25th Annual Meeting of the Berkeley Linguistics Society*, pp. 109–121.
- Fara, Delia Graff (2000). Shifting sands: An interest relative theory of vagueness. *Philosophical Topics* 28(1), 45–81.
- Feldhütter, Irmgard, Margret Schleidt, and Irenäus Eibl-Eibesfeldt (1990). Moving in the beat of seconds: Analysis of the time structure of human action. *Ethology and Sociobiology* 11(6), 511–520.

- Filip, Hana (1993/99). *Aspect, Eventuality Types, and Nominal Reference*. New York: Routledge. Page numbers refer to the 1999 published version of the 1993 Ph.D. dissertation.
- Filip, Hana (2000). The quantization puzzle. In C. Tenny and J. Pustejovsky (Eds.), *Events as Grammatical Objects, from the Combined Perspectives of Lexical Semantics, Logical semantics and Syntax*, pp. 39–91. Stanford: CSLI Press.
- Filip, Hana (2008). Events and maximalization: The case of telicity and perfectivity. In S. D. Rothstein (Ed.), *Theoretical and Crosslinguistic Approaches to the Semantics of Aspect*, pp. 217–256. John Benjamins.
- Filip, Hana (2011a). Aspectual class and aktionsart. In C. Maienborn, K. von Heusinger, and P. Portner (Eds.), *Semantics: An International Handbook of Natural Language Meaning*, Volume 2, pp. 1186–1217. Berlin & New York: Mouton de Gruyter.
- Filip, Hana (2011b). Sums and scales. Talk presented at the *Workshop on Scalarity in Verb-based Constructions*, Düsseldorf.
- Filip, Hana and Susan Rothstein (2006). Telicity as a semantic parameter. In J. Lavine, S. Franks, H. Filip, and M. Tasseva-Kurktchieva (Eds.), *Formal Approaches to Slavic Linguistics (FASL 14) The Princeton University Meeting*, Ann Arbor, MI, pp. 139–156. University of Michigan Slavic Publications.
- Fine, Kit (1975). Vagueness, truth and logic. Synthese 30(3-4), 265–300.
- Fleischhauer, Jens (2013). Interaction of telicity and degree gradation in change of state verbs. In B. Arsenijević, B. Gehrke, and R. Marín (Eds.), *Studies in the Composition and Decomposition of Event Predicates*, Volume 93 of *Studies in Linguistics and Philosophy*. Springer.
- Fleischhauer, Jens and Thomas Gamerschlag (2014). We are going through changes: How change of state verbs and arguments combine in scale composition. *Lingua* 141, 30–47.
- Fults, Scott (2011). Vagueness and scales. In P. Égré and N. Klinedinst (Eds.), *Vagueness and Language Use*, pp. 25–50. New York: Palgrave Macmillan.
- Garey, Howard B. (1957). Verbal aspect in French. Language 33, 91–110.
- Gawron, Jean Mark (2005). Generalized paths. In *Proceedings from the Fifteenth Conference on Semantics and Linguistic Theory (SALT XV)*, Ithaca, New York. CLC Publications.
- Gawron, Jean Mark (2009). The lexical semantics of extent verbs. MS.
- Gendler Szabó, Zoltán (2004). On the progressive and the perfective. *Noûs* 38(1), 29–59.
- Gendler Szabó, Zoltán (2008). Things in progress. Philosophical Perspectives 22(1), 499–525.
- Gruber, Jeffrey Steven (1965). *Studies in Lexical Relations*. Ph. D. thesis, Massachusetts Institute of Technology, Cambridge, MA.
- Gyarmathy, Zsófia (2011). The role of granularity in the semantics of achievements in the progressive. Talk presented at the *Workshop on Scalarity in Verb-based Constructions*, Düsseldorf.
- Gyarmathy, Zsófia (2012). The role of granularity in event semantics. In F. Kiefer and Z. Bánréti (Eds.), *Twenty Years of Theoretical Linguistics in Budapest*, pp. 93–109. Budapest: Tinta Publishing House.

- Gyarmathy, Zsófia (2014). Scale coercion and the progressive form of achievements. Talk presented at the 11th International Conference on Actionality, Tense, Aspect, Modality/Evidentiality (Chronos 11), Pisa.
- Gyarmathy, Zsófia (2015). Durativity, granularity, and vagueness. Talk presented at the *16th Szklarska Poreba Workshop*, Poland.
- Hay, Jennifer, Christopher Kennedy, and Beth Levin (1999). Scalar structure underlies telicity in "Degree Achievements". In T. Mathews and D. Strolovitch (Eds.), *Proceedings of SALT IX*, Ithaca, pp. 127–144.
- Heyde-Zybatow, Tatjana (2004). Achievements: two experimental studies and one semantic analysis. Talk presented at *Sinn und Bedeutung 9*, Nijmegen.
- Heyde-Zybatow, Tatjana (2008). What kind of events do achievements describe? In A. Steube (Ed.), *The Discourse Potential of Underspecified Structures*, Volume 8 of *Language*, *Context and Cognition*, pp. 109–144. Berlin: De Gruyter.
- Horn, Christian (in preparation). *Nominal concept types and determination in German*. Ph. D. thesis, Heinrich-Heine-Universität Düsseldorf.
- Jackendoff, Ray (1983). *Semantics and Cognition*, Volume 8 of *Current Studies in Linguistics*. Cambridge, MA: MIT Press.
- Jackendoff, Ray (1996). The proper treatment of measuring out, telicity, and perhaps even quantification in English. *Natural Language & Linguistic Theory* 14(2), 305–354.
- Janssen, Theo M.V. (1997). Compositionality. In J. van Benthem and A. ter Meulen (Eds.), *Handbook of Logic and Language*, pp. 417–473. New York: Elsevier.
- Jespersen, Otto (1931). *Modern English Grammar on Historical Principles, Part IV*. London, Copenhagen: Allen & Unwin, Ejnar Munksgaard.
- Kamp, Hans (1979/2013). Events, instants and temporal reference. In K. von Heusinger and A. ter Meulen (Eds.), *Meaning and the Dynamics of Interpretation: Selected Papers of Hans Kamp*, pp. 54–103. Leide: Brill. Reprint of the 1979 paper from *Semantics from Different Points of View* edited by R. Bäuerle, U. Egli and A. von Stechow.
- Kamp, Hans and Uwe Reyle (1993). From Discourse to Logic: Introduction to Modeltheoretic Semantics of Natural Language, Formal Logic and Discourse Representation Theory, Volume 42 of Studies in Linguistics and Philosophy. Dordrecht: Kluwer.
- Kardos, Éva Alíz (2012). *Toward a Scalar Semantic Analysis of Telicity in Hungarian*. Ph. D. thesis, Debreceni Egyetem.
- Karttunen, Lauri (1973). Presuppositions of compound sentences. *Linguistic Inquiry* 4(2), 169–193.
- Kearns, Katherine Susan (1991). *The Semantics of the English Progressive*. Ph. D. thesis, Massachusetts Institute of Technology.
- Kearns, Katherine Susan (2003). Durative achievements and individual-level predicates on events. *Linguistics and Philosophy* 26, 595–635.
- Kearns, Katherine Susan (2007). Telic senses of deadjectival verbs. Lingua 117, 26–66.

- Kennedy, Christopher (2001). Polar opposition and the ontology of 'degrees'. *Linguistics and Philosophy* 24(1), 33–70.
- Kennedy, Christopher (2007). Vagueness and grammar: The semantics of relative and absolute gradable adjectives. *Linguistics and Philosophy* 30(1), 1–45.
- Kennedy, Christopher and Beth Levin (2008). Measure of change: The adjectival core of degree achievements. In L. McNally and C. Kennedy (Eds.), *Adjectives and Adverbs: Syntax, Semantics and Discourse*, pp. 156–182. Oxford, UK: Oxford University Press.
- Kennedy, Christopher and Louise McNally (2005). Scale structure, degree modification, and the semantics of gradable predicates. *Language* 81, 345–381.
- Kenny, Anthony (1963). Action, Emotion and Will. Routledge.
- Kiefer, Ferenc (1992). Az aspektus és a mondat szerkezete [Aspect and the structure of the sentence]. In F. Kiefer (Ed.), *Strukturális Magyar Nyelvtan I. Mondattan [A Structural Grammar* of Hungarian I. Syntax], pp. 797–886. Budapest: Akadémiai Kiadó.
- Kiefer, Ferenc (2006). Aspektus és akcióminőség különös tekintettel a magyar nyelvre [Aspect and aktionsart with a special focus on Hungarian]. Akadémiai Kiadó.
- Klein, Wolfgang (1994). Time in Language. London: Routledge.
- Kratzer, Angelika (1995). Stage-level and individual-level predicates. In G. N. Carlson and F. J. Pelletier (Eds.), *The Generic Book*, pp. 125–175. Chicago, London: The University of Chicago Press.
- Kratzer, Angelika (2004). Telicity and the meaning of objective case. In J. Guéron and J. Lecarme (Eds.), *The Syntax of Time*, Volume 37 of *Current Studies in Linguistics*, pp. 389–423. MIT Press.
- Krifka, Manfred (1989). Nominal reference, temporal constitution and quantification in event semantics. In R. Bartsch, J. van Benthem, and P. von Emde Boas (Eds.), *Semantics and Contextual Expression*, pp. 75–115. Dordrecht: Foris.
- Krifka, Manfred (1992). Thematic relations as links between nominal reference and temporal constitution. In I. A. Sag and A. Szabolcsi (Eds.), *Lexical Matters*, pp. 29–53. Stanford: CSLI.
- Krifka, Manfred (1998). The origins of telicity. In S. Rothstein (Ed.), *Events and Grammar*, pp. 197–235. Dordrecht: Kluwer.
- Krifka, Manfred (2007). Approximate interpretations of number words: A case for strategic communication. In G. Bouma, I. Krämer, and J. Zwarts (Eds.), *Cognitive Foundations of Communication*, pp. 111–126. Koninklijke Nederlandse Akademie van Wetenschapen.
- Landman, Fred (1991). Structures for Semantics. Springer.
- Landman, Fred (1992). The progressive. Natural Language Semantics 1, 1–32.
- Lascarides, Alex (1991). The progressive and the imperfective paradox. *Synthese* 87(3), 401–447. Lasersohn, Peter (1999). Pragmatic halos. *Language* 75(3), 522–551.
- Lassiter, Daniel (2011a). *Measurement and Modality: The Scalar Basis of Modal Semantics*. Ph. D. thesis, New York University.
- Lassiter, Daniel (2011b). Vagueness as probabilistic linguistic knowledge. In R. Nouwen, R. van Rooij, U. Sauerland, and H.-C. Schmitz (Eds.), *Vagueness in Communication*, Volume 6517 of

Lecture Notes in Computer Science, pp. 127–150. Springer Berlin Heidelberg.

- Levin, Beth and Malka Rappaport-Hovav (2005). *Argument Realization*. Cambridge University Press.
- Link, Godehard (1983). The logical analysis of plurals and mass terms: A lattice-theoretic approach. In R. Bäuerle, C. Schwarze, and A. von Stechow (Eds.), *Meaning, Use and Interpretation of Language*, pp. 302–323. Berlin: de Gruyter.
- Link, Godehard (1998). *Algebraic semantics in language and philosophy*. CSLI Lecture Notes No. 74. CSLI Stanford.
- Löbner, Sebastian (1985). Definites. Journal of Semantics 4(4), 279–326.
- Löbner, Sebastian (2002). *Understanding Semantics*. Understanding Language. London: Hodder Education.
- Maienborn, Claudia (2005). On the limits of the Davidsonian approach: The case of copula sentences. *Theoretical Linguistics* 31(3), 275–316.
- Malink, Marko (2008). Right boundary achievements under conative negation. In A. Steube (Ed.), *The Discourse Potential of Underspecified Structures*, Volume 8 of Language, Context and Cognition, pp. 145–162. Berlin: De Gruyter.
- Martin, Fabienne (2011). Revisiting the distinction between accomplishments and achievements. In W. de Mulder, J. Mortelmans, and T. Mortelmans (Eds.), *From Now to Eternity*, Volume 22 of *Cahiers Chronos*, Amsterdam/New York, pp. 43–64. Rodopi.
- Martin, Fabienne (2015). Explaining the link between agentivity and non-culminating causation. Talk presented at *Semantics and Linguistic Theory (SALT)* 25.
- Merricks, Trenton (2001). Varieties of vagueness. *Philosophy and Phenomenological Research* 62, 145–157.
- Mittwoch, Anita (1991). In defence of Vendler's achievements. *Belgian Journal of Linguistics 6*, 71–85.
- Moens, Marc and Mark Steedman (1988). Temporal ontology and temporal reference. *Computational Linguistics* 14(2), 15–28.
- Moltmann, Friederike (1991). Measure adverbials. Linguistics and Philosophy 14(6), 629-660.
- Mori, Yoshiki, Sebastian Löbner, and Katharina Micha (1992). Aspektuelle Verbklassen im Japanischen. Zeitschrift für Sprachwissenschaft 11, 216–279.
- Mourelatos, Alexander P. D. (1978). Events, processes, and states. *Linguistics and Philosophy 2*, 415–434.
- Mourelatos, Alexander P. D. (1981). Events, processes, and states. In P. Tedeschi and A. Zaenen (Eds.), *Tense and Aspect*, Volume 14 of *Syntax and Semantics*, pp. 191–211. New York: Academic Press. Reprint of 1978 paper.
- Nicolas, David (2014). The logic of mass expressions. In E. N. Zalta (Ed.), *The Stanford Encyclopedia of Philosophy* (Spring 2014 ed.).

Parsons, Terence (1990). Events in the Semantics of English. MIT Press.

- Partee, Barbara H. and Vladimir Borschev (2003). Genitives, relational nouns, and argumentmodifier ambiguity. In E. Lang, C. Maienborn, and C. Fabricius-Hansen (Eds.), *Modifying Adjuncts*. Mouton de Gruyter.
- Partee, Barbara H. and Vladimir Borschev (2012). Sortal, relational, and functional interpretations of nouns and Russian container constructions. *Journal of Semantics* 29(4), 445–486.
- Pelletier, Francis Jeffrey and Lenhart K. Schubert (2003). Mass expressions. In D. M. Gabbay and F. Guenthner (Eds.), *Handbook of philosophical logic* (2 ed.), Volume 10, pp. 249–336. Dordrecht: Kluwer Academic Publishers.
- Piñón, Christopher (1997). Achievements in an event semantics. In *Proceedings of SALT VII*, pp. 273–296.
- Piñón, Christopher (2005). Adverbs of completion in an event semantics. In H. Verkuyl,
 H. de Swart, and A. van Hout (Eds.), *Perspectives on Aspect*, Volume 32 of *Studies in Theoretical Psycholinguistics*, pp. 149–166. Springer.
- Piñón, Christopher (2008a). Aspectual composition with degrees. In L. McNally and C. Kennedy (Eds.), *Adjectives and Adverbs in Semantics and Discourse*, pp. 183–219. Oxford University Press.
- Piñón, Christopher (2008b). Negating right boundary achievements (comments on Malink).
 In A. Steube (Ed.), *The Discourse Potential of Underspecified Structures*, Volume 8 of Language, Context and Cognition, pp. 163–175. Berlin: De Gruyter.
- Pöppel, Ernst (1978). Time perception. In R. Held, H. Leibowitz, and H.-L. Teuber (Eds.), Handbook of Sensory Physiology, Volume 8, pp. 713–729. Berlin, Heidelberg, New York: Springer.
- Portner, Paul (1998). The progressive in modal semantics. Language 74, 760-787.
- Portner, Paul (2003). The (temporal) semantics and (modal) pragmatics of the perfect. *Linguistics and Philosophy 26*, 459–510.
- Pustejovsky, James (1988). The geometry of events. In C. Tenny (Ed.), Studies in Generative Approaches to Aspect, Volume 24 of Lexicon Project Working Papers. Cambridge, Massachusetts: MIT Center for Cognitive Science.
- Pustejovsky, James (1991). The syntax of event structure. Cognition 41(1), 47-81.
- Pustejovsky, James (1998). The Generative Lexicon. MIT Press.
- Ramchand, Gillian Catriona (2008). *Verb Meaning and the Lexicon: A First Phase Syntax*. Cambridge Studies in Linguistics. Cambridge University Press.
- Rappaport Hovav, Malka (2008). Lexicalized meaning and the internal temporal structure of events. In S. D. Rothstein (Ed.), *Theoretical and Crosslinguistic Approaches to the Semantics of Aspect*, pp. 13–42. John Benjamins.
- Rector, Alan, Jeremy Rogers, and Thomas Bittner (2005). Granularity scale & collectivity: When size does and doesn't matter. *Journal of Bioinformatics* 39(3), 333–349.
- Reichenbach, Hans (1947). Elements of Symbolic Logic. New York: Random House.
- Romoli, Jacopo (2015). The presuppositions of soft triggers are obligatory scalar implicatures. *Journal of Semantics* 32(2), 173–219.

- Rothstein, Susan (2001). What are incremental themes? In G. Jäger, A. Strigin, C. Wilder, and N. Zhang (Eds.), *Papers on Predicative Constructions*, Volume 22 of *ZAS Papers in Linguistics*. Berlin: Zentrum für Allgemeine Sprachwissenschaft.
- Rothstein, Susan (2004). *Structuring events: A study in the semantics of lexical aspect*. Wiley-Blackwell.
- Rotstein, Carmen and Yoad Winter (2004). Total adjectives vs. partial adjectives: Scale structure and higher-order modifiers. *Natural Language Semantics* 12(3), 259–288.
- Ryle, Gilbert (1949). The Concept of Mind. University of Chicago Press.
- Sassoon, Galit Weidman (2010). Measurement theory in linguistics. Synthese 174, 151–180.
- Sauerland, Uli and Penka Stateva (2007). Scalar vs. epistemic vagueness: Evidence from approximators. In M. Gibson and T. Friedman (Eds.), *Proceedings of SALT 17*, Ithaca, NY. CLC Publications.
- Sauerland, Uli and Penka Stateva (2011). Two types of vagueness. In P. Égré and N. Klinedinst (Eds.), *Vagueness and Language Use*, pp. 121–145. New York: Palgrave Macmillan.
- Schwarzschild, Roger (2002). The grammar of measurement. In B. Jackson (Ed.), *Proceedings of Semantics and Linguistic Theory (SALT) XII.* CLC Publications.
- Schwarzschild, Roger and Karina Wilkinson (2002). Quantifiers in comparatives: A semantics of degree based on intervals. *Natural Language Semantics* 10(1), 1–41.
- Shanon, Benny (1976). On the two kinds of presuppositions in natural language. *Foundations* of Language 14(2), 247–249.
- Smith, Carlota S. (1991). The Parameter of Aspect. Kluwer Academic Press.
- Solt, Stephanie (2014). An alternative theory of imprecision. In T. Snider, S. D'Antonio, and M. Weigand (Eds.), *Proceedings of Semantics and Linguistic Theory (SALT)* 24, Washington DC, pp. 514–533. Linguistics Society of America.
- Solt, Stephanie (2015). Measurement scales in natural language. *Language and Linguistics Compass* 9(1), 14–32.
- Solt, Stephanie and Nicole Gotzner (2012). Experimenting with degree. In *Proceedings of SALT* 22, pp. 166–187.
- Sorensen, Roy (2013). Vagueness. In E. N. Zalta (Ed.), *The Stanford Encyclopedia of Philosophy* (Winter 2013 ed.).
- Steedman, Mark J (1977). Verbs, time, and modality. Cognitive Science 1(2), 216–234.
- Sutton, Peter (2015). Towards a probabilistic semantics for vague adjectives. In H.-C. Schmitz and H. Zeevat (Eds.), *Bayesian Natural Language Semantics and Pragmatics*, Volume 2 of *Language*, *Cognition, and Mind*, Chapter 10. Springer.
- Talmy, Leonard (1985). Lexicalization patterns: Semantic structure in lexical forms. In T. Shopen (Ed.), *Language typology and syntactic description*, Volume 3, pp. 57–149. Cambridge University Press.
- Talmy, Leonard (1988). The relation of grammar to cognition. In B. Rudzka-Ostyn (Ed.), *Topics in Cognitive Linguistics*, pp. 165–205. Amsterdam: John Benjamins.

- Tatevosov, Sergei (2008). Subevental structure and non-culmination. In O. Bonami and P. C. Hofherr (Eds.), *Empirical Issues in Syntax and Semantics* 7, pp. 393–422.
- Taylor, Barry (1977). Tense and continuity. Linguistics and Philosophy 1(2), 199–220.
- Taylor, Kenneth A. (1988). We've got you coming and going. *Linguistics and Philosophy* 11, 493–513.
- Tenny, Carol (1987). *Grammaticalizing Aspect and Affectedness*. Ph. D. thesis, Department of Linguistics and Philosophy, MIT, Cambridge, Massachusetts.
- ter Meulen, Alice (1981). An intensional logic for mass terms. *Philosophical Studies* 40(1), 105–125.
- ter Meulen, Alice (1983). The representation of time in natural language. In A. ter Meulen (Ed.), *Studies in Model Theoretic Semantics*, pp. 177–192. Dordrecht: Foris.
- Thieroff, Rolf (2000). On the areal distribution of tense-aspect categories in Europe. In Ö. Dahl (Ed.), *Tense and Aspect in the Languages of Europe*, pp. 265–305. Berlin, New York: Mouton de Gruyter.
- Tonhauser, Judith, David Beaver, Craige Roberts, and Mandy Simons (2013). Toward a taxonomy of projective content. *Language* 89(1), 66.109.
- Turner, Frederick and Ernst Pöppel (1983). The neural lyre: Poetic meter, the brain, and time. *Poetry* 142(5), 277–309.
- Tye, Michael (1990). Vague objects. Mind XCIX(396), 535–557.
- van Rooij, Robert (2011a). Measurement and interadjective comparisons. *Journal of Semantics* 28(3), 335–358.
- van Rooij, Robert (2011b). Vagueness and linguistics. In G. Ronzitti (Ed.), *Vagueness: A Guide*, Volume 19 of *Logic*, *Epistemology*, *and the Unity of Science*, pp. 123–170. Springer Netherlands.
- Vanden Wyngaerd, Guido (2001). Measuring events. Language 77(1), 61–90.
- Varasdi, Károly (2010). A higher order extensional framework for the progressive. *Journal of Semantics* 27(2), 177–206.
- Varasdi, Károly (2014). Making progressives: Necessary conditions are sufficient. *Journal of Semantics* 31(2), 179–207.
- Vendler, Zeno (1957). Verbs and times. The philosophical review 66(2), 143–160.
- Verkuyl, Henk J. (1972). On the compositional nature of the aspects. Dordrecht: Kluwer.
- Verkuyl, Henk J. (1989). Aspectual classes and aspectual composition. *Linguistics and Philoso-phy* 12(1), 39–94.
- Verkuyl, Henk J. (1993). A Theory of Aspectuality: The Interaction Between Temporal and Atemporal Structure, Volume 64 of Cambridge Studies in Linguistics. Cambridge University Press.
- Vlach, Frank (1981). The semantics of the progressive. In P. Tedeschi and A. Zaenen (Eds.), *Tense and Aspect*, Volume 14 of *Syntax and Semantics*, pp. 271–292. New York: Academic Press.
- von Stechow, Arnim (1984). Comparing semantic theories of comparison. *Journal of Semantics 3*, 1–77.

- von Stechow, Arnim (1996). The different readings of *wieder* 'again': A structural account. *Journal of Semantics* 13(2), 87–138.
- von Stechow, Arnim (2009a). The temporal adjectives *früh(er)/spät(er)* 'early(er)'/'late(r)' and the semantics of the positive. In A. Giannakidou and M. Rathert (Eds.), *Quantification*, *Definiteness*, *and Nominalization*, pp. 214–233. Oxford University Press.
- von Stechow, Arnim (2009b). Tenses in compositional semantics. In W. Klein and P. Li (Eds.), *The Expression of Time*, Volume 3, pp. 129–166. Walter de Gruyter.

Williamson, Timothy (1994). Vagueness. New York: Routledge.

- Winter, Yoad (2006). Closure and telicity across categories. In C. Tancredi, M. Kanazawa, I. Imani, and K. Kusumoto (Eds.), *Proceedings of Semantics and Linguistic Theory, SALT16*.
- Zimmermann, Thomas Ede (2011). Model-theoretic semantics. In C. Maienborn, K. von Heusinger, and P. Portner (Eds.), *Semantics: An International Handbook of Natural Language Meaning*, Volume 1, pp. 762–802. Berlin & New York: Mouton de Gruyter.
- Zinova, Yulia and Hana Filip (2014). Meaning components in the constitution of Russian verbs: Presuppositions or implicatures? In *Semantics and Linguistic Theory (SALT)* 24, pp. 353–372. New York: New York University.
- Zucchi, Sandro and Michael White (2001). Twigs, sequences and the temporal constitution of predicates. *Linguistics and Philosophy* 24, 223–270.
- Zwarts, Joost (2005). Prepositional aspect and the algebra of paths. *Linguistics and Philosophy* 28(28), 739–779.

APPENDICES

A The mathematical concepts used in the dissertation

The definitions to follow are taken mostly from Landman (1991).

Structures and orders.

- (*i*) $\wp(A)$ is the *power set* of A which consists of all subsets of A, that is, $\wp(A) = \{X \mid X \subseteq A\}$.
- (*ii*) The *union* $A \cup B$ of two sets A and B is the set containing all elements of A and B, that is, $A \cup B = \{x \mid x \in A \lor x \in B\}$ ("the set of all elements x such that x is an element of A or xis an element of B").

If *H* is a set whose elements are sets, then the set that is the union of all sets in *H* is

$$\bigcup_{X \in H} X = \{ x \in X | X \in H \}.$$

- (*iii*) The *complement* $B \setminus A$ of a set A in set B is the set consisting of the elements of B that are not elements of A, that is, $B \setminus A = \{x \in B \mid x \notin A\}$.
- (*iv*) A *binary relation* from *A* to *B* is a subset of $A \times B$, where $A \times B = \{ \langle a, b \rangle \mid a \in A \land b \in B \}$ is the *Cartesian product* of *A* and *B*.
- (v) The composition $S \circ R$ of $R \subseteq A \times B$ and $S \subseteq B \times C$ is $\{\langle a, c \rangle \in A \times C \mid \exists b \in B : \langle a, b \rangle \in R \land \langle b, c \rangle \in S\}$.
- (vi) A partition of a set A is a set P such that $P \subseteq \wp(A)$ and $\emptyset \notin P$ and $A = \bigcup \{B \mid B \in P\}$ (P covers A) and $\forall X, Y \in P$: $((X \cap Y = \emptyset) \lor X = Y)$ (elements of P do not overlap). Elements of P are called *blocks* or *cells*.
- (vii) $\langle A, \leq \rangle$ is a partially ordered set if \leq is a partial order over A, that is, it satisfies the following for all $x, y, z \in A$:

- (a) $x \leq x$ (reflexivity)
- (b) $(x \le y \land y \le z) \rightarrow x \le z$ (transitivity)
- (c) $(x \le y \land y \le x) \rightarrow x = y$ (antisymmetry)
- If $\langle A, \leq \rangle$ is a partially ordered set, the *A* is called its *carrier set*.
- (*viii*) A *strict partial order* < is defined for a partial order \leq as follows: $x < y \stackrel{\text{def}}{=} x \leq y \land x \neq y$. It satisfies the following for all x, y, z:
 - (a) $\neg(x < x)$ (*irreflexivity*)
 - (b) $(x < y \land y < z) \rightarrow x < z$ (transitivity)
 - (c) $x < y \rightarrow \neg(y < x)$ (asymmetry)
- (*ix*) $\langle A, \leq \rangle$ is a *linearly ordered set* if \leq is a *linear order* (also called a *total order*) over A, that is, it is a partial order which satisfies *totality* (or *connectedness*), that is: $\forall x, y \in A : x \leq y \lor y \leq x$, or for strict orders, $\forall x, y \in A : x < y \lor y < x \lor x = y$.
- (x) An ordering \leq is dense iff $\forall x, y (x < y \rightarrow \exists z (x < z \land z < y))$.
- (xi) An ordering is *discrete* iff it is not dense in any subset of the ordered set.
- (*xii*) A subset *Y* of an ordered set *X* is *convex* iff if *a*, *b* are in *Y*, so is any point between them, that is: $\forall a, b, c \in X$ ($(a < b \land a \in Y \land b \in Y) \rightarrow \forall c ((a < c \land c < b) \rightarrow c \in Y)$).

Bounds, maxima and related terms. For a partial order $\langle T, \leq \rangle$ and a subset *S* of *T*, the following concepts can be defined:

- (*i*) *S* has an *upper bound* (is bounded above) in *T* iff $\exists t \in T : \forall s \in S : s \leq t$.
- (*ii*) *S* has a *lower bound* (is bounded below) in *T* iff $\exists t \in T : \forall s \in S : t \leq s$.
- (*iii*) *S* has a supremum (least upper bound, join) $\sup(S)$ in *T* iff $\exists t \in T$ such that *t* is an upper bound of *S* and $\forall t' \in T$ if *t'* is an upper bound of *S*, then $t \leq t'$.
- (*iv*) *S* has a *infimum* (greatest lower bound, meet) $\inf(S)$ in *T* iff $\exists t \in T$ such that *t* is a lower bound of *S* and $\forall t' \in T$ if *t'* is a lower bound of *S*, then $t' \leq t$.
- (v) S is bounded in T iff S has both an infimum and a supremum.
- (vi) S is lower closed iff S has an infimum t and $t \in S$.
- (vii) S is upper closed iff S has a supremum t and $t \in S$.
- (*viii*) *S* is *closed* iff it is lower closed and upper closed.
- (*ix*) *S* has a maximal element iff $\exists t \in S : \forall s \in S : (s \neq t \rightarrow \neg t \leq s)$

- (*x*) *S* has a minimal element iff $\exists t \in S : \forall s \in S : (s \neq t \rightarrow \neg s \leq t)$
- (*xi*) *S* has a maximum (a one, or a greatest element) $\max(S)$ iff $\exists t \in S : \forall s \in S : (s \leq t)$
- (*xii*) *S* has a minimum (a zero, or a least element) $\min(S)$ iff $\exists t \in S : \forall s \in S : (t \leq s)$
- (*xiii*) *T* is a *join-semilattice* iff any nonempty finite subset of *T* has a join (a least upper bound).

A supremum and an infimum, and a maximum and a minimum, if it exists, is unique (and the maximum is of course also the supremum of a set, likewise a minimum is also the infimum), while a partially ordered set may have any number of upper and lower bounds and maximal and minimal elements. In my thesis, I am mostly concerned with scales, which are linearly ordered, in which case a maximal (minimal) element is also a maximum (minimum).

Existence of a supremum and an infimum. In the case of scales used in formal semantics, whether or not a lower (upper) bounded scale does or does not have a *greatest* lower (*least* upper) bound has no linguistic relevance. In particular, all scales used in formal semantics have the *greatest lower bound* (*least upper bound*) property: all nonempty subsets of the scale that are bounded from below (above) have a greatest lower (least upper) bound.

This is not a trivial property, e.g., it is not satisfied by the set of *rational* numbers: the set of rational numbers smaller than $\sqrt{2}$ is upper bounded, but it does not have a least upper bound in the set of rational numbers, since $\sqrt{2}$, its least upper bound in the set of real numbers, is not a rational number.

Intervals. Let $\mathcal{X} = \langle X, \langle \chi \rangle$ be a linearly ordered set.

i) $Int(\mathcal{X}) \subseteq \wp(X)$ designates the set of *intervals* (or *periods*) constructed from \mathcal{X} , that is, the set of all convex subsets of X.

Each $i \in Int(\mathcal{X})$ is linearly ordered with the *restriction* of $\langle_{\mathcal{X}}$ to i, that is, with $\langle_{\mathcal{X}} | i = \{\langle x, y \rangle \in \langle_{\mathcal{X}} | x, y \in i\}$.

- *ii*) I use $<_{\mathcal{X}}$ in an extended sense for the *complete precedence relation*, and so for any intervals $i_1, i_2 \in Int(\mathcal{X}), i_1 <_{\mathcal{X}} i_2$ iff $\forall x_1 \forall x_2[((x_1 \in i_1) \land (x_2 \in i_2)) \rightarrow x_1 <_{\mathcal{X}} x_2]$.
- *iii)* A *bounded interval* is an interval that is bounded. A *closed interval* is an interval that is closed.

If x is the infimum and y is the supremum of an open bounded interval, then the bounded interval can be given as (x, y) (in this case, x and y are not elements of the interval).

If x is a minimum and y is a maximum of a closed interval, then the interval can be given as [x, y] (in this case, x and y are elements of the interval).

- *iv*) Unless it is necessary to refer to sets in a given context, a *degenerate interval*, i.e., an interval consisting of a single point can be designated only as its element for simplicity, that is, for any $x \in X$, x can designate the degenerate interval [x, x] instead of $\{x\}$.
- *v*) The part-of relation over $Int(\mathcal{X})$ corresponds to the subset relation \subset .

I will use the notions of a supremum and infimum for intervals of strict linear orders (and not just for subsets of non-strict orders, as in the definitions above), for which they can be defined in the following way: if $S = \langle S, \langle S \rangle$ is a linearly ordered set, and $i \in Int(S)$ is an interval of *S*, then *s* is the supremum of *i* iff the following holds:

(i)
$$\forall x \in i: \left(\left(x <_S s \lor x = s \right) \land \forall y \in S \left(y <_S s \to \left[y \in i \lor \forall z (z \in i \to y <_S z) \right] \right) \right)$$

And *s* is the infimum of *i* iff the following holds

$$(ii) \qquad \forall x \in i: \left(\left(s <_S x \lor x = s \right) \land \forall y \in S \left(s <_S y \to \left[y \in i \lor \forall z (z \in i \to z <_S y) \right] \right) \right)$$

Mappings between structures.

- *i*) A *function* $f : A \to B$ from A into B is a relation $f \subseteq A \times B$ such that $\forall a \in A : \exists ! b \in B : \langle a, b \rangle \in f$, that is, the domain of f is A, and there is a unique element of B mapped to each element of A.
- *ii)* A *partial function* $f: A \rightarrow B$ from A into B is a function from some $X \subseteq A$ into B. The *domain* Dom(f) of the partial function f is the subset X of A where f is defined.
- *iii)* A function $f: A \to B$ is *injective* iff $\forall a, a' \in A: f(a) = f(a') \leftrightarrow a = a'$.
- *iv*) A function $f : A \to B$ is surjective iff $\forall b \in B : \exists a \in A : b = f(a)$.
- *v*) A function *f* is a *bijection* iff it is injective and surjective.
- *vi*) If $\mathbf{A} = \langle A, \leq_A \rangle$ and $\mathbf{B} = \langle B, \leq_B \rangle$ are partial orders, then a function $h: A \to B$ is a *homo-morphism* from \mathbf{A} into \mathbf{B} iff $\forall a, a' \in A: a \leq_A a' \to h(a) \leq_B h(a')$.
- *vii*) An *embedding* is an injective homomorphism.
- viii) An isomorphism is a bijective homomorphism.
- *ix)* A *metric* (or *distance function*) on a set *X* is a function $d: X \times X \to \mathbb{R}$, where \mathbb{R} is the set of real numbers, such that for all $x, y, z \in X$, the following holds (where the subscript \mathbb{R} indicates that the relations $\geq_{\mathbb{R}}, \leq_{\mathbb{R}}$ and the operation $+_{\mathbb{R}}$ are the usual ones defined over the real numbers):

- (a) $d(x,y) \geq_{\mathbb{R}} 0$
- (b) d(x, y) = 0 iff x = y
- (c) d(x, y) = d(y, x)
- (d) $d(x,z) \leq_{\mathbb{R}} d(x,y) +_{\mathbb{R}} d(y,z)$

B The scalar trace

Let us define the scalar trace $ST_{e,\sigma_{c,w}(\gamma,e)}$ associated with an event e and its scale of change $\sigma_{c,w}(\gamma,e)$, which is a partial function that maps an interval of time¹ to an interval of a scale of change associated with an eventuality.²

Definition 23 (Scalar trace function)

The partial surjective scalar trace function $ST_{e,\sigma_{c,w}(\gamma,e)}$: $Int(\mathcal{T}) \rightarrow Int(\sigma_{c,w}(\gamma,e))$ associated with eventuality *e* maps a subset of the set of time intervals to parts of the scale of change $\sigma_{c,w}(\gamma,e)$.

- 1. For any $i \in Int(\mathcal{T})$, if $ST_{e,\sigma_{c,w}(\gamma,e)}(i) = \{d\}$, where d is a set of states, then d must be instantiated at i which means that $\exists e' : i = \tau(\gamma, e') \land e' \in d$; and for any $i \in Int(\tau(\gamma, e))$, if $ST_{e,\sigma_{c,w}(\gamma,e)}(i) = \{d\}$, then there must be an $e' \sqsubseteq_E e$ such that $i = \tau(\gamma, e')$ and $e' \in d$.
- 2. $ST_{e,\sigma_{c,w}(\gamma,e)}$ is order-preserving relative to the non-strict linear orders over $Int(\mathcal{T})$ and $\sigma_{c,w}(\gamma,e)$, that is, $l_1 \leq_{\mathcal{T}} l_2 \Leftrightarrow ST_{e,\sigma_{c,w}(\gamma,e)}(l_1) \leq_{\sigma_{c,w}(\gamma,e)} ST_{e,\sigma_{c,w}(\gamma,e)}(l_2)$ (if $ST_{e,\sigma_{c,w}(\gamma,e)}$ is defined in l_1 and l_2).

The definition requires the scalar trace to be surjective, that is, all parts of the scale of change are mapped to some interval. Thus, the scale of change captures exactly the change that is associated with eventuality *e*. The scalar trace is a partial function, because it doesn't make sense to have the change associated with an eventuality (say, Mary's crossing the street at a given time) be defined for at all moments and intervals of time (say, at the time of the Big Bang or 2000 years after Mary's death).³

¹I here restrict scales of change and the scalar trace to time-relative cases, and so the following restrictions only apply to a scalar trace whose domain is a set of time intervals. But as Section 5.1.2 indicates, the system of scales of change and scalar traces can be extended to any linear order instead of time.

²The precise definition of a granularity-relative temporal trace $\tau(\gamma, e)$ is given in Section 4.2.

³This is an especially pertinent issue with respect to the maximum of the scale of change: intuitively, for an event of Mary's crossing the street, the final degree in its associated scale of change is Mary being on the other side of the street. But there is no degree on the scale of change to which we could map the times that are located years after Mary's death, or even after she moved from that side of the street: the final degree of Mary being on the other side of the street doesn't hold any more, and due to the requirement of order-preservation below, no other degrees can be images of these times, either.

The first requirement in the definition ensures that the mapping from times to a scale of change is not arbitrary: within the temporal trace of the event, the scale of change captures a property of the relevant temporal part of the event; and outside the temporal trace of the event, at least a state of the relevant kind has to hold. The second requirement in the definition ensures that the scale of change and the scalar trace respect temporal flow (i.e., an interval preceding another one cannot be mapped to a higher part of the scale).

Just as Beavers (2002, and elsewhere) put several additional constraints on his Movement Relation, there are some further constraints we have to put on the scalar trace, which are simply declared rather than derived from more basic principles. But they are not arbitrary: just as the formal constraints that Beavers (2002, and elsewhere) declared on his Movement Relation between events and scales and the formal constraints that Krifka (1992, and elsewhere) declared on the homomorphic mappings between an event and an incremental theme or path, the constraints on the scalar trace mapping were determined on the basis of the inspection of the linguistic behaviour of predicates denoting different sets of events. For better readability, let us here abbreviate $ST_{e,\sigma_{c,w}(\gamma,e)}$ as ST^e .

The domain of the scalar trace (i.e., the subset of Int(*T*) where ST^e is defined) is *downward closed* with respect to the subset relation between intervals: ∀I: I ∈ Dom(ST^e) ⇒ ∀I' ⊆ I: I' ∈ Dom(ST^e); and the domain must also have a greatest element, that is: ∃I ∈ Dom(ST^e): ∀I' ∈ Dom(ST^e): I' ⊆ I.

This ensures that the change that the scale of change captures happens over a given interval, and all parts of this interval correspond to a part of the change described by the scale of change (there are no "gaps").

2. $\forall I, I' \in \mathsf{Dom}(\mathsf{ST}^e) \colon (\gamma(\sup(I)) = \gamma(\sup(I')) \land \gamma(\inf(I)) = \gamma(\inf(I'))) \to \sigma_{c,w}(\gamma, e)(I) = \sigma_{c,w}(\gamma, e)(I)$

The scalar trace does not differentiate between intervals whose boundaries are not distinguished at granularity γ : all such intervals have the same image.

 Since the scale of change describes the series of states making up the change happening during an event *e*, the *scalar trace must be defined in the temporal trace* of *e*: τ(γ, *e*) ∈ Dom(ST^e).

Note that $\tau(\gamma, e)$ is not required to be the greatest element in $Dom(\sigma_e)$ with respect to set inclusion. (For example, in the case of events in the extension of achievements, points outside of the temporal trace of the event will also be mapped to the scale of change.)

4. The *final moment* of max $(Dom(ST^e))$ is identical to or precedes the moment up to which the state corresponding to max $(\sigma_{c,w}(\gamma, e))$ holds following the final moment of the temporal trace of the event *e*, where max is the greatest element of an ordered set (cf. Appendix A).

Intuitively, the last moment in time relevant to the change described by $\sigma_{c,w}(\gamma, e)$ is the moment up to which final state in $\sigma_{c,w}(\gamma, e)$ holds without interruption. This state may continue beyond the temporal trace of the eventuality e.⁴

- 5. The *initial moment* of max $(Dom(ST^e))$ is not so easily given a generic characterisation, as it is more subject to contextual determination. In the case of a durative dynamic event *e* (i.e., one to which activity and accomplishment predicates can apply), $\tau(\gamma, e)$ is an *initial subinterval* of Dom(ST^e), that is, the change associated with durative eventualities starts at the point an eventuality starts.
- 6. The *supremum* of the temporal trace of a *dynamic eventuality* e, if it exists, is mapped to a part of the scale to which no prior interval is mapped. That is, if $t = \sup (\tau(\gamma, e))$, then for any $i <_{\mathcal{T}} t$, $\mathsf{ST}^e(i) \neq \mathsf{ST}^e(t)$.
- 7. The *infimum* of the temporal trace of a *dynamic eventuality*, if it exists, is mapped to a part of the scale to which no interval following it is mapped. That is, if $t = \inf (\tau(\gamma, e))$, then for any *i* such that $t <_{\tau} i$, $ST^e(i) \neq ST^e(t)$.

This, together with the previous constraint, corresponds to the *minimality* requirement on the *movement relation* of Beavers (2002, 2008).

- 8. All intervals in the domain of the scalar trace that *precede the temporal trace* of the eventuality are mapped to the *same* degree of the scale. That is, for any *i*, *i'* such that both $i <_{\tau} \tau(\gamma, e)$ and $i' <_{\tau} \tau(\gamma, e)$, it holds that $ST^{e}(i) = ST^{e}(i')$, if ST^{e} is defined in *i*, *i'*.
- 9. All intervals in the domain of the scalar trace that *follow the temporal trace* of the eventuality are mapped to the *same* degree of the scale as the final moment of the temporal trace. That is, if $t = \max(\tau(\gamma, e))$, then for any *i* such that $t <_{\mathcal{T}} i$, it holds that $\mathsf{ST}^e(i) = \mathsf{ST}^e(t)$, if ST^e is defined in *i*.

This, together with the previous constraint ensures that no event-relevant change is happening outside the eventuality, i.e., outside of an event, only a pre-state and a post-state can hold (if at all), but no change.

⁴As such, the scale of change can be seen as an attempt to capture the *nucleus structure* of Moens and Steedman (1988), and the event structure of Pustejovsky (1991). In both cases, the event structure for telic predicates contains a result state, in particular, which is inherently associated with an event in the extension of the relevant predicate. Because an eventuality can be associated with different result states under different descriptions, the result state should not be a function of solely the event. On the present framework, this is accomplished with the contextually determined scale of change: the discourse context parameter selects the series of states that is relevant for the discourse purposes. Making the concept of a result state dependent on the context paves the way for explaining coercions like *Today, John swam in an hour* from Dowty (1979, p. 61): although only accomplishments, but not activities, require reference to a result state in a particular context, the discourse context may still specify a result state for activities, as well.

- 10. For a dynamic eventuality e, if $\tau(\gamma, e) = t$ for some $t \in T$, then $\exists i <_{\mathcal{T}} t : i \in \text{Dom}(ST^e)$. That is, if a dynamic event is instantaneous, then the domain of the scalar trace includes some interval preceding it.
- 11. For state $s, \forall i, i' \in \text{Dom}\left(\mathsf{ST}_{s,\sigma_{c,w}(\gamma,s)}\right)$: $\mathsf{ST}_{s,\sigma_{c,w}(\gamma,s)}(i) = \mathsf{ST}_{s,\sigma_{c,w}(\gamma,s)}(i')$.

That is, all elements of the domain of the scalar trace for a state are mapped to the same degree. Given the surjectivity of the scalar trace, this entails that states are associated with *singleton* scales.

Durativity and scale complexity. The constraints on the scalar trace ensure that if a dynamic eventuality is non-extended, then its scale of change is *not complex* (does not consist of more than two degrees), because times preceding the (temporal trace of) the event can only be mapped to a single degree (requirement 8), and the times following it can only mapped to the same degree as the time point corresponding to the event's temporal trace (requirement 9). With the added assumption in 10, that in the case of non-extended dynamic events, the context always includes an interval preceding their temporal trace in the domain of the scalar trace, this ensures that non-extended dynamic events are associated with a *binary scale*, i.e., one consisting of exactly two degrees.

Conversely, if an eventuality is dynamic and extended, then its scale of change is *complex*, because both greatest lower bound (requirement 7) and the least upper bound (requirement 6) of the temporal trace have different images as other parts of the temporal trace. Thus, the present framework retains the feature of the framework of Beavers (2002, 2008) according to which scale complexity corresponds to durativity, despite differences between the scalar trace and Beavers's movement relation.

Note that the requirements also ensure that the scale of change associated with any particular eventuality token is *closed*, i.e., it has a maximum and a minimum. Importantly, this is not at odds with the characterisation of atelic *predicates* as being associated with an open (rather than closed) scale (cf. Section 2.3.2.2). At the level of *events*, just as Filip and Rothstein (2006) and Filip (2008) have argued that there is always a *maximal event* with respect to an independent ordering in given context, so, in a given context, for a given event (which corresponds to the maximal event of Filip and Rothstein 2006 and Filip 2008), the change associated with it has a definite endpoint.

But for an activity predicate P, since it can continue potentially indefinitely, then for any $e \in P$, there is necessarily some $e' \in P$ occurring in some possible world such that $\sigma_{c,w}(\gamma, e) \subset_{nf} \sigma_{c,w'}(\gamma, e')$. Thus, the *generalized scale* associated with P is open, as for each $S \in PrSoC(^P)$, there is a bigger $S' \in PrSoC(^P)$ (where PrSoC returns the set of primary scales of change associated with an eventuality predicate, cf. Definition 15).

PLAGIARISM DECLARATION

I hereby declare that I prepared this thesis independently and without using any aids other than those indicated and that all ideas taken directly or indirectly from other sources have been indicated as such and that the thesis has so far neither been submitted to any examination authority nor was published in the same or a similar form. Parts that have already been published are indicated in the thesis.

Ich erkläre hiermit, dass ich die vorliegende Arbeit selbständig und ohne Benutzung anderer als der angegebenen Hilfsmittel angefertigt und die aus fremden Quellen direkt oder indirekt übernommenen Gedanken als solche kenntlich gemacht habe und dass die Arbeit bisher in gleicher oder ähnlicher Form keiner Prüfungsbehörde vorgelegt und auch noch nicht veröffentlicht wurde. Bereits veröffentlichte Teile sind in der Arbeit gekennzeichnet.

Düsseldorf, 5th June 2015

Zsófia Gyarmathy